%\documentclass[a4paper,12pt]{article} %%% Version of 
\documentclass[11pt,reqno]{amsart}
%\usepackage{amssymb,a4wide,eucal}




%\usepackage{hyperref}
\usepackage {amsmath, amsfonts, amssymb, amscd, amsthm, eucal,a4wide}



\usepackage{amsbsy}
\usepackage[all]{xy}
\SelectTips{cm}{} 
\SilentMatrices
\usepackage[dvipsnames,usenames]{color}
%\usepackage[pdftex,colorlinks,linkcolor=BrickRed,citecolor=OliveGreen]{hyperref}
\usepackage[pdftex,colorlinks,linkcolor=black,citecolor=black]{hyperref}

%\documentclass[a4paper,12pt]{article} %%% Version of March 22, 2008
%
%%\usepackage{hyperref}
%\usepackage {amsmath, amsfonts, amssymb, amscd, amsthm, eucal}
%
%
%
%\usepackage{amsbsy}
%\usepackage[all]{xy}
%\SelectTips{cm}{} \special{papersize=21cm,29.7cm} \topmargin 1 cm
%\topskip 0 cm \textwidth 16 cm \textheight 22 cm \oddsidemargin 0.5
%cm \evensidemargin 0.5 cm \headheight 0 cm \headsep 0 cm
%\marginparwidth 0 cm \footskip 1.2 cm
% MATH -------------------------------------------------------------------
\theoremstyle{plain}{
%\swapnumbers
  \newtheorem{thm}{Theorem}[section]
  \newtheorem{cor}[thm]{Corollary}
  \newtheorem{lem}[thm]{Lemma}
  \newtheorem{prop}[thm]{Proposition}
  \newtheorem{clm}[thm]{Claim}
  \newtheorem{notation}[thm]{Notation}
  \newtheorem{agr}[thm]{Agreement}
  \newtheorem{sssection}[thm]{}}

\theoremstyle{definition}{
  \newtheorem{defn}[thm]{Definition}
  \newtheorem{example}[thm]{Example}
  \newtheorem{question}[thm]{Question}}

\theoremstyle{remark}{
  \newtheorem{rem}[thm]{Remark}}

%\swapnumbers
%}
%\renewcommand{\subsubsection}{\sssection\rm}

\renewcommand{\subsubsection}{\sssection\rm}

\numberwithin{equation}{section}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%\theoremstyle{plain}
%
%\swapnumbers
%  \newtheorem{thm}{Theorem}[subsection]
%  \newtheorem{defn}[thm]{Definition}
%  \newtheorem{cor}[thm]{Corollary}
%  \newtheorem{lem}[thm]{Lemma}
%  \newtheorem{prop}[thm]{Proposition}
%  \newtheorem{rem}[thm]{rem}
%  \newtheorem{clm}[thm]{Claim}
%  \newtheorem{notation}[thm]{Notation}
%  \newtheorem{agr}[thm]{Agreement}
%

%\theoremstyle{definition}
%
%\swapnumbers
%  \newtheorem{sssection}[thm]{}
%
%\renewcommand{\subsubsection}{\sssection}
%-----------------------------------------------------------------------
\newcommand{\pt}{pt}
\newcommand{\Field}{k}
\newcommand{\gys}{gys}
\newcommand{\quil}{quil}
\newcommand{\id}{\mathrm{id}}
\newcommand{\Spec}{\mathrm{Spec}}
\newcommand{\BGL}{\mathrm{BGL}}
\newcommand{\Hom}{\operatorname{Hom}}
\newcommand{\Ker}{\operatorname{Ker}}
\newcommand{\Aut}{\mathrm{Aut}}
\DeclareMathOperator{\rk}{rk}
\newcommand{\Pic}{\mathrm{Pic}}
\newcommand{\MU}{\mathrm{MU}}
\newcommand{\charac}{\mathrm{char}}
\newcommand{\Id}{\mathrm{Id}}
\newcommand{\Nis}{\mathrm{Nis}}
\newcommand{\fib}{\mathrm{fib}}
\newcommand{\cofrepl}{\mathrm{cof}}
\newcommand{\op}{\mathrm{op}}
\newcommand{\cm}{\mathrm{cm}}
\newcommand{\Fr}{\mathrm{Fr}}
\newcommand{\Tr}{\mathrm{Tr}}
\newcommand{\TT}{\mathrm{TT}}
\newcommand{\Vect}{\mathrm{Vect}}
\newcommand{\tlder}{\mathcal{L}}
\newcommand{\trder}{\mathcal{R}}
\newcommand{\Sing}{\operatorname{Sing}}
\newcommand{\Ho}{\operatorname{Ho}}
\newcommand{\Func}{\operatorname{Fun}}
\DeclareMathOperator{\Th}{Th}
\DeclareMathOperator{\MGL}{\mathbf {MGL}}
\newcommand{\SmOp}{\mathcal Sm\mathcal Op}
\newcommand{\Sm}{\mathcal Sm}
\newcommand{\Spc}{\mathcal Spc}
\newcommand{\Spectra}{\mathrm{Sp}}
\newcommand{\Sp}{\mathrm{Sp}}
\newcommand{\SmCor}{\mathcal Sm \mathcal Cor}
\newcommand{\Ab}{\mathcal Ab}
\newcommand{\C}{\mathcal {C}}
\newcommand{\T}{\mathcal {T}}
\newcommand{\Aff}{\mathbf{A}}
\newcommand{\Pro}{\mathbf{P}}
\newcommand{\Gr}{Gr}
%\newcommand{\HGr}{\mathrm{HGr}}
\newcommand{\HG}{\mathrm{HG}}
\newcommand{\HP}{HP}
\newcommand{\F}{\mathbf {F}}
\newcommand{\Res}{\mathrm{Res}}
\newcommand{\td}{td}
\newcommand{\itd}{itd}
%%%%%%%%%%%%%%%%%%%%%%%%%
\newcommand{\A}{\mathcal A}
\newcommand{\B}{\mathcal B}
\newcommand{\K}{\mathcal K}
\newcommand{\J}{\mathcal J}



\newcommand{\ZZ}{\mathbb{Z}}
\newcommand{\one}{\mathbb{I}}

%%% More abbreviations

\newcommand{\real}{\mathbf{R}}
\newcommand{\M}{\mathbf{M}}
\newcommand{\MS}{\mathbf{MS}}
\newcommand{\MSS}{\mathbf{MSS}}
\newcommand{\sSet}{\mathbf{sSet}}
\newcommand{\Top}{\mathbf{Top}}

\newcommand{\an}{\mathrm{an}}
\newcommand{\SH}{\mathrm{SH}}


\newcommand{\iso}{\cong}
\newcommand{\union}{\cup}

\newcommand{\CC}{\mathbb{C}}
%\newcommand{\AA}{\mathbb{A}}
%\newcommand{\P}{\mathbb{P}^1}

\newcommand{\colim}{\operatornamewithlimits{colim}}
\newcommand{\coeq}{\operatorname{coeq}}
\newcommand{\intHom}{\underline{\Hom}}
\newcommand{\Cyl}{\mathrm{Cyl}}
\newcommand{\defeq}{\colon \!\! =}
\newcommand{\push}{\Box}
\newcommand{\cof}{\mathrm{cof}}

%\let\smash=\wedge
%%% ----------------------------------------------------------------------
\newcommand \xra {\xrightarrow }
\newcommand \ra {\rightarrow }
\newcommand \lra {\longrightarrow }
\newcommand \hra {\hookrightarrow }
\newcommand \mt {\mapsto }
\newcommand \Lra {\Leftrightarrow }
\newcommand \Ra {\Rightarrow }
\newcommand \xla {\xleftarrow }

%%%--------------------------------------
\newcommand{\MSp}{\mathbf{MSp}}
\newcommand{\MSL}{\mathbf{MSL}}
\newcommand{\angles}[1]{\langle #1 \rangle}
\DeclareMathOperator{\HGr}{\mathnormal{HGr}}
\DeclareMathOperator{\USp}{\mathnormal{HU}}
\newcommand{\OO}{\mathcal{O}}
\newcommand{\parens}[1]{\textup{(}#1\textup{)}}
\DeclareMathOperator{\GL}{\mathnormal{GL}}
\newcommand{\veps}{\varepsilon}
\newcommand{\sm}[1]{\left(\begin{smallmatrix}#1\end{smallmatrix}\right)}
\DeclareMathOperator{\thom}{\mathnormal{th}}
\newcommand{\onto}{\twoheadrightarrow}
\DeclareMathOperator{\coker}{coker}
\DeclareMathOperator{\res}{res}
\newcommand{\iHom}{\operatorname{\mathbf{Hom}}_{\bullet}}
\newcommand{\fhom}{\operatorname{\mathbf{hom}}_{\bullet}}
\newcommand{\sym}{\Sigma}
\DeclareMathOperator{\hocolim}{hocolim}
\newcommand{\finite}{\text{\textit{fin}}}
\newcommand{\geom}{\text{\textit{geom}}}
\newcommand{\mono}{\rightarrowtail}
\newcommand{\shf}{\mathcal}
\newcommand{\ft}{\text{\textit{ft}}}
\newcommand{\homog}{\text{\textit{hom}}}
\newcommand{\sphere}{\mathbb{S}}
\DeclareMathOperator{\Mat}{Mat}
\newcommand{\GrSL}{SGr}
%{\operatorname{\mathnormal{Gr}}_{SL}}
\newcommand{\NN}{\mathbb{N}}
\newcommand{\HH}{\mathbb{H}}
\newcommand{\GG}{\mathbb{G}}
\DeclareMathOperator{\dirsum}{dirsum}
\DeclareMathOperator{\concat}{concat}
\DeclareMathOperator{\mult}{mult}
\newcommand{\TGL}[1]{\mathcal{T}GL_{#1}}
\newcommand{\TSL}[1]{\mathcal{T}SL_{#1}}
\newcommand{\TSp}[1]{\mathcal{T}Sp_{#1}}
\newcommand{\Ev}{\mathrm{Ev}}

%%%%%%%%%%%%%%%%%%%%

\begin{document}
\begin{thebibliography}{10}

\bibitem{Balmer:2008fk}
{\sc P.~Balmer and B.~Calm{\`e}s}, {\em Witt groups of {G}rassmann varieties},
  (2008).
\newblock Preprint, 2008.

\bibitem{Dundas:2003aa}
{\sc B.~I. Dundas, O.~R{\"o}ndigs, and P.~A. {\O}stv{\ae}r}, {\em Motivic
  functors}, Doc. Math., 8 (2003), pp.~489--525 (electronic).

\bibitem{Hovey:2001aa}
{\sc M.~Hovey}, {\em Spectra and symmetric spectra in general model
  categories}, J. Pure Appl. Algebra, 165 (2001), pp.~63--127.

\bibitem{Isaksen:2005aa}
{\sc D.~C. Isaksen}, {\em Flasque model structures for simplicial presheaves},
  $K$-Theory, 36 (2005), pp.~371--395.

\bibitem{Jardine:2000aa}
{\sc J.~F. Jardine}, {\em Motivic symmetric spectra}, Doc. Math., 5 (2000),
  pp.~445--552 (electronic).

\bibitem{Levine:2007ys}
{\sc M.~Levine and F.~Morel}, {\em Algebraic cobordism}, Springer Monographs in
  Mathematics, Springer, Berlin, 2007.

\bibitem{Macdonald:1995zl}
{\sc I.~G. Macdonald}, {\em Symmetric functions and {H}all polynomials}, Oxford
  University Press, Oxford, 2nd~ed., 1995.

\bibitem{Morel:1999ab}
{\sc F.~Morel and V.~Voevodsky}, {\em {${\bf A}^1$}-homotopy theory of
  schemes}, Inst. Hautes \'Etudes Sci. Publ. Math.,  (1999), pp.~45--143
  (2001).

\bibitem{Panin:2003rz}
{\sc I.~Panin}, {\em Oriented cohomology theories of algebraic varieties},
  $K$-Theory, 30 (2003), pp.~265--314.
\newblock Special issue in honor of Hyman Bass on his seventieth birthday. Part
  III.

\bibitem{Panin:2008db}
{\sc I.~Panin, K.~Pimenov, and O.~R{\"o}ndigs}, {\em A universality theorem for
  {V}oevodsky's algebraic cobordism spectrum}, Homology, Homotopy Appl., 10
  (2008), pp.~211--226.

\bibitem{Panin:2009aa}
\leavevmode\vrule height 2pt depth -1.6pt width 23pt, {\em On {V}oevodsky's
  algebraic {$K$}-theory spectrum}, in Algebraic topology, vol.~4 of Abel
  Symp., Springer, Berlin, 2009, pp.~279--330.

\bibitem{Panin:2010aa}
{\sc I.~Panin and C.~Walter}, {\em On the motivic commutative ring spectrum
  {$\mathbf{BO}$}}.
\newblock Preprint, 2010.

\bibitem{Panin:2010fk}
\leavevmode\vrule height 2pt depth -1.6pt width 23pt, {\em {Q}uaternionic
  {G}rassmannians and {P}ontryagin classes in algebraic geometry}.
\newblock Preprint, 2010.

\bibitem{Vezzosi:2001aa}
{\sc G.~Vezzosi}, {\em Brown-{P}eterson spectra in stable
  {$\mathbb{A}^1$}-homotopy theory}, Rend. {S}em. {M}at. Univ. Padova, 106
  (2001), pp.~47--64.

\bibitem{Voevodsky:1998kx}
{\sc V.~Voevodsky}, {\em {$\mathbf{A}\sp 1$}-homotopy theory}, Doc. Math.,
  Extra Vol. I (1998), pp.~579--604 (electronic).

\bibitem{Voevodsky:2007aa}
{\sc V.~Voevodsky, O.~R{\"o}ndigs, and P.~A. {\O}stv{\ae}r}, {\em Voevodsky's
  {N}ordfjordeid lectures: {M}otivic homotopy theory}, in Motivic homotopy
  theory, Universitext, Springer, Berlin, 2007, pp.~147--221.

\end{thebibliography}


\title{On the algebraic cobordism spectra $\MSL$ and $\MSp$}

\author{Ivan Panin} 
\address{Steklov Institute of Mathematics at St.~Petersburg, Russia}
%\and K.~Pimenov\footnotemark[1]
\author {Charles Walter}
\address{Laboratoire J.-A.\ Dieudonn\'e (UMR 6621 du CNRS)\\
 D\'epartement de math\'ematiques\\
 Universit\'e de Nice -- Sophia Antipolis\\ 06108 Nice Cedex 02\\
 France}


\date{October 28, 2010}\thanks{The work was supported by
Universite de Nice - Sophia-Antipolis ,
the Presidium of RAS Program
``Fundamental Research in modern mathematics",
the joint DFG-RFBR grant 09-01-91333 NNIO-a,
and the RFBR-grant 10-01-00551. }



%\author{I.~Panin\footnote{Steklov Institute of Mathematics at St.~Petersburg, Russia}
%%\and K.~Pimenov\footnotemark[1]
%\and Ch.~Walter\footnote{University, Nice, France}}
%
%\date{May 1, 2010\thanks{The first author thanks
%the RFFI-grant ??????, University of Nice, DFG--RFFI project
%09-01-91333-NNIO-a.}}


\begin{abstract}
We construct algebraic cobordism spectra $\MSL$ and $\MSp$.
They are commutative monoids in the
category of symmetric $T^{\wedge 2}$-spectra.
The spectrum $\MSp$ comes with a natural symplectic orientation given either
by a tautological Thom class
$th^{\MSp} \in \MSp^{4,2}(\MSp_2)$, a tautological Pontryagin class
$p_{1}^{\MSp} \in \MSp^{4,2}(HP^{\infty})$ or any of six other equivalent structures.
For a  commutative monoid $E$ in the category
${SH}(S)$
we prove that assignment
$\varphi \mapsto \varphi(\thom^{\MSp})$
identifies the set of homomorphisms of monoids
$\varphi\colon \MSp \to E$
in the motivic stable homotopy category
$SH(S)$
with the set of tautological Thom elements of symplectic orientations of $E$.
A weaker universality result is obtained for $\MSL$ and 
special linear orientations.
\end{abstract}

\maketitle


\section{Introduction}


A dozen years ago Voevodsky \cite{Voevodsky:1998kx}
constructed the algebraic cobordism spectrum $\MGL$
in the motivic stable homotopy category $SH(S)$.  This gave a new cohomology theory
$\MGL^{*,*}$ on smooth schemes and on motivic spaces.  Later Vezzosi \cite{Vezzosi:2001aa}
put a commutative monoid structure on $\MGL$.  This gave a product to $\MGL^{*,*}$.
The commutative monoid structure can even be constructed in the symmetric monoidal model category
of symmetric $T$-spectra, with $T = \Aff^{1}/(\Aff^{1}-0)$ the Morel-Voevodsky
object  (Panin, Pimenov and R\"ondigs \cite{Panin:2008db}).

In this paper we construct the algebraic special linear and symplectic cobordism
spectra $\MSL$ and $\MSp$.   The construction of $\MSL$ is straightforward although there is one slightly 
subtle point.  We equip each space $BSL_{n}$ and $\MSL_{n}$ with an action of $GL_{n}$ which
is compatible with the monoid structure $BSL_{m} \times BSL_{n} \to BSL_{m+n}$ induced by the
direct sum of subbundles.  This gives an action of the subgroup $\sym_{n} \subset GL_{n}$
of permutation matrices.  But to define the unit of the monoid structure we need the action 
on $BSL_{n}$ to have fixed points.  The natural action of $SL_{n}$ has fixed points, but the natural action
of $GL_{n}$ does not.  So we use an embedding $\sym_{n} \subset Sp_{2n} \subset SL_{2n}$.
This means that our $\MSL$ is a commutative monoid in the category of 
symmetric $T^{\wedge 2}$-spectra.
%
The categories of symmetric $T$-spectra and of symmetric $T^{\wedge 2}$-spectra
are both symmetrical monoidal, and their homotopy categories are equivalent symmetric
monoidal categories  (Theorem \ref{T:comparison}).  So a symmetric $T^{\wedge 2}$-spectrum
structure is quite satisfactory, and it seems to be a natural structure for this spectrum.

Cobordism spectra and the cohomology theories they define are expected to have some universal
properties among certain classes of cohomology theories.  For instance Voevodsky's and
Levine and Morel's algebraic cobordism theories  are universal
among oriented cohomology theories \cite{Levine:2007ys,Panin:2008db,Vezzosi:2001aa}.
%
We should therefore expect $\MSL$ to have 
some degree of universality for special linearly oriented theories.
Recall that a special linear bundle $(E,\lambda)$ over $X$ is a pair consisting of a vector bundle
$E$ and an isomorphism of line bundles $\lambda \colon \OO_{X} \cong \det E$.  A special linear
orientation on a cohomology theory $A^{*,*}$ is an assignment to every special linear bundle of 
a Thom class
$\thom(E,\lambda) \in A^{2n,n}(E,E-X) = A^{2n,n}_{X}(E)$ with $n = \rk E$ which is functorial, 
multiplicative, and
such that the multiplication maps ${-}\cup \thom(E,\lambda) \colon A^{*,*}(X) \to A^{*+2n,*+n}(E,E-X)$
are isomorphisms. In the motivic context we generally also require 
that the Thom class of the trivial line bundle over a point be
$\Sigma_{T}1_{A} \in A^{2,1}(T) = A^{2,1}(\Aff^{1},\Aff^{1}-0)$.  Hermitian $K$-theory and
Balmer's derived Witt groups are examples of special linearly oriented theories which are not
oriented.


The universality properties we show for $\MSL$ are as follows.
%
%\begin{itemize}
%\item 
A morphism of commutative monoids $\varphi \colon (\MSL,\mu^{SL},e^{SL}) \to (A,\mu,e)$
in $SH(S)$ determines naturally a special linear orientation on $A^{*,*}$ with Thom classes
written $\thom^{\varphi}(E,\lambda)$.
The compatibility of $\varphi$ with the monoid structure ensures the multiplicativity of the
Thom classes (Theorem \ref{T:thom.phi}).

%\item 


Conversely, a special linear orientation on $A^{*,*}$ with Thom classes $\thom(E,\lambda)$ determines a 
morphism $\varphi \colon \MSL \to A$ in $SH(S)$ with 
$\thom^{\varphi}(E,\lambda) = \thom(E,\lambda)$ for all $(E,\lambda)$.  This $\varphi$ is
unique modulo a certain subgroup $\varprojlim^{1} A^{2n-1,n}(MSL_{n}^{\angles{n}}) \subset
Hom_{SH(S)}(\MSL,A)$.  
The obstruction $\varphi \circ \mu^{SL} - \mu_{A} \circ (\varphi \wedge \varphi)$ 
to having a morphism of monoids lies in a similarly defined subgroup
of $Hom_{SH(S)}(\MSL \wedge \MSL, A)$ (Theorem \ref{T:univ.SL}).


%\end{itemize}

It would be interesting to know if these obstruction subgroups vanish for Witt groups
and hermitian $K$-theory. 
The necessary calculations are likely very close to Balmer and Calm\`es's computation of Witt groups
of Grassmannians \cite{Balmer:2008fk}.

Our $\MSp$ is defined similarly with an action of $Sp_{2n}$ on the spaces
$BSp_{2n}$ and $\MSp_{2n}$.  The actions of the subgroups $\sym_{n} \subset Sp_{2n}$ make
$\MSp$ a commutative monoid in the category of symmetric $T^{\wedge 2}$-spectra.  
For $\MSp$ we can do much more than for $\MSL$ because 
we have the quaternionic projective bundle theorem \cite[Theorem 8.2]{Panin:2010fk}
for symplectically oriented cohomology theories.  Therefore for any symplectically oriented
cohomology theory $A^{*,*}$ we have Pontryagin classes for symplectic bundles, and we
can compute the cohomology of quaternionic Grassmannians
\cite[\S 11]{Panin:2010fk}
and of the spaces $BSp_{2r}$ and $\MSp_{2r}$ 
(\S\S\ref{S:projective.bundle}--\ref{S:cohom.BSp.MSp}).
Our main result is the following theorem.

%\dots
%
%
%We use the parts of \cite{Panin:2010fk} which concern the geometry of quaternionic projective
%spaces and Grassmannians, the quaternionic projective bundle theorem, the definition of Pontryagin classes,
%the Cartan sum formula for Pontryagin classes, and the calculation of the cohomology of quaternionic 
%Grassmannians.
%
%But we redo in a more motivic context the final part of \cite{Panin:2010fk} where four different 
%axiomatizations of a symplectically oriented cohomology theory were shown to be equivalent.

\begin{thm}
\label{T:main.Sp}
Let $(A,\mu,e)$ be a commutative monoid in $SH(S)$.  Then the following sets are in canonical
bijection\textup{:}

\parens{a} symplectic Thom structures on the bigraded $\epsilon$-commutative ring cohomology theory 
$(A^{*,*},\partial,\times,1_{A})$ such that for the trivial rank $2$
bundle $\Aff^{2} \to \pt$ we have 
$\thom(\Aff^{2},\omega_{2}) = \Sigma_{T}^{2}1_{A}$ in $A^{4,2}(T^{\wedge 2})$,

\parens{b} Pontryagin structures on 
%the bigraded $\epsilon$-commutative ring cohomology theory 
$(A^{*,*},\partial,\times,1_{A})$ for which 
$p_{1}(\shf U_{HP^{1}},\phi_{HP^{1}}) \in A^{4,2}(HP^{1},h_{\infty}) \subset A^{4,2}(HP^{1})$
corresponds to $-\Sigma_{T}^{2}1_{A}$ in $A^{4,2}(T^{\wedge 2})$ under the canonical 
motivic homotopy equivalence $(HP^{1},h_{\infty}) \simeq T^{\wedge 2}$,

\parens{c} Pontryagin classes theories on $(A^{*,*},\partial,\times, 1_{A})$ 
with the same normalization condition on $p_{1}(\shf U_{HP^{1}},\phi_{HP^{1}})$ as in \parens{b},

\parens{d} symplectic Thom classes theories on 
%the bigraded $\epsilon$-commutative ring cohomology theory 
$(A^{*,*},\partial,\times,1_{A})$ such that for the trivial rank $2$
bundle $\Aff^{2} \to \pt$ we have 
$\thom(\Aff^{2},\omega_{2}) = \Sigma_{T}^{2}1_{A}$ in $A^{4,2}(T^{\wedge 2})$,



\parens{$\alpha$} classes $\vartheta \in A^{4,2}(\MSp_{2})$ with 
$\vartheta |_{T^{\wedge 2}} = \Sigma_{T}^{2}1_{A}$ in $A^{4,2}(T^{\wedge 2})$,


\parens{$\beta$} classes $\varrho \in A^{4,2}(HP^{\infty},h_{\infty})$ with 
$\varrho|_{HP^{1}} \in A^{4,2}(HP^{1},h_{\infty})$ corresponding to 
$-\Sigma_{T}^{2}1_{A} \in A^{4,2}(T^{\wedge 2})$ under the canonical motivic homotopy equivalence 
$(HP^{1},h_{\infty}) \cong T^{\wedge 2}$, 

\parens{$\delta$} sequences of classes 
$\boldsymbol{\vartheta} = (\vartheta_{1},\vartheta_{2},\vartheta_{3},\dots)$
with $\vartheta_{r} \in A^{4r,2r}(\MSp_{2r})$ for each $r$ satisfying 
$\mu_{rs}^{*} \vartheta_{r+s} = \vartheta_{r} \times \vartheta_{s}$ for all $r,s$,
and $\vartheta_{1} |_{T^{\wedge 2}} = \Sigma_{T}^{2}1_{A}$,


\parens{$\varepsilon$} morphisms $\varphi \colon (\MSp,\mu^{Sp},e^{Sp}) \to (A,\mu,e)$ 
of commutative monoids
in $SH(S)$.
\end{thm}

The bijections are explicit and are given in a series of theorems in the last part of the paper.
The presence of $(\varepsilon)$ among them is the universality of $\MSp$ as a symplectically 
oriented theory.

The equivalence of (a), (b), (c) and (d) was already shown in \cite{Panin:2010fk} in a 
different axiomatic context.  
%
The ability of the motivic language used here to handle tautological classes such as the
$(\alpha)$, $(\beta)$ and $(\delta)$ is very useful by itself.  But
our main new observation is that in the motivic unstable homotopy category $H_{\bullet}(S)$ 
we have a commutative diagram (Theorem \ref{T:MSp=BSp/BSp}).
\[
\xymatrix @M=5pt {
& BSp_{2r} \ar@/_/[ld]_-{\text{\textup{structure map}}} \ar@/^/[rd]^-{\text{\textup{quotient}}}
\\
\MSp_{2r} \ar@{<->}[rr]^-{\cong}_-{\text{\textup{$\Aff^{N}$-bundles and excision}}} 
&& BSp_{2r}/BSp_{2r-2}.
}
\]
What is surprising about this diagram is that it is the homotopy colimit of diagrams \eqref{E:amazing.1}
of finite-dimensional schemes and their quotient spaces which have a fourth side which is an inclusion
of quaternionic Grassmannians of different dimensions 
which is in no way a
motivic equivalence.  
But in the infinite-dimensional colimit the fourth side becomes $\Aff^{1}$-homotopic to the identity map
of $BSp_{2r}$, and the picture simplifies significantly.

The fact that this diagram is three-sided instead of four-sided 
helps us to see more conceptual proofs of two of the trickier points of 
\cite{Panin:2010fk}.  One is the construction of the higher-rank symplectic Thom classes and the
proof of their multiplicativity.  The commutativity of the diagram and the computations of the
cohomology of quaternionic Grassmannians imply that given a symplectic Thom structure on $A^{*,*}$,
the pullback along the structure map gives an injection $A^{*,*}(\MSp_{2r}) \to A^{*,*}(BSp_{2r})$,
and the isomorphism $A^{*,*}(\pt)[[p_{1},\dots,p_{r}]]^{\homog} \cong A^{*,*}(BSp_{2r})$
defined by the symplectic Thom structure identifies the image of $A^{*,*}(\MSp_{2r})$
with the principal two-sided ideal generated by $p_{r}$.  This makes it easy to define
the higher-rank tautological symplectic Thom classes (the $\vartheta_{r}$ of ($\delta$)) with the classes
of $A^{*,*}(\MSp_{2r})$ identified with $(-1)^{r}p_{r} \in A^{*,*}(\pt)[[p_{1},\dots,p_{r}]]^{\homog}$.
Their multiplicativity is also easily established.

The other tricky point of \cite{Panin:2010fk} for which the diagram helps is 
the reconstitution of the symplectic Thom structure
from the Pontryagin structure.  The tautological rank $2$ Thom class is a $\vartheta \in A^{4,2}(\MSp_{2})$,
and it is tempting to identify it (up to sign) with the tautological rank $2$ Pontryagin class 
$\varrho \in A^{4,2}(BSp_{2}/BSp_{0}) = A^{4,2}(HP^{\infty},h_{\infty})$ using the horizontal
motivic homotopy equivalence.  But the Pontryagin is actually (up to sign) the pullback of $\vartheta$
along the structure map of the Thom space.  For the three-sided diagram this is no problem:  
in $H_{\bullet}(S)$ the structure map $HP^{\infty} \to \MSp_{2}$ is the composition of the
horizontal isomorphism $(HP^{\infty},h_{\infty}) \cong \MSp_{2}$ 
with the pointing map.


%There are four different structures on ring cohomology theories
%on $\SmOp/S$ 
%which are equivalent to symplectic
%orientations.  They are as follows for bigraded theories.
%
%\begin{itemize}
%\item
%A \emph{symplectic Thom structure} 
%\cite[Definition 7.1]{Panin:2010fk} 
%where one specifies a Thom class $\thom(E,\phi) \in A^{4,2}(E,E-X)$
%for every rank $2$ symplectic bundle.
%
%\item
%A \emph{Pontryagin structure} 
%\cite[Definition 12.1]{Panin:2010fk} 
%where one specifies a Pontryagin class $p_{1}(E,\phi) \in A^{4,2}(X)$
%for every rank $2$ symplectic bundle.
%
%\item
%A \emph{symplectic Thom classes theory}
%\cite[Definition 14.2]{Panin:2010fk} 
%where one specifies a Thom class $\thom(U,\psi) \in A^{4r,2r}(U,U-X)$
%for every symplectic bundle. The bidegree depends on $\rk U = 2r$.
%
%
%
%\item
%A \emph{Pontryagin classes theory}
%\cite[Definition 14.1]{Panin:2010fk} 
%where one specifies Pontryagin classes $p_{i}(U,\psi) \in A^{4i,2i}(X)$ 
%for $1 \leq i \leq r$ for every symplectic bundle.
%\end{itemize}
%
%In each structure the characteristic classes obey a certain number of axioms.  
%A structure of any of the four kinds determines unique structures of each of the three other
%kinds \cite[Theorem 14.4]{Panin:2010fk}.
%
%The Thom,
%Pontryagin and Euler classes of a symplectic bundle $(U,\psi)$ of rank $2r$ over $X$
%are related by
%%
%\begin{equation}
%\label{Pontryagin.Euler}
%(-1)^{r}p_{r}(U,\psi) = e(U,\psi) = z^{*}\thom(U,\psi)
%\end{equation}
%%
%where $z \colon X = (X,\varnothing) \to (U,U-X)$ is the morphism in $\SmOp/S$
%induced by the zero section of $U$.



It is not difficult to define spectra $\mathbf{MO}$ and $\mathbf{MSO}$ which resemble formally
$\MGL$ and our $\MSL$ and $\MSp$.  However, our proof of even our
most basic result about $\MSL$ (Theorem \ref{T:thom.phi}) 
uses the fact that special linear bundles are locally trivial in the 
Zariski topology.  So we omit
$\mathbf{MO}$ and $\mathbf{MSO}$.




\section{Preliminaries}
\label{S:prelim}
%Here we describe a ring structure on the algebraic cobordism of Voevodsky
%\cite{Voevodsky:1998kx}.
%\subsubsection{\bf Algebraic cobordism theory.}
%\label{SymmetricRingStructureOnCobordism}
%The algebraic cobordism theory of Voevodsky is a cohomology theory represented by
%the $T$-spectrum $MGL$
%(see \cite{Voevodsky:1998kx}).
%To introduce a ring
%structure on the algebraic cobordism theory
% (\ref{RepresentableTheories1})
%it would be convenient to enrich
%the $T$-spectrum $MGL$
%with a symmetric ring structure
%\cite[Sect.4]{Ja}.
%For that
%we construct another $T$-spectrum $\Bbb{MGL}$ which is a
%commutative symmetric ring spectrum by the very construction and
%which is weekly equivalent to $MGL$ as the $T$-spectrum. Now we
%are going to describe $\Bbb{MGL}$. A ring structure on the
%algebraic cobordism theory was introduced as well in
%\cite{PoHu}.

Let $S$ be a noetherian scheme of finite Krull dimension, and let $\Sm/S$ be the category of smooth
quasi-projective schemes over $S$.  
%\textbf{\textit{
We will assume that $S$ admits an ample family of line bundles
%}} 
so that for any $X$ in $\Sm/S$ there exists an affine bundle $Y \to X$ with $Y$ an
affine scheme.  This condition is used a number of times in this paper, and it was also used in the
proof of the symplectic splitting principle in \cite[Theorem 10.2]{Panin:2010fk}.

The category $\SmOp/S$ has objects $(X,U)$ where $X$ is in $\Sm/S$
and $U \subset X$ is an open subscheme.  A morphism $f \colon (X,U) \to (X',U')$ in $\SmOp/S$ 
is a morphism
$f \colon X \to X'$ of $S$-schemes with $f(U) \subset U'$.  
We often write $X$ in place of $(X,\varnothing)$.

A \emph{bigraded ring cohomology theory} $(A^{*,*},\partial, \times, 1)$ on $\SmOp/S$ is 
a contravariant functor $A^{*,*}$  from $\SmOp/S$ to the category of bigraded abelian groups
which satisfies \'etale excision and $\Aff^{1}$-homotopy invariance and which has  
localization long exact sequences
\[
\cdots \to A^{*,*}(X,U) \to A^{*,*}(X) \to A^{*,*}(U) \xra{\partial} A^{*+1,*}(X,U) \to \cdots.
\]
The $\times$ product is assumed to be functorial, bilinear, associative, and compatible with the
bigrading with a two-sided unit $1$.

In this paper we work mainly with the motivic unstable and stable homotopy categories $H_{\bullet}(S)$
and $SH(S)$.  The former is the homotopy category of a model category $\M_{\bullet}(S)$
of pointed motivic spaces over $S$ with motivic weak equivalences.  There are several versions
of this model category with different underlying categories and with different choices of 
fibrations and cofibrations.
See \cite{Dundas:2003aa,Isaksen:2005aa,Jardine:2000aa,Morel:1999ab,Panin:2009aa,Voevodsky:1998kx,Voevodsky:2007aa}
among other papers.
It is not essential which of these model category structures is used.  
However, we give geometric constructions symmetric $T$- and $T^{\wedge 2}$-spectra using the
the Morel-Voevodsky object $T = \Aff^{1}/(\Aff^{1}-0)$ itself.  So the best adapted model
structures is perhaps the flasque motivic model category of \cite{Isaksen:2005aa} 
which is known to be cellular
(so we can apply Hovey's results \cite{Hovey:2001aa} in the proof of Theorem \ref {T:comparison}) 
but for which $T$ is cofibrant.
 

The category $\M_\bullet(S)$ is equipped with a symmetric monoidal structure
$(\M_\bullet(S), \wedge, S^{0})$. The smash-product is taken sectionwise, and 
$S^{0}$
is the constant simplicial zero-sphere. This smash-product induces a smash-product
on $H_{\bullet}(S)$ such that the natural functor
$\M_\bullet(S) \to H_{\bullet}(S)$
becomes a strict symmetric monoidal functor
$(\M_\bullet(S), \wedge, S^{0}) \to (H_\bullet(S), \wedge, S^{0})$.


We set $T=\Aff^1/(\Aff^1-0)$.
A \emph{$T$-spectrum} $E$ is a sequence
$(E_0,E_1,\dotsc)$ of pointed motivic spaces equipped with a sequence of structure maps
$\sigma_{n}\colon E_n \wedge T \to E_{n+1}$ of pointed motivic spaces.
A morphism of $T$-spectra $E \to E^{\prime}$ is a sequence of maps
$f_n\colon E_n \to E'_n$ of pointed motivic spaces which commute
with the structure maps. The category of $T$-spectra $Sp(\M_\bullet(S),T)$ can be equipped
with a motivic stable model structure as in
\cite{Jardine:2000aa,Voevodsky:1998kx, Voevodsky:2007aa}. Its homotopy category 
is  $SH(S)$.
%
%A \emph{symmetric $T$-spectrum} $E$ given by a sequence of pointed motivic spaces
%$(E_0,E_1,\dotsc)$ with each $E_{n}$ equipped with a left action of the symmetric group 
%$\sym_{n}$ plus structure maps $\sigma_{n} \colon E_{n} \wedge  T \to E_{n+1}$
%such that the induced maps $E_{n} \wedge T^{\wedge m} \to E_{n+m}$ are all
%$\sym_{n} \times \sym_{m}$-equivariant.  A morphism of symmetric $T$-spectra
%is a sequence of $\sym_{n}$-equivariant maps $f_n\colon E_n \to E'_n$ which commute
%with the structure maps.  The category of symmetric $T$-spectra $Sp^{\sym}(\M_\bullet(S),T)$ can be equipped
%with a motivic stable model structure as in
%\cite{Jardine:2000aa}. The functor $u \colon Sp^{\sym}(\M_\bullet(S),T) \to Sp(\M_\bullet(S),T)$  
%sending a symmetric spectrum to its underlying spectrum is a right Quillen equivalence 
%\cite[Theorem 4.31]{Jardine:2000aa} with left adjoint denoted $v$.  The homotopy category
%of $Sp^{\sym}(\M_\bullet(S),T)$ is therefore equivalent to $SH(S)$ via $u$ and $\mathbf{L}v$
%
This category can be equipped with a structure of a symmetric monoidal category
$(SH(S), \wedge, {\boldsymbol 1})$, satisfying the conclusions of
\cite[Theorem 5.6]{Voevodsky:1998kx}.
The unit ${\boldsymbol 1}$ of that
monoidal structure is the $T$-sphere spectrum
$\sphere =(S^{0},T,T\wedge T, \dots )$.




%All technical facts concerning $(SH(k), \wedge, {\bf 1})$
%can be found in
%\cite[Appendix]{Panin:2009aa}.
%Unfortunately, that text has dealings with a different description of $SH(k)$.
%However the resulting homotopy category is as well symmetric monoidal and it is
%naturally equivalent with $SH(k)$ as a symmetric monoidal category.
%Since we use in the present text only results concerning
%$(SH(k), \wedge, {\bf 1})$,
%it makes sense to refer to
%\cite[Appendix]{Panin:2009aa}.

Every $T$-spectrum $E = (E_0,E_1,\dotsc)$ represents a cohomology theory on the category
of pointed motivic spaces $\M_\bullet(S)$.
Namely, let $S^n_s$ and $S^n_t$ be as in
\cite[(16)]{Voevodsky:1998kx}.
Let
$S^{p,q}=S^{p-q}_s \wedge S^q_t$.
We write
$$E^{p,q}(A)=Hom_{SH(S)}(\Sigma^{\infty}_{T} A, E \wedge S^{p,q})$$
as in \cite[\S 6]{Voevodsky:1998kx}.
%Namely, for a pointed motivic space $(A,a)$ set
%\[E^{p,q}(A,a)=\Hom_{\mathrm{SH}(k)}(\Sigma^\infty_{T}(A,a), \Sigma^{p,q}(E))\]
%and
%$E^{\ast,\ast}(A,a)= \oplus_{p,q} E^{p,q}(A,a)$.
%This definition extends to unpointed motivic spaces via the functor $A\mapsto A_+$ which
%adds a disjoint base point.
%That is, for a non-pointed motivic space
%$A$ set
%$E^{p,q}(A)=E^{p,q}(A_+,+)$
%and
%$E^{\ast,\ast}(A)= \bigoplus_{p,q} E^{p,q}(A)$.
There is a canonical element in $E^{2n,n}(E_n)$, denoted as
$$\Sigma^\infty_{T}E_n(-n) \xra{u_n} E.$$
It is represented by
the canonical map $(\ast,\dotsc,\ast,E_n,E_n\wedge T,\dotsc) \to (E_0,E_1,\dotsc,E_n,\dotsc)$
of $T$-spectra.

%As proved in \cite{Voevodsky:1998kx} an arbitrary $T$-spectrum $A$ defines a cohomology theory
%$(A^{*,*}, \partial)$ on the category $\M_\bullet(k)$.
%The differential $\partial$ increases the first index by $1$
%and preserve the second index. Recall that for a non-pointed space $X$
%$A^{*,*}(X):= A^{*,*}(X_+)$. In particular,
%$A^{*,*}(\pt):= A^{*,*}(S^{0}, *)$, where $*$ is the distinguished point of $S^{0}$.

A \emph{$T$-ring spectrum} is a monoid
$(E,\mu,e)$
in
$({SH(S)},\wedge, {\bf 1})$.
The cohomology theory $E^{\ast,\ast}$ defined by a
$T$-ring spectrum is a ring cohomology theory on $\M_{\bullet}(S)$. To see recall
the standard isomorphism 
$S^{i,j} \wedge S^{k,l} \cong S^{i+k,j+l}$
given by the composition
$$(S^{i-j}_s \wedge S^j_t) \wedge (S^{k-l}_s \wedge S^l_t) \cong
(S^{i-j}_s \wedge S^{k-l}_s) \wedge (S^{j}_t \wedge S^{l}_t)
\cong S^{i-j+k-l}_s \wedge S^{j+l}_t.$$
For $X, Y \in \M_\bullet(k)$
let
$\alpha \colon \Sigma^{\infty}(X) \to S^{i,j} \wedge E$
and
$\beta \colon \Sigma^{\infty}(Y) \to S^{k,l} \wedge E$
be elements of
$E^{i,j}(X)$ and $E^{k,l}(Y)$ respectively.
Following Voevodsky
\cite{Voevodsky:1998kx} define
$\alpha \times \beta \in E^{i+k,j+l}(X \wedge Y)$
as the composition
\[\Sigma^{\infty}(X \wedge Y) \cong \Sigma^{\infty}(X) \wedge \Sigma^{\infty}(Y) \xra{\alpha \wedge \beta}
E \wedge S^{i,j} \wedge E \wedge S^{k,l} \cong E \wedge E \wedge S^{i+k,j+l} \xra{id \wedge \mu}
E \wedge S^{i+k,j+l}.
\]
This gives a functorial product which is associative, has a two-sided unit, and takes cofibration sequences to
long exact sequences.


A \emph{commutative} $T$-ring
spectrum is a commutative monoid
$(E,\mu,e)$
in
$(SH(S),\wedge, {\bf 1})$.  To describe the properties
of the associated cohomology theory we make some definitions.


\begin{defn}
\label{epsilon}
Let $in_T \colon T \to T$ be a morphism of pointed motivic spaces induced by the morphism
$\Aff^1 \to \Aff^1$ sending $t \mapsto -t$. One has the equality
$$Hom_{SH(S)}(\pt_{+}, \pt_{+})=Hom_{SH(S)}(T,T).$$
We write
$in$ for $in_T$
regarded as an element of
$Hom_{SH(S)}(\pt_{+}, \pt_{+})$.
For a commutative monoid $(A,m,e)$
set $\epsilon=in^*(e) \in A^{0,0}(\pt_{+})$.
%Warning: our $\epsilon$ is the $-\epsilon$ of Morel \cite[\S 6.1]{Morel:2002aa}.
\end{defn}

\begin{rem}
\label{epsilontimesepsilon}
The morphism $\Aff^2 \to \Aff^2$ which sends
$(t_1,t_2) \mapsto (-t_1,-t_2)$ is $\Aff^1$-homotopic to the identity morphism
because 
$\bigl(
\begin{smallmatrix}
-1 & 0 \\ 0 & -1
\end{smallmatrix}
\bigr)
\in SL_{2}(\mathbb Z)$
is a product of elementary matrices.
Whence we have
$\epsilon \times \epsilon= e \in A^{0,0}(\pt_{+})$.
\end{rem}



%As proved by Morel the cohomology theory $E^{\ast,\ast}$ defined by a commutative
%$T$-ring spectrum is an $\epsilon$-commutative ring cohomology theory on $\M_{\bullet}(k)$
%(see \ref{BO**asringcohomology}).

\begin{defn}
A ring cohomology theory on $\M_{\bullet}(S)$ is 
\emph{$\epsilon$-commutative} if for
$\alpha \in E^{i,j}(X)$, $\beta \in E^{k,l}(Y)$
one has
$\sigma^*_{X,Y}(\alpha \times \beta)=(-1)^{ik}\epsilon^{jl}(\beta \times \alpha) \in 
E^{i+k,j+l}(X \wedge Y)$
where $\sigma_{X,Y} \colon X \wedge Y \to Y \wedge X$ switches the factors.
\end{defn}

Thus $\epsilon$-commutativity is a specific form of bigraded commutativity. 


\begin{thm}[Morel]
\label{BO**asringcohomology}
Let $(E,\mu,e)$ be a commutative monoid in $SH(S)$.
Then the data $(E^{*,*}, \partial, \times, e)$ is an associative and
$\epsilon$-commutative ring cohomology theory on
$\M_\bullet(k)$\textup{;}
$e \in E^{0,0}(S^{0})$ is the two-sided unit of the ring structure.
\end{thm}

Note, that if $(i,j)=(2m,2n)$ or $(k,l)=(2m,2n)$, then
$\sigma^*_{A,B}(\alpha \times \beta)=\beta \times \alpha$.  

%Let $S$ be a noetherian base scheme of finite Krull dimension.  The category $\SmOp/S$ has as
%objects pairs $(X,U)$ with $X$ a smooth scheme of finite type over $S$, and and $U \subset X$
%an open subscheme.  A morphism $f \colon (X,U) \to (Y,V)$ in $\SmOp/S$ is a morphism
%$f \colon X \to Y$ of $S$-schemes satisfying $f(U) \subset V$.  
%We often write $X= (X, \varnothing)$ and $\pt = (S,\varnothing)$.


%\section{The symplectic cobordism \texorpdfstring{$T$}{T}-spectrum}
%
%%Recall that for a $T$-spectrum $A$ there are exact sequences \cite[Corollary 3.5]{Panin:2009aa}
%%%
%%\begin{gather*}
%%0 \to \varprojlim\nolimits^{1} A^{4n-1,2n}(A_{n} \wedge A_{n})
%%\to Hom_{SH(k)}(A \wedge A,A) \to
%%%\to [A \wedge A,A]_{SH(k)} \to
%%\varprojlim A^{4n,2n}(A_{n} \wedge A_{n}) \to 0,
%%\\
%%0 \to \varprojlim\nolimits^{1} A^{6n-1,3n}(A_{n} \wedge A_{n} \wedge A_{n})
%%\to Hom_{SH(k)}(A \wedge A \wedge A,A) \to
%%%\to [A \wedge A \wedge A,A]_{SH(k)} \to
%%\varprojlim A^{6n,3n}(A_{n} \wedge A_{n} \wedge A_{n}) \to 0.
%%\end{gather*}
%%%
%%We will often need kind of objects described in the next definition.
%%\begin{defn}
%%\label{almostringTspectrum}
%%An \emph{almost commutative monoid} in $SH(k)$ is a triple
%%$(A,m,e)$ consisting of a $T$-spectrum $A$,
%%an element
%%$m \in \varprojlim A^{4n,2n}(A_n \wedge A_n)$
%%and an element
%%$e \in A^{0,0}(\pt)$
%%such that the following conditions hold:
%%if
%%$\tilde m \in Hom_{SH(k)}(A \wedge A,A)$
%%is an element which descends to the element 
%%$m \in \varprojlim A^{4n,2n}(A_n \wedge A_n)$, then
%%the following relations hold modulo elements from
%%${\varprojlim}^1$ terms: 
%%%
%%\begin{enumerate}
%%\item
%%$\tilde m \circ (id_{A} \wedge \tilde m)= \tilde m \circ (\tilde m \wedge id_{A} )$.
%%\item
%%$\tilde m = \tilde m \circ \sigma$, where
%%$\sigma \colon A \wedge A \to A \wedge A$ is the morphism in $SH(k)$
%%switching the factors, 
%%\item
%%$\tilde m \circ (e \wedge id_{A})= can_{l} \colon S^{0} \wedge A \xra{\cong} A$.
%%\item
%%$\tilde m \circ (id_{A} \wedge e)= can_{r} \colon A \wedge S^{0}  \xra{\cong} A$.
%%\end{enumerate}
%%\end{defn}
%%
%Recall that a motivic space $X$ is called \emph{small} if $Hom_{SH(k)}(\Sigma_{T}^{\infty}X,{-})$
%commutes with arbitrary small coproducts.   
%For a small motivic space $X$ and a $T$-spectrum $A$ we have by 
%\cite[Theorem 5.2]{Voevodsky:1998kx} or 
%\cite[Lemma A.34(1)]{Panin:2009aa}
%%
%\begin{equation}
%A(X) = Hom_{SH(k)}(\Sigma_{T}^{\infty}X, A) = 
%\colim\nolimits_{n} Hom_{H_{\bullet}^{m}(k)}(X \wedge T^{\wedge n}, A_{n}).
%\end{equation}
%%
%%
%%This is because the products of cohomology groups of small motivic spaces only depend on 
%%$\bar{m} \in \varprojlim \mathbb{BO}^{4n,2n}(\mathbb{K}O^{[n]} \wedge \mathbb{K}O^{[n]})$
%%and not on the choice of lifting $m \in \mathbb{BO}^{0,0}(\mathbb{BO} \wedge  \mathbb{BO})$.
%%This is a consequence of the fact that for small motivic spaces $A$ and a
%%$T$-spectrum $E$ we have by \cite[Theorem 5.2]{Voevodsky:1998kx} or 
%%\cite[Lemma A.34(1)]{Panin:2009aa}
%%%
%%\begin{equation}
%%E(A) = Hom_{SH(S)}(\Sigma_{T}^{\infty}A, E) = 
%%\colim_{n} Hom_{H_{\bullet}^{m}(S)}(A \wedge T^{\wedge n}, E_{n}).
%%\end{equation}
%%%
%%So the product on cohomology groups of small motivic spaces induced by 
%%$m \colon E \wedge E \to E$ actually only depends on the image
%%$\bar m \in \varprojlim E^{4n,2n}(E_{n} \wedge E_{n})$.
%%
%Let $\M_{\bullet}^{\ft}(k) \subset \M_{\bullet}(k)$
%be the full subcategory of small motivic spaces.
%
%\begin{defn}
%\label{BO**BO**toBO**}
%Let $X, Y \in \M_\bullet^{\ft}(k)$.
%Let
%$\alpha \in A^{i,j}(X)$
%and
%$\beta \in A^{k,l}(Y)$ be elements induced respectively by the morphisms 
%$a \colon X \wedge T^{\wedge n} \to S^{i,j} \wedge A_{n}$ and
%$b \colon Y \wedge T^{\wedge n} \to S^{k,l} \wedge A_{n}$ in $H_{\bullet}^{m}(k)$.
%Define $\alpha \times \beta \in A^{i+k,j+l}(X \wedge Y)$
%by applying $\Sigma_{T}^{\infty}$ to the composition
%\[
%X \wedge Y \wedge T^{\wedge 2n} \cong X \wedge T^{\wedge n} \wedge Y \wedge T^{\wedge n}
%\xra{a \wedge b} S^{i,j} \wedge A_{n} \wedge S^{k,l} \wedge A_{n} \cong 
%S^{i+k,j+l}\wedge A_{n} \wedge A_{n}
%\]
%and then composing with the morphism 
%$id \wedge m \colon S^{i+k,j+l} \wedge \Sigma_{T}^{\infty} (A_{n} \wedge A_{n}) \to S^{i+k,j+l} \wedge A[2n]$ in $SH(k)$.  This gives a member of
%$Hom_{SH(k)}(\Sigma_{T}^{\infty}(X \wedge Y)[2n],S^{i+k,j+l} \wedge A[2n])$ whose translate
%is $\alpha \times \beta$.
%\end{defn}
%
%For an almost commutative monoid induced by a commutative monoid the two definitions 
%of the product coincide for small motivic spaces.
%
%\begin{lem}
%\label{BO**asringcohomology.2}
%Let $(A,m,e)$ be an almost commutative monoid in $SH(k)$.
%Then  $(A^{*,*}, \partial, \times, e)$ is an associative and
%$\epsilon$-commutative ring cohomology theory on
%$\M^{\ft}_\bullet(k)$.
%\end{lem}


\section{Commutative \texorpdfstring{$T$- and $T^{\wedge 2}$}{T- and T\^{ }2}-monoids}

We compare the categories of symmetric $T$-spectra and symmetric $T^{\wedge 2}$-spectra.
Recall the definition for $K = T$ or $K = T^{\wedge 2}$.  

\begin{defn}
A \emph{symmetric $K$-spectrum} $E$ is a sequence of pointed spaces $(E_{0},E_{1},E_{2},\dots)$
with each $E_{n}$ equipped with an action of the symmetric group $\sym_{n} \times E_{n} \to E_{n}$
and with a morphism $\sigma_{n} \colon E_{n} \wedge K \to E_{n+1}$ such that the induced maps
$E_{n} \wedge K^{\wedge m} \to E_{n+m}$ are $(\sym_{n} \times \sym_{m})$-equivariant
for all $n$ and $m$.
\end{defn}

The categories of symmetric $T$- and $T^{\wedge 2}$-spectra both have a symmetric monoidal
product $\wedge$.  They are symmetric monoidal model categories for the stable model structure
\cite{Hovey:2001aa,Jardine:2000aa}.  

\begin{thm}
\label{T:comparison}
The homotopy categories of $Sp^{\sym}(\M_{\bullet}(S),T)$ and of 
$Sp^{\sym}(\M_{\bullet}(S),T^{\wedge 2})$ are equivalent symmetric monoidal categories.
\end{thm}

\begin{proof}
The proof of this theorem is essentially the same as that given for topological $S^{1}$- and $S^{2}$-spectra in 
\cite[Theorem A.44]{Panin:2009aa}.  The inclusion 
$Sp^{\sym}(\M_{\bullet}(S),T) \to Sp^{\sym}(Sp^{\sym}(\M_{\bullet}(S),T),T^{\wedge 2})$
is a Quillen equivalence by \cite[Theorem 9.1]{Hovey:2001aa} because ${-}\wedge T^{\wedge 2}$
is a Quillen self-equivalence of $Sp^{\sym}(\M_{\bullet}(S),T)$.  Similarly the inclusion  
$Sp^{\sym}(\M_{\bullet}(S),T^{\wedge 2}) \to Sp^{\sym}(Sp^{\sym}(\M_{\bullet}(S),T^{\wedge 2}),T)$
is a Quillen equivalence.  The two categories of symmetric 
bispectra are isomorphic with the identical stable model 
structure by arguments like those used in the proof of \cite[Theorem 10.1]{Hovey:2001aa}.  
Hovey's work requires that the model structure have certain properties, but in the flasque model structure
\cite{Isaksen:2005aa} these properties hold, and $T$ and $T^{\wedge 2}$ are cofibrant.

The symmetric monoidal structures are the same
because (i) the inclusions of the categories of symmetric spectra in the categories of symmetric 
bispectra are symmetric monoidal
functors like any inclusion $\Sigma_{K}^{\infty} \colon \mathcal C \to Sp^{\sym}(\mathcal C,K)$, and 
(ii) the symmetric monoidal structures on the two isomorphic  categories of symmetric  bispectra 
are the same.
\end{proof}

For natural numbers $m,n$ we denote by $c_{m,n} \in \sym_{m+n}$ 
the \emph{$(m,n)$-shuffle permutation}.
It acts by $c_{m,n}(i) = i+n$ for $1 \leq i \leq m$ and $c_{m,n}(i) = i-m$ for $m+1 \leq i \leq m+n$.


\begin{defn}
\label{D:T.monoid}
A \emph{commutative $K$-monoid} $E$ in $\M_{\bullet}(S)$ is a sequence of 
pointed motivic spaces
$(E_{0},E_{1},E_{2},\dots)$ with each space equipped with an action $\sym_{n} \times E_{n} \to E_{n}$
of the symmetric group, plus morphisms
%
\begin{equation}
\label{E:T.monoid.structure}
\begin{aligned}
e_{0} \colon & \boldsymbol{1}_{\M_{\bullet}(S)} \to E_{0},
\\
e_{1} \colon & K \to E_{1}
\\
\mu_{mn} \colon & E_{m} \wedge E_{n} \to E_{m+n}
\end{aligned}
\end{equation}
%
in $\M_{\bullet}(S)$ such that each $\mu_{mn}$ is $(\sym_{m} \times \sym_{n})$-equivariant and such that the compositions
%
\begin{equation}
\label{E:T.unit}
\begin{gathered}
E_{n} \xra{\cong} E_{n} \wedge \boldsymbol{1}_{M_{\bullet}(S)} \xra{1 \wedge e_{0}} E_{n}\wedge E_{0}
\xra{\mu_{n0}} E_{n}
\\
E_{n} \xra{\cong} \boldsymbol{1}_{M_{\bullet}(S)} \wedge E_{n}  \xra{e_{0} \wedge 1} E_{0}\wedge E_{n}
\xra{\mu_{0n}} E_{n}
\end{gathered}
\end{equation}
%
are the identity maps, and the diagrams
\begin{equation}
%\label{E:associativity}
\vcenter{
\xymatrix @M=5pt @C=30pt {
E_{\ell } \wedge E_{m} \wedge E_{n} \ar[r]^-{\mu_{\ell m}\wedge 1} \ar[d]_-{1 \wedge \mu_{mn}}
&
E_{\ell +m} \wedge E_{n} \ar[d]^-{\mu_{\ell +m,n}}
\\
E_{\ell } \wedge E_{m+n} \ar[r]^-{\mu_{\ell ,m+n}}
&
E_{\ell +m+n}
}}
\qquad \quad
%\end{equation}
%%
%commute in $H_{\bullet}(S)$, and the diagrams
%%
%\begin{equation}
\label{E:commutativity}
\vcenter{
\xymatrix @M=5pt @C=35pt {
E_{m} \wedge E_{n} 
\ar[d]_-{\text{switch}} 
\ar[r]^-{\mu_{mn}}
%\ar@{}[rd]|-{h}
& 
E_{m+n} 
\ar[d]_-{\cong}^-{c_{m,n}}
\\
E_{n} \wedge E_{m} 
\ar[r]^-{\mu_{nm}}
&
E_{n+m}
}}
\end{equation}
%
commute in $\M_{\bullet}(S)$ with $c_{m,n}$ the isomorphism given by the action
of the $(m,n)$-shuffle permutation.
\end{defn}

\begin{thm}
Let $E$ be a commutative $K$-monoid.  Define maps $\sigma_{n}$ as the compositions
%
\begin{equation}
\label{E:T.bonding}
\sigma_{n} \colon E_{n} \wedge K \xra{1 \wedge e_{1}} E_{n} \wedge E_{1} \xra{\mu_{n1}} E_{n+1}.
\end{equation}
%
Then the spaces $(E_{0},E_{1},E_{2},\dots)$ equipped with the actions $\sym_{n} \times E_{n} \to E_{n}$
and the bonding maps $\sigma_{n}$ form a symmetric $K$-spectrum $E$.
Moreover, the morphisms
$\mu \colon E \wedge E \to E$ induced by the $\mu_{mn}$
and $e \colon \Sigma_{K}^{\infty} \boldsymbol{1}_{\M_{\bullet}(S)}\to E$ 
composed of the maps $e_{n} \colon K^{\wedge n} \to E_{n}$ induced by $e_{0}$, $e_{1}$ and
the $\mu_{mn}$ make $(E,\mu,e)$
a commutative monoid in $Sp^{\sym}(\M_{\bullet}(S),K)$.
\end{thm}

\begin{proof}[Sketch of proof]
To show that $E$ is a symmetric $K$-spectrum one has to verify that the induced maps
$E_{n} \wedge K^{\wedge j} \to E_{n+j}$ are $(\sym_{n} \times \sym_{j})$-equivariant.  
To show that the maps $\mu_{mn}$ define a morphism $E \wedge E \to E$
one has to verify that they are $K$-linear and $K$-bilinear in the sense of 
\cite[(4.6)--(4.7)]{Jardine:2000aa}.  One has to verify that each $e_{n}$ is $\sym_{n}$-equivariant.
Finally, one has to verify the commutative monoid axioms.  
All the verifications are formal, straightforward and left to the reader.
\end{proof}

\section{The symmetric \texorpdfstring{$T^{\wedge 2}$}{T\^{ }2}-spectrum 
\texorpdfstring{$\MSL$}{MSL}}


We construct a commutative $T$-monoid $\MSL$.  
Each space $\MSL_{n}$ comes equipped
with an action of $GL_{n}$ such that the multiplication maps $\mu_{mn}$ of the monoid
structure are $(GL_{m} \times GL_{n})$-equivariant.  We then get actions of the $\sym_{n}$
from the embeddings $\sym_{n} \to GL_{n}$ given by permutation matrices.
The need for an action of $GL_{n}$ with fixed points --- necessary for the proper definition of the unit maps
---
rather than merely of $SL_{n}$ is the delicate part of the construction.

We begin by reviewing the construction of $\MGL$ originally done in
\cite[\S 6.3]{Voevodsky:1998kx} and of its monoid structure given in 
\cite{Panin:2008db,Vezzosi:2001aa}.  

For each integer $n \geq 0$ let $\Gamma_{n} = \OO_{S}^{\oplus n}$ be the trivial rank-$n$ vector bundle.
For each integer $p \geq 1$ let $Gr(n,np) =  Gr(n,\Gamma_{n}^{\oplus p})$.  Let
$\TGL{n,np} \to Gr(n,np)$ be the tautological
subbundle.   The inclusions 
$(1,0) \colon \Gamma_{n}^{\oplus p} \to \Gamma_{n}^{\oplus p} \oplus \Gamma_{n}
= \Gamma_{n}^{\oplus p+1}$
%\[
%\Gamma_{n}^{\oplus p} \xra{(1,0)} \Gamma_{n}^{\oplus p} \oplus \Gamma_{n}
%= \Gamma_{n}^{\oplus p+1}
%\]
induce closed embeddings $Gr(n,np) \hra Gr(n,np+n)$ and monomorphisms
%\[
%\xymatrix @M=5pt @C=30pt {
%\TGL{n,np} \ar[r] \ar[d] 
%& \TGL{n+1,np+p} \ar[d]
%\\
%Gr(n,np) \ar[r]
%& Gr(n+1,np+p)
%}
%\]
%and therefore morphisms 
$\Th \TGL{n,np} \to \Th \TGL{n,np+n}$
of Thom spaces.  We set
%
\begin{gather*}
BGL_{n} = \colim\nolimits_{p\in \NN} Gr(n,np),
\\
\TGL{n,n\infty} = \colim\nolimits_{p\in \NN} \TGL{n,np},
\\
\MGL_{n} = \colim\nolimits_{p\in \NN} \Th \TGL{n,np}.
\end{gather*}
%
The diagonal action of $GL_{n} = GL(\Gamma_{n})$ on each $Gr(n,np) = Gr(n,\Gamma_{n}^{\oplus p})$
is compatible with the inclusions over increasing $p$.  Moreover, the $\TGL{n,np}$ are $GL_{n}$-equivariant vector bundles.
This induces actions
\[
\sym_{n} \times \MGL_{n} \subset GL_{n} \times \MGL_{n} \to \MGL_{n}.
\]
Concatenation of bases induces isomorphisms $\Gamma_{m} \oplus \Gamma_{n} = \Gamma_{m+n}$
which induce $(GL_{m} \times GL_{n})$-equivariant maps
%
\begin{equation*}
%\label{E:sum.GL}
\oplus \colon 
Gr(m,mp) \times Gr(n,np) \to Gr(m+n,mp+np)
\end{equation*}
%
and therefore $(GL_{m} \times GL_{n})$-equivariant maps
%
\begin{gather*}
%\beta_{mn} \colon 
\oplus \colon BGL_{m} \times BGL_{n} \to BGL_{m+n},
\\
\mu_{mn}^{GL} \colon \MGL_{m} \wedge \MGL_{n} \to \MGL_{m+n}.
\end{gather*}
%
Finally each $Gr(n,np)$ is pointed by the point corresponding to the trivial rank-$n$ subbundle 
\[
\Gamma_{n} \xra{(1,0,\dots,0)} \Gamma_{n}^{\oplus p}.
\]
In the colimit this gives a $S$-valued point $x_{n}$ of $BGL_{n}$, which is fixed by the action of $GL_{n}$.  
The Thom space of the fiber
of $\mathcal T(n,n\infty)$ over $x_{n}$ is $\Gamma_{n} \cong \Aff^{n}$, and the inclusion
$x_{n} \hra BGL_{n}$ induces a map of Thom spaces
\[
e_{n}^{GL} \colon T^{\wedge n} \to \MGL_{n}.
\]
%The spaces $\MGL_{n}$ together with the actions $\sym_{n} \times \MGL_{n}$, the maps
%$e_{0}$ and $e_{1}$ and the maps $\mu_{mn}$ satisfy the axioms of a commutative 
%$T$-monoid.

\begin{defn}
The \emph{algebraic cobordism spectrum} $\MGL$ is the commutative monoid in the category of
symmetric $T$-spectra associated to the 
commutative $T$-monoid composed of
the spaces $\MGL_{n}$, the actions $\sym_{n} \times \MGL_{n} \to \MGL_{n}$, the maps
$e_{0}^{GL} \colon \pt_{+} \to \MGL_{0}$ and $e_{1}^{GL} \colon T \to \MGL_{1}$ 
and the maps $\mu_{mn}^{GL} \colon \MGL_{m} \wedge \MGL_{n} \to \MGL_{m+n}$.
\end{defn}

We now move on to defining $\MSL$. We begin with the spaces.
For $n = 0$ we have $SL_{0} = GL_{0} = \{1\}$.  So we set $BSL_{0} = \pt$.  The Thom space of
a zero vector bundle over a scheme $X$ is the externally pointed space $X_{+}$.  
So we set $\MSL_{0} = \pt_{+}$.

Now suppose $n > 0$.
Over each $Gr(n,np)$ there is the line bundle $\OO_{Gr(n,np)}(-1) = \det \TGL{n,np}$.
Removing the zero section gives a smooth scheme
\[
\GrSL(n,np) = \OO_{Gr(n,np)}(-1) - Gr(n,np).
\]
The projection
\[
\pi = \pi_{n,np} \colon \GrSL(n,np)  \to Gr(n,np)
\]
is a principal $\GG_{m}$-bundle.
Write 
\[
\TSL{n,np}  = \pi^{*} \TGL{n,np}.
\]
The inclusion $\GrSL(n,np) \hra \OO_{Gr(n,np)}(-1)$ and the cartesian diagram
\[
\xymatrix @M=5pt {
\pi^{*}\OO_{Gr(n,np)}(-1) \ar[r] \ar[d] \ar@{}[rd]|{\Box}
& \OO_{Gr(n,np)}(-1) \ar[d]
\\
\GrSL(n,np) \ar[r]_-{\pi}
& Gr(n,np)
}
\]
gives a nowhere vanishing section of $\pi^{*}\OO_{Gr(n,np)}(-1) = \det \TSL{n,np}$.
The corresponding isomorphism 
$\lambda_{n,np} \colon \OO_{\GrSL(n,np)} \cong \det \TSL{n,np}$
makes $(\TSL{n,np}, \lambda_{n,np})$ the \emph{tautological special linear bundle}
over $\GrSL(n,np)$.

We set
%
\begin{gather*}
BSL_{n} = \colim\nolimits_{p\in \NN} \GrSL(n,np),
\\
\TSL{n,n\infty} = \colim\nolimits_{p\in \NN} \TSL{n,np},
\\
\MSL_{n} = \colim\nolimits_{p\in \NN} \Th \TSL{n,np}.
\end{gather*}

We next define the multiplication maps.  Morphisms of $S$-schemes $X \to \GrSL(n,np)$ are in bijection
with pairs $(f,\lambda)$ with $f \colon X \to Gr(n,np)$ a morphism and 
$\lambda \colon \OO_{X} \cong \det f^{*} \TGL{n,np}$ an isomorphism.
There are unique maps
\[
(\oplus, \otimes) \colon \GrSL(m,mp) \times \GrSL(n,np) \to \GrSL(m+n,mp+np)
\]
corresponding to the morphisms of representable functors
\[
\begin{array}{ccc}
Hom(X,\GrSL(m,mp)) \times Hom(X,\GrSL(n,np)) & \lra & \Hom(X,\GrSL(m+n,mp+np))
\\
((f,\lambda),(g,\lambda_{1})) & \longmapsto & (f \oplus g, \lambda \otimes \lambda_{1}).
\end{array}
\]
They induce maps
%
\begin{gather*}
(\oplus,\otimes) \colon BSL_{m} \times BSL_{n} \to BSL_{m+n}
\\
\mu^{SL}_{mn} \colon \MSL_{m} \wedge \MSL_{n} \to \MSL_{m+n}.
\end{gather*}

We now discuss the group actions.   
%The $\sym_{n}$-action cannot be induced by an action of $SL_{n}$ 
%because $\sym_{n} \not\subset SL_{n}$.  Therefore we will give a $GL_{n}$-action.  
%But there are complications
%when dealing with $GL_{n}$-actions on $BSL_{n}$.
%
Since $\TGL{n,np}$ is a $GL_{n}$-equivariant bundle over $Gr(n,np)$,
there is an induced action of $GL_{n}$ on the complement of the zero section of the 
determinant line bundle.
This is an action $GL_{n} \times \GrSL(n,np) \to \GrSL(n,np)$.  In the colimit this gives an action
\(
GL_{n} \times BSL_{n} \to BSL_{n}.
\)
But there is a problem.



%The projection $BSL_{n} \to BGL_{n}$ is $GL_{n}$-equivariant.  
%Nevertheless, \textbf{\textit{this is the wrong
%action}} of $GL_{n}$ and of $\sym_{n}$ on the $\GrSL(n,np)$ and on $BSL_{n}$ for the following reason.
%Recall that when one has an equivariant vector bundle, the group acts linearly on the fibers over the
%fixed points.  
The unit maps $e_{n}^{GL} \colon T^{\wedge n} \to \MGL_{n}$ were defined using points
$x_{n} \colon \pt \to BGL_{n}$ which were fixed under the action of $GL_{n}$.  
To define unit maps for a $T$-monoid $\MSL$ we need fixed points for the action of at least
$\sym_{n}$ on $BSL_{n}$,
preferably lying over $x_{n}$.  
We have a cartesian diagram
\[
\xymatrix @M=5pt {
\GG_{m} \ar[r] \ar[d] \ar@{}[rd]|{\Box}
&
BSL_{n} \ar[d]
\\
\pt \ar[r]_-{x_{n}}
&
BGL_{n}.
}
\]
The action of $GL_{n} = GL(\Gamma_{n})$ on the fiber $\Gamma_{n}$ of $\TGL{n,n\infty}$
over the fixed point $x_{n}$ is the standard representation of $GL_{n}$.  So the induced action on the fiber
$\GG_{m}$ over $x_{n}$ is $g \cdot t = \det(g)t$.  Thus there are fixed points 
for the action of the alternating group $\mathfrak A_{n} \subset SL_{n}$ on $BSL_{n}$
lying over the fixed point $x_{n} \in BGL_{n}(\pt)$ used to define the unit maps on $\MGL_{n}$
but not for the action of $\sym_{n} \subset GL_{n}$ (except in characteristic $2$).

So we use the embedding $\sym_{n} \subset Sp_{2n} \subset SL_{2n}$ which sends $\sigma \in \sym_{n}$
to the permutation matrix associated to $\bar\sigma \in \sym_{2n}$ where we have
$\bar\sigma(2i-1) = 2\sigma(i)-1$ and $\bar\sigma(2i) = 2\sigma(i)$.  This gives us an action
$\sym_{n} \times BSL_{2n} \to BSL_{2n}$ which fixes pointwise the fiber over $x_{n}$.


Therefore we define the spaces of the commutative $T^{\wedge 2}$-monoid $\MSL$ to be the
$\MSL_{2n}$.   Each is equipped with the action of $\sym_{n} \times \MSL_{2n} \to \MSL_{2n}$ 
induced by the action of $SL_{2n}$.

%We therefore need to modify the action of $GL_{n}$ on $BSL_{n}$.  We do this by defining
%a second action $\gamma_{2}$ of $GL_{n}$.  This action is trivial on $BGL_{n}$ and
%$\TGL{n,n\infty}$.
%The action by $\gamma_{2}$ of $g \in GL_{n}$ on the fibers $\GG_{m}$ of $BSL_{n} \to BGL_{n}$
%is by multiplication by $\det(g)^{-1}$.  The two actions $\gamma_{1}$ and $\gamma_{2}$
%commute.  So we may define a third action
%\[
%\gamma_{3} = \gamma_{1}\gamma_{2} \colon GL_{n} \times BSL_{n} \to BSL_{n}.
%\]
%It extends to a compatible action on the vector bundle $\TSL{n,n\infty}$.
%
%\begin{lem}
%The locus of fixed points for the action $\gamma_{3}$ of $GL_{n}$
%on $BSL_{n}$ is the inverse image of the locus of fixed points 
%for the action of $GL_{n}$ on $BGL_{n}$.
%\end{lem}
%
%The multiplication maps $\MSL_{m} \wedge \MSL_{n} \to \MSL_{m+n}$
%are $(GL_{m} \times GL_{n})$-equivariant for all three actions $\gamma_{1}$, $\gamma_{2}$
%and $\gamma_{3}$.

We now define the unit maps.  
Points $\pt \to BSL_{n}$ lifting the point $x_{n} \colon \pt \to BGL_{n}$ are in bijection
with isomorphisms 
$\lambda \colon \OO_{S} \cong \det x_{n}^{*} \TGL{n,n\infty} = \Lambda^{n} \Gamma_{n}$.
Let $f_{1},\dots,f_{n}$ be the standard basis of $\Gamma_{n} = \OO_{S}^{\oplus n}$.  
We let $y_{n} \colon \pt \to BSL_{n}$ be the lifting of $x_{n}$ corresponding to
$\lambda = f_{1} \wedge \cdots \wedge f_{n}$.  
%By the lemma $y_{n}$ is a fixed point of the action $\gamma_{3}$ of $GL_{n}$ on $BSL_{n}$.  
The fiber of $\TSL{n,n\infty}$ over $y_{n}$
is $\Gamma_{n} \cong \Aff^{n}$, and we let
\[
e_{n}^{SL} \colon T^{\wedge n} \to \MSL_{n}
\]
be the map of Thom spaces induced by $y_{n}$.  It is $SL_{n}$-equivariant.
Note that $e_{0}^{SL} \colon \pt_{+} \to \MSL_{0} = \pt_{+}$ is the identity.

Having identified the components of the structure $\MSL$, we have to assemble them.
It appears as if $\MSL$ is a 
%\textit{\textbf{
commutative monoid in the category of alternating $T$-spectra.
%}}  
We do not know how to work in that category.  But there is underlying structure.

\begin{defn}
The \emph{algebraic special linear cobordism spectrum} $\MSL$ refers to three
related objects.

(a)
The commutative monoid in the category of symmetric $T^{\wedge 2}$-spectra associated 
to the commutative $T^{\wedge 2}$-monoid composed of 
the spaces $\MSL_{2n}$, the actions $\sym_{n} \times \MSL_{2n} \to \MSL_{2n}$, the maps 
$e_{0}^{SL} \colon \pt_{+} \to \MSL_{0}$ and $e_{2}^{SL} \colon T^{\wedge 2} \to \MSL_{2}$ 
and the maps $\mu_{2m,2n}^{SL} \colon \MSL_{2m} \wedge \MSL_{2n} \to \MSL_{2m+2n}$.

(b) The $T$-spectrum with spaces $\MSL_{n}$, bonding maps
$\MSL_{n} \wedge T \to \MSL_{n} \wedge \MSL_{1} \to \MSL_{n+1}$ induced by $e_{1}^{SL}$
and $\mu_{n,1}^{SL}$, equipped with the morphism of $T$-spectra 
$e \colon \Sigma_{T}^{\infty}\pt_{+} \to \MSL$
and the structural maps $\mu_{mn}^{SL}$.

(c) Their common underlying $T^{\wedge 2}$-spectrum.
\end{defn}



The properties of the commutative monoid structure that we require are given in the following theorem.

\begin{thm}
\label{T:MSL.structure}
The $(\MSL,\mu^{SL},e^{SL})$ is a commutative monoid in $SH(S)$, and the canonical maps
$u_{n} \colon \Sigma_{T}^{\infty} \MSL_{n}(-n) \to \MSL$ and the $\mu_{mn}^{SL}$ make
the following diagram commute for all $m$ and $n$
%
\begin{equation}
\label{E:MSLm.wedge.MSLn}
\vcenter{
\xymatrix @M=5pt @C=50pt {
\Sigma_{T}^{\infty}\MSL_{m}(-m) \wedge \Sigma_{T}^{\infty}\MSL_{n}(-n) 
\ar[r]^-{\Sigma_{T}^{\infty}\mu_{mn}^{SL}} \ar[d]_-{u_{m} \wedge u_{n}}
& \Sigma_{T}^{\infty}\MSL_{m+n}(-m-n) \ar[d]^-{u_{m+n}}
\\
\MSL \wedge \MSL \ar[r]^-{\mu^{SL}}
&
\MSL.
}}
\end{equation}
\end{thm}

\begin{proof}
A commutative monoid in $Sp^{\sym}(\M_{\bullet}(S),T^{\wedge 2})$ gives a commutative monoid
in $SH(S)$ by Theorem \ref{T:comparison}.  When $m$ and $n$ are even, the diagram 
in $Sp^{\sym}(\M_{\bullet}(S),T^{\wedge 2})$ corresponding to \eqref{E:MSLm.wedge.MSLn}
commutes by formal arguments.  When say $m$ is even and $n$ is odd, the diagram
\[
\xymatrix @M=5pt @C=50pt {
\Sigma_{T}^{\infty}\MSL_{m}(-m) \wedge \Sigma_{T}^{\infty}\MSL_{n}(-n) 
\wedge \Sigma_{T}^{\infty}T
\ar[r]^-{\Sigma_{T}^{\infty}\mu_{mn}^{SL}\wedge 1} \ar[d]_-{1 \wedge \Sigma_{T}^{\infty}\sigma_{n}}
& \Sigma_{T}^{\infty}\MSL_{m+n}(-m-n)  \wedge \Sigma_{T}^{\infty}T
\ar[d]^-{\Sigma_{T}^{\infty}\sigma_{m+1}}
\\
\Sigma_{T}^{\infty}\MSL_{m}(-m) \wedge \Sigma_{T}^{\infty}\MSL_{n+1}(-n) 
\ar[r]^-{\Sigma_{T}^{\infty}\mu_{m,n+1}^{SL}} \ar[d]_-{u_{m} \wedge u_{n+1}}
& \Sigma_{T}^{\infty}\MSL_{m+n+1}(-m-n) \ar[d]^-{u_{m+n+1}}
\\
\MSL \wedge \MSL(1) \ar[r]^-{\mu^{SL}}
&
\MSL(1)
}
\]
commutes because $m$ and $n+1$ are even. One may desuspend.  The other cases are similar.
\end{proof}


%The projection maps $BSL_{2n} \to BGL_{2n}$ induce maps $\MSL_{2n} \to \MGL_{2n}$
%which are compatible with the actions and the structural maps. 
%They therefore induce morphisms
%\(
%\MSL \to \MGL
%\)
%of commutative monoids in the category of symmetric $T^{\wedge 2}$-spectra.

\section{Special linear orientations}


%In \cite[Definitions 2.1 and 2.13]{Panin:2003rz} a notion of a \emph{ring cohomology theory
%$(A,\partial, \times, 1)$ on $\SmOp/S$} was axiomatized.  It is a contravariant functor
%$A \colon \SmOp/S^{op} \to \operatorname{\mathcal{A}\mathnormal{b}}$ satisfying \'etale 
%excision and $\Aff^{1}$-homotopy invariance, 
%with long exact sequences
%\[
%\cdots \to A(X,U) \to A(X)\to A(U) \xra{\partial_{X,U}} A(X,U) \to \cdots
%\]
%and with an exterior product 
%\[
%\times \colon A(X,U) \times A(Z,W) \to A(X \times Z, X \times W \cup U \times Z)
%\]
%which is functorial, bilinear and associative with a two-sided unit $1 \in A(\pt)$, and with
%$\partial_{X \times Z, U\times Z}(a \times b) = \partial_{X,U} a \times b$ for $a \in A(U)$ and $b \in A(Z)$.
%
%A ring cohomology theory is \emph{bigraded} if the groups $A(X,U)$ are bigraded, the pullback
%maps $f^{*}$ and the product $\times$ are homogeneous of bidegree $(0,0)$, the
%connecting maps $\partial_{X,U}$ are homogeneous of bidegree $(1,0)$, and it has $1 \in A^{0,0}(\pt)$.
%A bigraded ring cohomology
%theory is \emph{$\epsilon$-commutative} if there is an $\epsilon \in A^{0,0}(\pt)$ with 
%$\epsilon \times \epsilon = 1$ such that for any $a \in A^{p,q}(X,U)$ and $b \in A^{p',q'}(Z,W)$ we have
%$b \times a = (-1)^{pp'}\epsilon^{qq'} \sigma^{*}(b \times a)$ where $\sigma$ is the switch map.
%
%A commutative monoid $(E,\mu,e)$ in $SH(S)$ determines an $\epsilon$-commutative bigraded 
%cohomology theory on $\SmOp/S$ with 
%$E^{p,q}(X,U) = \Hom_{SH(S)}(\Sigma_{T}^{\infty} (X_{+}/U_{+}), S^{p,q} \wedge E)$

We now investigate the relationship between special linear orientations 
on a ring cohomology theory $E$, as defined in 
\cite[Definition 3.1]{Panin:2010aa} and homomorphisms $\varphi \colon \MSL \to A$ 
of commutative monoids in $SH(S)$.

A \emph{special linear vector bundle} over $X$ is a pair $(E,\lambda)$ with $E \to X$ a vector bundle
and $\lambda \colon \OO_{X} \cong \det E$ an isomorphism of line bundles.  
An \emph{isomorphism} $\phi \colon (E,\lambda) \cong (E',\lambda')$ of special linear vector bundles 
is an isomorphism $\phi \colon E \cong E'$ of vector bundles such that 
$(\det \phi) \circ \lambda = \lambda'$.

\begin{defn}
\label{D:SL.orientation}
A \emph{special linear orientation} on a bigraded $\epsilon$-commutative 
ring cohomology theory $A^{*,*}$ on $\SmOp/S$
is a rule which assigns to every special linear vector bundle $(E, \lambda)$ of rank $n$ over an $X$ in
$\Sm/S$ a class $\thom(E,\lambda) \in A^{2n,n}(E,E-X)$
satisfying the following conditions:

%
\begin{enumerate}
\item For an isomorphism $f \colon (E,\lambda) \cong (E_{1},\lambda_{1})$ 
we have $\thom(E,\lambda) = f^{*}\thom(E_{1},\lambda_{1})$.

\item For $u \colon Y \to X$ 
we have $u^{*}\thom(E,\lambda) = \thom(u^{*}(E,\lambda))$ in $A^{2n,n}(u^{*}E,u^{*}E - Y)$.

\item The maps ${-} \cup \thom(E,\lambda) \colon A^{*,*}(X) \to A^{*+2n,*+n}(E,E-X)$ are isomorphisms.

\item We have 
\[
\thom (E_{1} \oplus E_{2}, \lambda_{1} \otimes \lambda_{2})
= q_{1}^{*}\thom(E_{1},\lambda_{1}) \cup q_{2}^{*}\thom(E_{2},\lambda_{2}),
\] 
where $q_{1},q_{2}$ are the projections from $E_{1} \oplus E_{2}$ 
onto its summands.  Moreover, for the zero bundle $\boldsymbol 0 \to \pt$ we have
$\thom(\boldsymbol{0}) = 1_{A} \in A^{0,0}(\pt)$.
\end{enumerate}
%
The class $\thom(E,\lambda)$ is the \emph{Thom class} of the special linear bundle, and
$e(E,\lambda) = z^{*} \thom(E,\lambda) \in A^{2n,n}(X)$ is its \emph{Euler class}.
\end{defn}

This definition is analogous to the Thom classes theory version of the definition of an orientation
\cite[Definition 3.32]{Panin:2003rz}.

For any $n$ the functor ${-}\wedge T^{\wedge n} \colon SH(S) \to SH(S)$ is a self-equivalence.  
So it induces isomorphisms
\[
{-}\wedge T^{\wedge n}  \colon 
Hom_{SH(S)}(X,A \wedge S^{p,q} ) \xra{\cong}
Hom_{SH(S)}( X \wedge T^{\wedge n},  A \wedge S^{p,q} \wedge T^{\wedge n})
\]
for any $X$ and $(p,q)$ and 
any cohomology theory on $\Sm/S$ defined by a commutative monoid $(A,\mu,e)$ in $SH(S)$.
We also write these isomorphisms as
\[
\Sigma_{T}^{n} \colon A^{p,q}(X)  \xra{\cong}
A^{p+2n,q+n}(X \times \Aff^{n}, X \times (\Aff^{n}-0))
\]
This isomorphism coincides with ${-} \times \Sigma_{T}^{n}1_{A}$.  
Thus $A^{*,*}$ automatically has Thom classes for trivial bundles: the pullbacks of $\Sigma_{T}^{n}1_{A}$.

%We have proven:
%
%\begin{lem}
%\label{L:triv.bdl}
%For any cohomology theory on $\SmOp/S$ represented by a commutative monoid $A$ in $SH(S)$
%the class $\Sigma_{T}^{n}1_{A} \in A^{2n,n}(\Aff^{n},\Aff^{n}-0)$ has the property that the maps
%\[
%{-} \times \Sigma_{T}^{n}1_{A} \colon A^{*,*}(X) \to A^{*+2n,*+n}(X \times \Aff^{n}, X \times (\Aff^{n}-0))
%\]
%are isomorphisms for all $X$ in $\Sm/S$.
%\end{lem}



\begin{defn}
\label{D:SL.normalized}
A special linear orientation on a bigraded ring cohomology theory $A^{*,*}$ on $\SmOp/S$ which is 
representable by a commutative monoid in $SH(S)$ is \emph{normalized} if 
%
\begin{enumerate}
\addtocounter{enumi}{4}
\item
for the trivial 
line bundle $\Aff^{1} \to \pt$ we have $\thom(\Aff^{1},1) = \Sigma_{T} 1_{A} \in A^{2,1}(\Aff^{1},\Aff^{1}-0)$.
\end{enumerate}
\end{defn}

From the multiplicativity and functoriality conditions (4) and (2) in the definition of a special linear
orientation one deduces the following result. 

\begin{lem}
\label{L:triv.Thom}
Suppose $A^{*,*}$ is a bigraded ring cohomology theory on $\SmOp/S$ representable by a commutative
monoid in $SH(S)$ with a normalized special linear orientation.   For $X \in \Sm/S$ let
$(\OO_{X}^{\oplus n},\lambda_{n})$ be the trivial special linear bundle of rank $n$ over $X$.  
Then $\thom(\OO_{X}^{\oplus n},\lambda_{n})$ is the pullback to $X$ of $\Sigma_{T}^{n}1_{A}$,
and
\[
{-} \cup \thom(\OO_{X}^{\oplus n},\lambda_{n}) 
%= {-} \times \Sigma_{T}^{n}1_{A} = \Sigma_{T}^{n} 
\colon A^{*,*}(X)  \xra{\cong}
A^{*+2n,*+n}(X \times \Aff^{n}, X \times (\Aff^{n}-0))
\]
is an isomorphism.
\end{lem}

%Recall that for a (symmetric) $T$-spectrum $M$ with spaces $(M_{0},M_{1},M_{2}, \dots)$ there are 
%canonical morphisms
%%
%\begin{equation}
%\label{E:canonical.inclusion}
%u_{n} \colon \Sigma_{T}^{\infty}M_{n}(-n) \to M
%\end{equation}
%%
%of (symmetric) $T$-spectra which are the counits of the adjunctions
%between the forgetful functors $M \mapsto M_{n}$ and their left adjoints.  When
%$(M,\mu,e)$ is a commutative monoid in the category of symmetric $T$-spectra, the 
%following diagram commutes.
%%
%\begin{equation}
%\label{E:Mm.wedge.Mn}
%\vcenter{
%\xymatrix @M=5pt @C=40pt {
%\Sigma_{T}^{\infty}M_{m}(-m) \wedge \Sigma_{T}^{\infty}M_{n}(-n) \ar[r]^-{\mu_{mn}} \ar[d]_-{u_{m} \wedge u_{n}}
%& \Sigma_{T}^{\infty}M_{m+n}(-m-n) \ar[d]^-{u_{m+n}}
%\\
%M \wedge M \ar[r]^-{\mu}
%&
%M
%}}
%\end{equation}

Now suppose $\varphi \colon \MSL \to A$ is a morphism 
%of commutative monoids 
in $SH(S)$.
We associate to $\varphi$ and a special linear bundle $(E,\lambda)$ of rank $n$ over an
$X$ in $\Sm/S$ a class $\thom^{\varphi}(E,\lambda)$ defined as follows.  By assumption
the scheme $X$ admits an ample family of line bundles.  So there exists an
affine bundle $f \colon Y \to X$ with $Y$ an affine scheme.  Then for some $p$ there exist global sections
$s_{1},\dots,s_{np}$ of $f^{*}E^{\vee}$ generating $f^{*}E^{\vee}$.  The data
$(f^{*}E, s_{1}, \dots, s_{np})$ determine a morphism $\psi \colon Y \to Gr(n,np)$, and the data
$(\psi,f^{*}\lambda)$ determine a morphism $\widetilde \psi \colon Y \to \GrSL(n,np)$.
We have $\widetilde\psi^{*} \TSL{n,np} \cong f^{*}E$.  We deduce maps
%
\begin{equation}
\label{E:Th.E.to.MSLn}
\Th E \xla[\sim\text{mot}]{\overline f} \Th f^{*}E 
\,\cong\, \Th \widetilde\psi^{*} \TSL{n,np} 
\xra{\overline{\psi}} 
\Th \TSL{n,np} 
\end{equation}
%
of pointed motivic spaces, which can be composed with the maps
%
\begin{equation}
\label{E:MSLn.to.A}
\Th \TSL{n,np} \xra{\text{inclusion}} 
 \MSL_{n} \xra{u_{n}} \MSL{} \wedge T^{\wedge n} \xra{1 \wedge \varphi} A \wedge T^{\wedge n}.
\end{equation}
%
in $SH(S)$.
The composition of \eqref{E:Th.E.to.MSLn} and \eqref{E:MSLn.to.A} gives a class
\[
\thom^{\varphi}(E,\lambda) \in Hom_{SH(S)}(\Th E, A \wedge T^{\wedge n}) = A^{2n,n}(E,E-X).
\]

\begin{lem}
\label{L:well.defined.SL}
The classes $\thom^{\varphi}(E,\lambda)$ depend only on the special linear bundle $(E,\lambda)$
and the morphism  $\varphi \colon \MSL \to A$ in $SH(S)$.
\end{lem}

\begin{proof}
First suppose $f$ fixed.  Let $(s_{1},\dots,s_{np})$ and $(t_{1},\dots,t_{nq})$ be two families
of sections generating $f^{*}E^{\vee}$ with $p \geq q$.
There are $\Aff^{1}$-homotopies between the morphisms $\Th f^{*}E \to \MSL_{n}$ in $\M_{\bullet}(S)$
defined by the family $(s_{1},\dots, s_{np})$, the family $(s_{1},\dots,s_{np},t_{1},\dots,t_{nq})$,
the family $(t_{1},\dots,t_{nq},0,\dots,0,t_{1},\dots,t_{nq})$, and the family
$(t_{1},\dots,t_{nq})$.  So we get the same morphism $\Th f^{*}E \to \MSL_{n}$ in $H_{\bullet}(S)$
and the same morphism $\Th E \to A \wedge T^{\wedge n}$ in $SH(S)$.

Now suppose given a second affine bundle $g \colon Z \to X$ with $Z$ affine 
and sections $(u_{1},\dots,u_{nr})$ generating $g^{*}E^{\vee}$.  Let $g' \colon Y \times_{X} Z \to Y$
and $f' \colon Y \times_{X} Z \to Z$ be the projections.  The morphisms $\Th E \to \MSL_{n}$
in $H_{\bullet}(S)$ defined by $f$ and $(s_{1},\dots,s_{np})$, by $g'f$ and $(g'^{*}s_{1}, \dots,
g'^{*}s_{np})$, by $f'g$ and $(f'^{*}u_{1},\dots,f'^{*}u_{nr})$ and by $g$ and $(u_{1},\dots,u_{nr})$
are then the same.  So we again get the same morphism $\Th E \to A\wedge T^{\wedge n}$ in $SH(S)$.
\end{proof}

\begin{thm}
\label{T:thom.phi}
For a homomorphism $\varphi \colon \MSL \to A$ of commutative monoids in $SH(S)$, the classes
$\thom^{\varphi}(E,\lambda)$ define a normalized special linear orientation on the bigraded ring
cohomology theory $A^{*,*}$ on 
$\SmOp(S)$.
\end{thm}

In particular the identity homomorphism induces a normalized special linear orientation on
$\MSL^{*,*}$.

\begin{proof}
The functoriality conditions (1) and (2) follow easily from the construction of the
classes $\thom^{\varphi}(E,\lambda)$.  The multiplicativity condition (4) holds because
of Theorem \ref{T:MSL.structure} and because $\varphi$ is a homomorphism of monoids.
The normalization condition (5) holds because $\thom^{\varphi}(\Aff^{1},1)$ and $\Sigma_{T}1_{A}$
are both equal to the composition 
\[
T \xra{e_{1}^{SL}} \MSL_{1} \xra{u_{1}} \MSL{} \wedge T\xra{\varphi \wedge 1} A \wedge T.
\]
The isomorphism condition (3) holds for trivial special linear bundles because of
the normalization condition and Lemma \ref{L:triv.Thom}.  
It then holds for general special linear bundles by a Mayer-Vietoris argument
because special linear bundles are locally trivial in the Zariski topology.
\end{proof}


Now suppose that $M$ and $A$ are (symmetric) $T$-spectra.  Then we have an 
inverse system of abelian groups
%
\begin{equation}
\label{E:inverse.system}
\cdots \to A^{2n+2,n+1}(M_{n+1}) \xra{\alpha_{n+1}} A^{2n,n}(M_{n}) \to \cdots \to
%A^{p+2,q+1}(M_{1}) \to 
A^{0,0}(M_{0})
\end{equation}
%
where the map $\alpha_{n}$ associates to the map $v \colon M_{n+1} \to A \wedge T^{\wedge n+1}$
in $SH(S)$ the composition
\[
M_{n} \xra{\sigma_{n}^{*}} \Omega_{T}M_{n+1} \xra{v} \Omega_{T}(A \wedge T^{\wedge n+1})
\cong A \wedge T^{\wedge n}
\]
in $SH(S)$.  There is a similar inverse system
%
\begin{equation}
\label{E:inverse.system.2}
\cdots \to A^{4n+4,2n+2}(M_{n+1} \wedge M_{n+1}) \to A^{4n,2n}(M_{n} \wedge M_{n}) \to \cdots \to
%A^{p+2,q+1}(M_{1}) \to 
A^{0,0}(M_{0} \wedge M_{0}).
\end{equation}
%


For the following theorem see for example 
\cite
[Corollaries 3.4 and 3.5]
%[Lemma 3.3 and Corollaries 3.4 and 3.5]
{Panin:2009aa}.

\begin{thm}
\label{T:inverse.lim}
For any \parens{symmetric} $T$- or $T^{\wedge 2}$-spectra $M$ and $A$  we have exact sequences
of abelian groups
%
\begin{gather*}
0 \to \sideset{}{^{1}}\varprojlim A^{2n-1,n}(M_{n}) \to Hom_{SH(S)}(M,A) \to 
\varprojlim A^{2n,n}(M_{n}) \to 0,
\\
0 \to \sideset{}{^{1}}\varprojlim A^{4n-1,2n}(M_{n} \wedge M_{n}) \to Hom_{SH(S)}(M \wedge M,A) \to 
\varprojlim A^{4n,2n}(M_{n} \wedge M_{n}) \to 0.
\end{gather*}
\end{thm}

This theorem is actually a special case of the following result \cite[Lemma 3.3]{Panin:2009aa}.

\begin{thm}
\label{T:hocolim}
Let $E = \hocolim_{i \in \NN} E^{\angles{i}}$ be a 
sequential homotopy colimit of $T$-spectra. 
%\parens{or motivic spaces}.  
Then for any $T$-spectrum $A$ and any $(p,q)$ we have an exact sequence
of abelian groups
\[
0 \to \sideset{}{^{1}}\varprojlim_{i \in\NN} A^{p-1,q}(E^{\angles{i}}) \to A^{p,q}(E) \to 
\varprojlim\limits_{i\in\NN} A^{p,q}(E^{\angles{i}}) \to 0.
\]
\end{thm}

We wish to apply Theorem \ref{T:inverse.lim} when $A$ is a commutative monoid in $SH(S)$ 
with a normalized special linear orientation on $A^{*,*}$ and when $M$ is a commutative monoid
isomorphic to $\MSL$ in $SH(S)$.  (Note that the exact sequences depend on the levelwise weak 
equivalence class of $M$, which is a finer invariant than its isomorphism class in $SH(S)$.)
However, the special linear orientation 
provides Thom classes for special linear bundles over finite-dimensional smooth schemes 
and not over the infinite-dimensional ind-schemes $BSL_{n}$.  
So the orientation does not provide us with classes in 
the $A^{2n,n}(\MSL_{n})$.  But we can solve this problem as follows.

For each $n$ and $p$ write
\[
MSL_{n}^{\angles{p}} = \Th \TSL{n,np}.
\]
For $n=0$ this is $MSL_{0}^{\angles{p}} = \pt_{+}$.
The actions of $\sym_{n}$ on $\MSL_{n}$ and the structural maps $e_{n}^{SL}$ and $\mu_{mn}^{SL}$
constructed in the previous section are colimits of actions and structural maps
%
\begin{gather*}
\sym_{n} \times MSL_{n}^{\angles{p}} \to MSL_{n}^{\angles{p}},
\\
e_{n}^{\angles{p}} \colon T^{\wedge n} \to MSL_{n}^{\angles{p}},
\\
\mu_{mn}^{\angles{p}} \colon MSL_{m}^{\angles{p}} \wedge MSL_{n}^{\angles{p}}
\to MSL_{m+n}^{\angles{p}}.
\end{gather*}
%
We thus get a direct system of commutative $T$-monoids
\[
MSL^{\angles{1}} \to MSL^{\angles{2}} \to \cdots \to MSL^{\angles{p}} \to \cdots
\]
whose colimit is $\MSL$.  We can now define a ``diagonal'' commutative $T$-monoid
$\MSL^{\finite}$ with spaces
\[
\MSL^{\finite}_{n} = MSL_{n}^{\angles{n}}
\]
with the actions $\sym_{n} \times MSL_{n}^{\angles{n}} \to MSL_{n}^{\angles{n}}$ and unit maps
$e_{n}^{\angles{n}}$ given above and with multiplication maps the compositions
\[
\mu_{mn}^{\finite} \colon
MSL_{m}^{\angles{m}} \wedge MSL_{n}^{\angles{n}} \xra{\text{inclusion}}
MSL_{m}^{\angles{m+n}} \wedge MSL_{n}^{\angles{m+n}} \xra{\mu_{mn}^{\angles{m+n}}}
MSL_{m+n}^{\angles{m+n}}.
\]
A cofinality argument now gives the nontrivial part of the following result.

\begin{thm}
The inclusion $\MSL^{\finite} \hra \MSL$ defines a homomorphism of commutative monoids
in the category of symmetric $T$-spectra which is a motivic stable weak equivalence.
\end{thm}

Thus the inclusion becomes an isomorphism of commutative monoids in $SH(S)$.  So 
Theorem \ref{T:inverse.lim} gives us an exact sequence
%
\begin{equation}
\label{E:exact.seq}
0 \to \varprojlim\nolimits^{1} A^{2n-1,n}(MSL_{n}^{\angles{n}}) \to
Hom_{SH(S)}(\MSL,A) \to \varprojlim A^{2n,n}(MSL_{n}^{\angles{n}}) \to 0
\end{equation}
%
and a similar exact sequence for $Hom_{SH(S)}(\MSL \wedge \MSL,A)$.

\begin{thm}
\label{T:univ.SL}
Suppose $(A,\mu_{A},e_{A})$ is a commutative monoid in $SH(S)$ with a normalized special linear
orientation on $A^{*,*}$ given by Thom classes $\thom(E,\lambda)$.  Then there exists a
morphism $\varphi \colon \MSL \to A$ in $SH(S)$ such that 
$\thom^{\varphi}(E,\lambda) = \thom(E,\lambda)$
for all special linear bundles over all $X$ in $\Sm/S$.   This $\varphi$ is unique modulo the subgroup
\[
\sideset{}{^{1}}\varprojlim A^{2n-1,n}(MSL_{n}^{\angles{n}}) \subset Hom_{SH(S)}(\MSL,A).
\]
It satisfies $\varphi(e_{\MSL}) = e_{A}$.  
The obstruction
\(
\varphi \circ \mu_{\MSL} - \mu_{A} \circ (\varphi \wedge \varphi)
\)
to $\varphi$ being a
homomorphism of monoids lies in the subgroup
\[
\sideset{}{^{1}}\varprojlim A^{4n-1,2n}(MSL_{n}^{\angles{n}} \wedge MSL_{n}^{\angles{n}}) 
\subset Hom_{SH(S)}(\MSL \wedge \MSL,A).
\]
\end{thm}

\begin{proof}
%The first $\varprojlim^{1}$ group is the kernel of a surjection
%%\[
%%\xymatrix @M=5pt @C=25pt {
%%Hom_{SH(S)}(\MSL,A) \ar@{->>}[r] 
%%& 
%%\varprojlim\limits_{n} A^{2n,n}(MSL_{n}^{\angles{n}})
%%= \varprojlim\limits_{n,p} A^{2n,n}(MSL_{n}^{\angles{p}}).
%%}
%%\]
%\[
%Hom_{SH(S)}(\MSL,A) \lra
%\Bigl[ \varprojlim\limits_{n} A^{2n,n}(MSL_{n}^{\angles{n}})
%= \varprojlim\limits_{n,p} A^{2n,n}(MSL_{n}^{\angles{p}}) \Bigr]
%\lra 0.
%\]
For every $n$ and $p$  the tautological special linear bundle 
$(\TSL{n,np}, \lambda_{n,np})$ 
over the scheme $\GrSL(n,np)$
has a Thom class, which we will abbreviate to
%$\thom(\TSL{n,np}, \lambda_{n,np}) 
$\thom_{n,np}
\in A^{2n,n}(MSL_{n}^{\angles{p}})$.
%= Hom_{SH(S)}(MSL_{n}^{\angles{p}}, A \wedge T^{\wedge n}).
Pullback along the inclusion $MSL_{n}^{\angles{p-1}} \hra MSL_{n}^{\angles{p}}$ sends
%
\begin{equation}
\label{E:th(n,np).1}
\thom_{n,np} \mapsto \thom_{n,n(p-1)}.
%\thom(\TSL{n,np}, \lambda_{n,np}) 
%\mapsto
%\thom(\TSL{n,n(p-1)}, \lambda_{n,n(p-1)}) 
%\thom(n,np) \in A^{2n,n} \bigl( MSL_{n}^{\angles{p}} \bigr) \Bigr] \longmapsto
%\Bigl[ \thom(n,n(p-1)) \in A^{2n,n} \bigl( MSL_{n}^{\angles{p-1}} \bigr) \Bigr] 
\end{equation}
%
Pullback along the bonding map $MSL_{n-1}^{\angles{p}} \wedge T \to MSL_{n}^{\angles{p}}$
induced by $e_{1}^{\angles{p}}$ and $\mu_{n-1,1}^{\angles{p}}$ sends
%
\begin{equation}
%\begin{multline}
\label{E:th(n,np).2}
%\thom(\TSL{n,np}, \lambda_{n,np}) 
%\mapsto
%\thom(\TSL{n{-}1,(n{-}1)p}, \lambda_{n-1,(n-1)p}) \times \thom(\Aff^{1},1)
%\\ =
%\Sigma_{T} \thom(\TSL{n{-}1,(n{-}1)p}, \lambda_{n-1,(n-1)p}).
\thom_{n,np} \mapsto
\thom_{n-1,(n-1)p} \times \thom(\Aff^{1},1) = 
\Sigma_{T}\thom_{n-1,(n-1)p}.
\end{equation} 
%%\end{aligned}
%\end{multline}
%\begin{multline}
%\label{E:th(n,np).2}
%%\begin{aligned}
%\Bigl[ \thom(n,np) \in A^{2n,n} \bigl( MSL_{n}^{\angles{p}} \bigr) \Bigr] 
%\longmapsto
%\Bigl[ \thom(n-1,np-p) \times \Sigma_{T}1 \in A^{2n,n} \bigl( MSL_{n-1}^{\angles{p}} \wedge T \bigr) \Bigr] 
%\\ \longmapsto 
%\Bigl[ \thom(n-1,np-p) \in A^{2n-2,n-1} \bigl( MSL_{n-1}^{\angles{p}} \bigr) \Bigr]
%%\end{aligned}
%\end{multline}
%
So as $n$ and $p$ vary, we get an element
\[
\bar\varphi = (
\thom_{n,np} 
%\thom(\TSL{n,np}, \lambda_{n,np}) 
)
_{n,p}
%_{(n,p) \in \NN^{2}} 
\in \varprojlim\limits_{n,p} A^{2n,n}(MSL_{n}^{\angles{p}}) 
= \varprojlim\limits_{n} A^{2n,n}(MSL_{n}^{\angles{n}}).
\]
Let $\varphi \in Hom_{SH(S)}(\MSL,A)$ be an element mapping onto $\bar\varphi$ under the
surjection in the exact sequence \eqref{E:exact.seq}.

The image of $\varphi$ under the composition
\[
Hom_{SH(S)}(\MSL,A) \to \varprojlim A^{2n,n}(MSL_{n}^{\angles{n}}) \to 
A^{2n,n}(MSL_{n}^{\angles{n}})  
\]
is the composition
\[
\Th \TSL{n,n^{2}} = MSL_{n}^{\angles{n}} \xra{u_{n}} \MSL^{\finite}{}  \wedge T^{\wedge n}
\xra{\sim}
\MSL{}  \wedge T^{\wedge n} \xra{\varphi \wedge 1} A \wedge T^{\wedge n}
\]
which is the $\thom^{\varphi}(\TSL{n,n^{2}},\lambda_{n,n^{2}})$ defined
by \eqref{E:Th.E.to.MSLn}--\eqref{E:MSLn.to.A}.
%
%The surjective map in the exact sequence sends $\varphi \in Hom_{SH(S)}(\MSL,A)$ to
%\[
%\bigl( \thom^{\varphi}(\TSL{n,n^{2}},\lambda_{n,n^{2}}) \bigr)_{n \in \NN} \in 
%\varprojlim A^{2n,n}(MSL_{n}^{\angles{n}})
%\]
%
%
Thus we have 
$\thom^{\varphi}(E,\lambda) = \thom(E,\lambda)$ for 
$(E,\lambda) = (\TSL{n,n^{2}}, \lambda_{n,n^{2}})$.  
The Thom classes for the $(\TSL{n,n^{2}}, \lambda_{n,n^{2}})$ determine the
Thom classes for all $(\TSL{n,np}, \lambda_{n,np})$ by 
formulas \eqref{E:th(n,np).1}--\eqref{E:th(n,np).2}.
These in turn determine the Thom classes for all $(E,\lambda)$ by formulas
\eqref{E:Th.E.to.MSLn}--\eqref{E:MSLn.to.A}.  
So we have $\thom^{\varphi}(E,\lambda) = \thom(E,\lambda)$ for 
all special linear bundles.

Similarly for $\psi \colon \MSL \to A$ we have 
$\thom^{\psi}(E,\lambda) = \thom^{\varphi}(E,\lambda)$ for all special linear bundles
if and only if $\psi$ and $\varphi$ have the same image in 
$\varprojlim A^{2n,n}(MSL_{n}^{\angles{n}})$.  This happens if and only if
$\psi - \varphi$ is in the kernel, which is the first $\varprojlim^{1}$ of the statement of the theorem.
% $\sideset{}{^{1}}\varprojlim A^{2n-1,n}(MSL_{n}^{\angles{n}})$.



By construction $e_{\MSL}$ is the canonical map 
$\Sigma_{T}^{\infty} \pt_{+} = \Sigma_{T}^{\infty} \MSL_{0} \to \MSL$.
Therefore we have $\varphi(e_{\MSL}) = \thom_{0,0} = \thom(\boldsymbol{0}) = e_{A} \in A^{0,0}(\pt)$ 
as declared.


%Write $\thom_{n} = \thom(\TSL{n,n^{2}},\lambda_{n,n^{2}})$ for all $n$. 
%
By multiplicativity and functoriality we have an equality
%
%\begin{multline*}
%\thom(\TSL{n,n^{2}},\lambda_{n,n^{2}}) \times 
%\thom(\TSL{n,n^{2}},\lambda_{n,n^{2}}) = 
%\thom(\TSL{n,n^{2}} \oplus \TSL{n,n^{2}},\lambda_{n,n^{2}} \otimes \lambda_{n,n^{2}})
%\\ =
%\mu_{nn}^{\finite *}
%\thom(\TSL{2n,4n^{2}},\lambda_{2n,4n^{2}}).
%\end{multline*}
%
\[
\thom_{n,n^{2}} \times \thom_{n,n^{2}} = 
\thom(p_{1}^{*}\TSL{n,n^{2}} \oplus p_{2}^{*}\TSL{n,n^{2}},p_{1}^{*}\lambda_{n,n^{2}} \otimes p_{2}^{*}\lambda_{n,n^{2}})
=
\mu_{nn}^{\finite *}
\thom_{2n,4n^{2}}
\]
of members of 
$A^{4n,2n}(MSL_{n}^{\angles{n}} \wedge MSL_{n}^{\angles{n}})$.  This equality means that the 
outer perimeter of the diagram
\[
\xymatrix @M=5pt {
\Sigma_{T}^{\infty} MSL_{n}^{\angles{n}} (-n) 
\wedge
\Sigma_{T}^{\infty} MSL_{n}^{\angles{n}} (-n) 
\ar[r]^-{\mu_{nn}^{\finite}} 
\ar[d]^-{u_{n} \wedge u_{n}}
\ar@/_25mm/[ddd]_-{\thom_{n,n^{2}} \wedge \thom_{n,n^{2}}}
%{\thom(\TSL{n,n^{2}},\lambda_{n,n^{2}}) \times 
%\thom(\TSL{n,n^{2}},\lambda_{n,n^{2}})}
&
\Sigma_{T}^{\infty} MSL_{2n}^{\angles{2n}} (-2n) 
\ar[d]^-{u_{2n}}
\ar@/^25mm/[ddd]^-{\thom_{2n,4n^{2}}}
%{\thom(\TSL{2n,4n^{2}},\lambda_{2n,4n^{2}})}
\\
\MSL^{\finite} \wedge \MSL^{\finite} 
\ar[r]^-{\mu_{\MSL}^{\finite}}
\ar[d]
%_-{\sim\text{stable}}
%^-{i \wedge i}
^-{\text{inclusion}}
&
\MSL^{\finite} 
\ar[d]
%_-{\sim\text{stable}}
%^-{i}
^-{\text{inclusion}}
\\
\MSL \wedge \MSL
\ar[r]^-{\mu_{\MSL}}
\ar[d]^-{\varphi \wedge \varphi}
&
\MSL 
\ar[d]^-{\varphi}
\\
A \wedge A \ar[r]^-{\mu_{A}}
& A
}
\]
commutes.  The half-circles commute by the previous calculations, and the top two squares commute.
Therefore we have
\[
\bigl( \varphi \circ \mu_{\MSL} - \mu_{A} \circ (\varphi \wedge \varphi) \bigr) 
\circ 
%{\sim\text{stable}}
\text{inclusion}
\circ (u_{n} \wedge u_{n}) = 0
\]
for all $n$.  So the image of the obstruction class
$\varphi \circ \mu_{\MSL} - \mu_{A} \circ (\varphi \wedge \varphi)$
under the surjection
\[
Hom_{SH(S)}(\MSL \wedge \MSL,A) 
\lra 
\varprojlim A^{4n,2n}(MSL_{n}^{\angles{n}} \wedge MSL_{n}^{\angles{n}})
\lra 0
\]
vanishes. Therefore the obstruction class lies in the kernel, which is the second $\varprojlim^{1}$
of the statement of the theorem.
\end{proof}

\section{The symmetric \texorpdfstring{$T^{\wedge 2}$}{T\^{ }2}-spectrum \texorpdfstring{$\MSp$}{MSp}}

We now define the commutative $T^{\wedge 2}$-monoid and 
symmetric $T^{\wedge 2}$-spectrum $\MSp$.  


We write the standard symplectic form on the trivial vector bundle of rank $2n$ as
\[
\omega_{2n} =
\begin{pmatrix}
\begin{matrix}
0&1 \\ -1&0
\end{matrix}
&& 0 \\
& \ddots & \\
0 &&
\begin{matrix}
0&1 \\ -1&0
\end{matrix}
\end{pmatrix}
\] 
From the symplectic isometry 
$(\OO_{S}^{\oplus 2n},\omega_{2n}) \cong (\OO_{S}^{\oplus 2},\omega_{2})^{\oplus n}$
we see that the action of $\sym_{n}$ given by permutations of the $n$ orthogonal direct summands
$(\OO_{S}^{\oplus 2},\omega_{2})$
gives an embedding $\sym_{n} \to Sp_{2n}$.  Hence $Sp_{2n}$-actions restrict to 
$\sym_{n}$-actions.

In \cite{Panin:2010fk} we defined the quaternionic Grassmannian $HGr(r,n)$ as the open subscheme
of the Grassmannian $Gr(2r,2n)$ parametrizing rank $2r$ subspaces of $\OO_{S}^{\oplus 2n}$ on which
the restriction of $\omega_{2n}$ is nondegenerate.  The restriction of the tautological subbundle 
over the Grassmannian is the tautological symplectic subbundle $\TSp{r,n}$.  It is equipped with
the symplectic form $\phi_{r,n} = \omega_{2n} |_{\TSp{r,n}}$.  
For 
$r=1$ we write $HP^{n} = HGr(1,n+1)$ and $HP^{\infty} = \colim_{n} HP^{n}$.

To construct $\MSp$ we look at the particular schemes
$HGr(n,np) = HGr(n, (\OO_{S}^{\oplus 2n}, \omega_{2n})^{\oplus p})$.  Each has a natural action
of $Sp_{2n}$ induced by the diagonal action of $Sp_{2n}$ on the $p$ summands of 
$(\OO_{S}^{\oplus 2n}, \omega_{2n})^{\oplus p}$.  The vector bundles 
$\TSp{n,np} \to HGr(n,np)$ and the inclusions $HGr(n,np) \to HGr(n,np+n)$
are $Sp_{2n}$-equivariant.  We set
%
\begin{gather*}
BSp_{2n} = \colim\nolimits_{p \in \NN} HGr(n,np), \\
\TSp{n,n\infty} = \colim\nolimits_{p \in \NN} \TSp{n,np}, \\
\MSp_{2n} = \colim\nolimits_{p \in \NN} \Th \TSp{n,np}.
\end{gather*}

As with $\MGL$ and $\MSL$ the isomorphisms
\[
(\OO_{S}^{\oplus 2m}, \omega_{2m}) \oplus (\OO_{S}^{\oplus 2n}, \omega_{2n}) 
\cong (\OO_{S}^{\oplus 2m+2n}, \omega_{2m+2n})
\]
and the direct sum induce $(Sp_{2m} \times Sp_{2n})$-equivariant maps
%
\begin{equation}
\label{E:monoid.Sp}
\begin{gathered}
\oplus \colon BSp_{2m} \times BSp_{2n} \to BSp_{2m+2n},
\\
\mu^{Sp}_{mn} \colon \MSp_{2m} \wedge \MSp_{2n} \to \MSp_{2m+2n}.
\end{gathered}
\end{equation}

Each $HGr(n,np)$ is pointed by the point corresponding to the symplectic subbundle
which is the first direct summand  
$(\OO_{S}^{\oplus 2n}, \omega_{2n}) \oplus 0^{\oplus p-1} \subset 
(\OO_{S}^{\oplus 2n}, \omega_{2n}) ^{\oplus p}$.  In the colimit this yields
points $z_{2n} \colon \pt \to BSp_{2n}$.  The point $z_{2n}$ is fixed by the
$Sp_{2n}$-action.  The action of $Sp_{2n}$ on the fiber of $\TSp{n,n\infty}$
over $z_{2n}$ is the standard representation of $Sp_{2n}$.  The inclusion of the 
fiber induces an inclusion of Thom spaces
%
\begin{equation}
\label{E:e.Sp}
e_{2n}^{Sp} \colon T^{\wedge 2n} \to \MSp_{2n}
\end{equation}
%
which is $Sp_{2n}$-equivariant.  The action of the subgroup $\sym_{n} \subset Sp_{2n}$
on $T^{\wedge 2n} = (T^{\wedge 2})^{\wedge n}$ permutes the $n$ factors $T^{\wedge 2}$.

The spaces $\MSp_{2n}$ with the actions and structural maps verify the axioms of a 
commutative $T^{\wedge 2}$-monoid.

\begin{defn}
The \emph{algebraic symplectic cobordism spectrum} $\MSp$ is the 
commutative monoid in the category of symmetric $T^{\wedge 2}$-spectra associated 
to the commutative $T^{\wedge 2}$-monoid composed of 
the spaces $\MSp_{2n}$, the actions $\sym_{n} \times \MSp_{2n} \to \MSp_{2n}$ 
%induced by the action of $Sp_{2n}$, 
the maps
$e_{0}^{Sp} \colon \pt_{+} \to \MSp_{0}$ and $e_{2}^{Sp} \colon T^{\wedge 2} \to \MSp_{2}$ 
and the maps $\mu_{mn}^{Sp} \colon \MSp_{2m} \wedge \MSp_{2n} \to \MSp_{2m+2n}$.
\end{defn}

This $\MSp$ defines a commutative monoid in $SH(S)$ by Theorem \ref{T:comparison}.


\section{Quaternionic Grassmannian bundles}


We review the geometry of quaternionic projective bundles and Grassmannian bundles studied
in \cite[\S\S 3--5]{Panin:2010fk}.  We then translate some of the results into a more motivic language.


Given $(E,\phi)$ a symplectic bundle of rank $2n$ over a scheme $X$ and an integer $0 \leq r \leq n$,
there is a quaternionic Grassmannian bundle $p \colon HGr(r,E,\phi) \to X$ whose fiber over
$x \in X$ is the quaternionic Grassmannian parametrizing $2r$-dimensional subspaces of $E_{x}$  
on which $\phi_{x}$ is nondegenerate.  
We write $\shf U_{r,E} \subset p^{*}E$ for the tautological rank
$2r$ subbundle over $HGr(r,E,\phi)$.  
Morphisms $f \colon Y \to HGr(r,E,\phi)$ are in bijection
with pairs $(g,U)$ where $g \colon Y \to X$ is a morphism and $U \subset g^{*}(E,\phi)$ is a symplectic
subbundle of rank $2r$ over $Y$.

Since $\shf U_{r,E}$ is a subbundle on which the symplectic form is fiberwise nondegenerate, it has
an orthogonal complement such that $\shf U_{r,E} \oplus \shf U_{r,E}^{\perp} = p^{*}E$.  The symplectic 
subbundle $\shf U_{r,E}^{\perp } \subset p^{*}E$ classifies an isomorphism
%
\begin{equation}
\label{E:perp}
HGr(r,E,\phi) \xra{\cong} HGr(n-r,E,\phi).
\end{equation}

Now let $(F,\psi) = (\OO_{X}^{\oplus 2},\omega_{2}) \oplus (E,\phi)$.  We have a natural embedding
%
\begin{equation}
\label{E:embed.1}
HGr(r,E,\phi) \hra HGr(r,F,\psi)
\end{equation}
%
classified by the symplectic subbundle $0 \oplus \shf U_{r,E} \subset \OO_{X}^{\oplus 2} \oplus E = F$.
The normal bundle of this embedding can be naturally identified with the vector bundle
$N = \shf Hom(\shf U_{r,E},\OO_{X}^{\oplus 2})$ over $HGr(r,E,\phi)$.  This bundle is a direct sum
decomposition $N = N^{+} \oplus N^{-}$ where
%
\begin{align*}
N^{+} & = \shf Hom(\shf U_{r,E}, \OO_{X} \oplus 0), 
&
N^{-} & = \shf Hom(\shf U_{r,E}, 0 \oplus \OO_{X}).
\end{align*}
%
The basic result concerning the geometry of the closed embedding \eqref{E:embed.1} is the following.

%We work in a relative situation.  Our basic setup is as follows.
%%We return to the basic situation of \eqref{E:basic.setup.1}.  
%Suppose $(E,\phi)$ is a symplectic bundle of rank $2n-2$ over $S$, and let $(F,\psi)$
%be the symplectic bundle of rank $2n$ with 
%%
%\begin{align}
%\label{E:basic.setup}
%F & = \OO_{S} \oplus E \oplus \OO_{S}, &
%\psi & = 
%\begin{pmatrix}
%0 & 0 & 1 \\ 0 & \phi & 0 \\ -1 & 0 & 0
%\end{pmatrix}.
%\end{align}
%%
%We will consider the natural embedding of 
%$\HGr_{S}(r;E,\phi)$ in $\HGr_{S}(r;F,\psi)$.
%Let $(\shf U_{E},\phi_{\shf U_{E}})$ and $(\shf U_{F},\psi_{\shf U_{F}})$ be the
%tautological symplectic subbundles of rank $2r$.
%We will abbreviate
%%
%\begin{align}
%\label{E:HGr.abbrv}
%\HGr(E) & = \HGr_{S}(r;E,\phi)
%&
%\HGr(F) & = \HGr_{S}(r;F,\psi).
%\end{align}
%%\end{setup}
%%
%We now study the embedding of $\HGr(E)$ in $\HGr(F)$.

\begin{thm}[\protect{\cite[Theorem 4.1]{Panin:2010fk}}]
\label{T:normal.geom}
\parens{a} The normal bundle of the embedding \eqref{E:embed.1} has
a canonical open embedding $\nu \colon N \hra Gr(2r,F)$.  The zero section is sent
identically onto $HGr(r,E,\phi)$.

\parens{b} We have $\nu(N^{+}) = HGr(r,F,\psi) \cap \Gr_{S}(2r,\OO_{X}\oplus 0\oplus E)$.  
Consequently
$\nu(N^{+}) \subset HGr(r,F,\psi)$ is a closed subscheme, as is $\nu(N^{-}) \subset HGr(r,F,\psi)$.

\parens{c} There are natural isomorphisms of vector bundles
$N^{+} \cong N^{-} \cong \shf U_{r,E}^{\vee} \cong \shf U_{r,E}$.

\parens{d} There is a natural section $s_{+}$ of $\shf U_{r,F}$ 
intersecting the zero section transversally in $N^{+}$ and similarly for $N^{-}$.

\parens{e} Let $\pi_{+} \colon N^{+} \to HGr(r,E,\phi)$ be the structural map.  Then
$\pi_{+}^{*}(\shf U_{r,E},\phi|_{\shf U_{r,E}})$ is isometric to 
$(\shf U_{r,F},\psi|_{\shf U_{r,F}})|_{N^{+}}$ and similarly for $N^{-}$.
\end{thm}

The second basic result about the geometry of symplectic Grassmannian bundles involves the 
following embeddings
%
\begin{equation}
\label{E:embed.2}
\xymatrix @M=5pt @C=40pt {
HGr(r-1,E,\phi) \ar@{^{(}->}[r]^-{\sigma}_-{\text{closed}} & 
HGr(r,F,\psi) - \nu(N^{+}) \ar@{^{(}->}[r]^-{\text{open}} & HGr(r,F,\psi).
}
\end{equation}
%
The composition is the closed embedding classified by the symplectic subbundle
$\OO^{\oplus 2} \oplus \shf U_{r-1,E} \subset \OO^{\oplus 2} \oplus E = F$.

\begin{thm}[\protect{\cite[Theorems 5.1 and 5.2]{Panin:2010fk}}]
\label{T:affine.bundles}
There are morphisms over $X$
\[
HGr(r,F,\psi) - \nu(N^{+}) \xleftarrow{g_{1}} Y_{1} \xleftarrow{g_{2}} 
Y_{2} \xrightarrow{q} HGr(r-1,E,\phi)
\]
with $g_{1}$ an $\Aff^{2r-1}$-bundle, $g_{2}$ an $\Aff^{2r-2}$-bundle, and 
$q$ an $\Aff^{4n+1}$-bundle.  Moreover, 
there is a section $s$
%$s \colon HGr(r-1,E,\phi) \to Y_{2}$ 
of $q$ such that the composition
$g_{1}g_{2}s \colon HGr(r-1,E,\phi) \to HGr(r,F,\psi)- \nu(N^{+})$ 
is the closed embedding $\sigma$ of \eqref{E:embed.2}.
\end{thm}

The section $s$ appears at the end of the proof of \cite[Theorem 5.2]{Panin:2010fk}.  The statement
of the theorem only contains the consequence that $\sigma$ induces isomorphisms of cohomology groups.



These two theorems have the following consequence.

\begin{thm}
\label{T:HGr/HGr}
Let $(E,\phi)$ be a symplectic bundle of rank $2n$ over a smooth $S$-scheme $X$, and let
$(F,\psi) = (\OO_{X}^{\oplus 2}, \omega_{2}) \oplus (E,\phi)$.  For $1 \leq r \leq n$ let
$\shf U_{r,E}$ be the tautological symplectic subbundle over $HGr(r,E,\phi)$.  
Let $HGr(r-1,E,\phi) \hra HGr(r,F,\psi)$ be the closed embedding of \eqref{E:embed.2}. 
Then there is a canonical zigzag of motivic weak equivalences
\[
\Th \shf U_{r,E} 
%\cong N^{-}/(N^{-}-HGr(r,E,\phi)) 
\xra{\sim} HGr(r,F,\psi)/(HGr(r,F,\psi) - \nu(N^{+}))
\xla{\sim}
HGr(r,F,\psi) / HGr(r-1,E,\phi)
\]
inducing an isomorphism in the motivic unstable homotopy category $H_{\bullet}(S)$.  
These isomorphisms commute with the maps induced by inclusions
$(E,\phi) \hra (E,\phi) \oplus (E_{1},\phi_{1})$ of symplectic bundles and with the maps 
induced by base changes $Y \to X$.
\end{thm}

\begin{proof}
In the geometry of Theorem \ref{T:normal.geom}
we identify $N$, $N^{+}$ and $N^{-}$ with their images in $Gr(2r,F)$ under the open embedding $\nu$.
Then there 
%is an isomorphism
%\[
%\Th \shf U_{r,E} \cong 
%%\Th N^{-} = 
%N^{-}/(N^{-}-HGr(r,E,\phi)) \overset{\text{v.b.}}{\underset{\sim}{\longleftrightarrow}} 
% N/(N-N^{+}) \cong \nu(N)/(\nu(N)-\nu(N^{+})),
%\]
are motivic weak equivalences
\[
\xymatrix @M=5pt @C=20pt {
N/(N-N^{+}) & \ar[l]_-{\sim}^-{\text{excision}} 
(N \cap HGr(r,F,\psi))/(N \cap HGr(r,F,\psi) - N^{+}) \ar[d]_-{\sim}^-{\text{excision}} 
\\ 
\Th \shf U_{r,E} \cong 
N^{-}/(N^{-}-HGr(r,E,\phi)) 
\ar[u]_-{\sim}^-{\text{section of vector bundle}} 
\ar[ru] \ar[r]_-{\text{inclusion}}
& 
HGr(r,F,\psi)/(HGr(r,F,\psi)-N^{+})
}
\]
The arrows not explicitly labeled $\sim$ are nevertheless motivic weak equivalences by the 2-out-of-3 
axiom.  The morphisms $g_{1}$, $g_{2}$ and
$q$ of Theorem \ref{T:affine.bundles} are affine bundles, so they, the section $s$,
the composition $\sigma$, and the map
\[
HGr(r,F,\psi)/HGr(r-1,E,\phi) \xra{\sim} HGr(r,F,\psi)/(HGr(r,F,\psi)-N^{+})
\]
of quotient spaces induced by $\sigma$ are all motivic weak equivalences.  

The functoriality is straightforward.
\end{proof}


Because of \eqref{E:perp} the quotient appearing in Theorem \ref{T:HGr/HGr} is also isomorphic to
\[
HGr(n-r+1,F,\psi)/HGr(n-r+1,E,\phi).
\]
The case $r=n$ of Theorem \ref{T:HGr/HGr} therefore gives the following result.  
The corresponding result for ordinary Grassmannian
bundles is \cite[Proposition 3.2.17(3)]{Morel:1999ab}.

\begin{thm}
Suppose that $(E,\phi)$ is a symplectic bundle of rank $2n$ over a smooth $S$-scheme $X$.
Let $HP(E,\phi) \hra HP(\OO_{X}^{\oplus 2} \oplus E, \omega_{2} \oplus \phi)$ 
be the natural closed embedding.  Then we have isomorphisms in $H_{\bullet}(S)$
\[
\Th E \cong HGr(n,(\OO_{X}^{\oplus 2}, \omega_{2}) \oplus (E,  \phi))/HGr(n-1,E,\phi) =
HP((\OO_{X}^{\oplus 2}, \omega_{2}) \oplus (E,  \phi)) / HP(E,\phi).
\]
\end{thm}

For $X = \pt$ and trivial $E$ Theorem \ref{T:HGr/HGr} gives the following result.

\begin{thm}
\label{T:BSp/BSp}
There are canonical isomorphisms 
\[
\Th \shf U_{HGr(r,n)} \cong HGr(r,1+n)/HGr(r-1,n)
\]
in $H_{\bullet}(S)$.  These isomorphisms are compatible with the inclusions of quaternionic projective spaces.  
Therefore we have commutative diagrams of
inclusions and isomorphisms in $H_{\bullet}(S)$
%
\begin{equation}
\label{E:BSp/BSp}
\vcenter{
\xymatrix @M=5pt @C=40pt {
T^{\wedge 2r} \ar[r]^-{e_{r}^{Sp}} \ar[d]_-{\cong}
& \MSp_{2r} \ar[d]^-{\cong} 
\\
HP^{r}/HP^{r-1} \ar[r]_-{\text{\textup{inclusion}}} 
& BSp_{2r}/BSp_{2r-2}
}}
\end{equation}
\end{thm}

The motivic weak equivalences of Theorem \ref{T:BSp/BSp} fit into a commutative diagram
%
\begin{equation}
\label{E:amazing.1}
\vcenter{
\xymatrix @M=5pt @C=30pt {
HGr(r,n) \ar[rr]^-{\text{inclusion}} \ar[d]_-{\text{section map}}
&&
HGr(r,1+n) \ar[d]^-{\text{quotient}}
\\
\Th \shf U_{HGr(r,n)} \ar[r]^-{\sim \, mot} 
&
Y_{r,n}
&
HGr(r,1+n)/HGr(r-1,n) \ar[l]_-{\sim \,mot}
}}
\end{equation}
%
The \emph{section map} is the structure map of the Thom space induced by a section of the vector bundle.
We have a colimit as $n \to \infty$:
%
\begin{equation}
\label{E:amazing.2}
\vcenter{
\xymatrix @M=5pt @C=30pt {
BSp_{2r} = HGr(r,\infty) \ar[rr]^-{\text{shift}} 
\ar[d]_-{\text{section map}}
&&
BSp_{2r} = HGr(r,1+\infty) \ar[d]^-{\text{quotient}}
\\
\MSp_{2r} \ar[r]^-{\sim \, mot} 
&
Y_{r,\infty}
&
BSp_{2r}/BSp_{2r-2} \ar[l]_-{\sim \,mot}
}}
\end{equation}
%
The \emph{shift map} is the map induced by the \emph{shift endomorphism} of
$(\OO^{\oplus 2},\omega_{2})^{\oplus \infty}$ acting on sequences of sections of $\OO$ by
\[
(s_{1},s_{2},s_{3},s_{4},\dots) \mapsto (0,0,s_{1},s_{2},s_{3},s_{4},\dots). 
\]

\begin{lem}
There is an $\Aff^{1}$-homotopy of symplectic endomorphisms of 
$(\OO^{\oplus 2},\omega_{2})^{\oplus \infty}$ linking the 
shift endomorphism of $(\OO^{\oplus 2},\omega_{2})^{\oplus \infty}$ to the identity.
\end{lem}

%More precisely there is a compatible system of $\Aff^{1}$-homotopies of symplectic embeddings
%$F_{n} \colon (\OO^{\oplus 2},\omega_{2})^{\oplus n} \times \Aff^{1} \to (\OO^{\oplus 2},\omega_{2})^{\oplus n+1}$ such that $F_{n}(0)$ is $(x_{1},\dots, x_{2n}) \mapsto (x_{1},\dots, x_{2n},0,0)$ and
%$F_{n}(1)$ is $(x_{1},\dots, x_{2n}) \mapsto (0,0,x_{1},\dots, x_{2n})$.

\begin{proof}
%We describe an explicit well-defined infinite product of infinite elementary symplectic 
%matrices providing the $\Aff^{1}$-homotopy of symplectic 
%endomorphisms of $(\OO^{\oplus 2},\omega_{2})^{\oplus \infty}$.
%
%When we have a sequence of endomorphisms $f_{1},f_{2},f_{3},\dots$ of a set $V$
%such that for all $v \in V$ there is an $N(v) \in \NN$ such that $f_{n}(v) = v$ for all $n > N(v)$,
%then we may define the infinite composition $F = f_{1} \circ f_{2} \circ f_{3} \circ \cdots$ by the formula
%$F(v) = f_{1} \circ f_{2} \circ \cdots \circ f_{N(v)} (v)$.  If the $f_{n}$ are all injective, then so is $F$.
%For example the infinite composition of bijections $\NN \to \NN$ given by
%\[
%F = (1\,2) \circ (2\,3) \circ (3\,4) \circ (4\,5) \circ \cdots
%\]
%is the shift map of $\NN$ with $F(n) = n+1$.

Any matrix in any $Sp_{2r}(\ZZ)$ is a product of elementary symplectic matrices.  So there
exists an $M(t) \in Sp_{4}(\ZZ[t])$ with 
%
\begin{align*}
M(0) & = 
\begin{pmatrix}
1 & 0 & 0 & 0 \\0 & 1 & 0 & 0 \\0 & 0 & 1 & 0 \\0 & 0 & 0 & 1
\end{pmatrix},
&
M(1) & =
\begin{pmatrix}
0 & 0 & 1 & 0 \\0 & 0 & 0 & 1 \\1 & 0 & 0 & 0 \\0 & 1 & 0 & 0
\end{pmatrix},
\end{align*}
%
for example
\[
M(t) =
\begin{pmatrix}
1-t^{2} & 0 & -2t+13t^{3}-14t^{5}+4t^{7} & 8t^{2}-12t^{4}+4t^{6}
\\
0 & 1-t^{2} & -2t^{2}+2t^{4} & -t + 2t^{3}
\\
t & 0 & 1-7t^{2}+10t^{4}-4t^{6} & -4t + 8t^{3} -4t^{5}
\\
0 & 2t-t^{3} & 2t-4t^{3}+2t^{5} & 1 - 3t^{2} + 2t^{4}
\end{pmatrix}.
\]

Consider the sequence of infinite matrices
\[
f_{n}(t) =
\begin{pmatrix}
I_{2n-2} & 0 & 0 \\ 0 & M(t) & 0 \\ 0 & 0 & I_{\infty}
\end{pmatrix}.
\] 
Then $f_{n}(1)$ acts on  $(\OO^{\oplus 2},\omega_{2})^{\oplus \infty}$ by exchanging the
$n^{\text{th}}$ and $(n+1)^{\text{st}}$ summands $(\OO^{\oplus 2},\omega_{2})$, while
$f_{n}(0)$ is the identity.  The infinite product
\[
F(t) = f_{1}(t) f_{2}(t) f_{3}(t) \cdots 
\]
is well-defined
because the first $2n$ columns of $f_{1}f_{2}\cdots f_{N}$ are independent of $N$ for all $N \geq n$.
Each column of $F(t)$ contains only a finite number of nonzero entries.  (More precisely, writing
$F(t) = (a_{ij}(t))$, we have $a_{i,2n-1}(t) = a_{i,2n}(t)= 0$ for $i > 2n+2$.)
The endomorphism $F(t)$ is 
preserves the symplectic form $\omega_{2}^{\oplus \infty}$ because the automorphisms $f_{n}(t)$ all do.
Clearly $F(0)$ is the identity.  The finite product $f_{1}(1)f_{2}(1) \cdots f_{N}(1)$ permutes cyclically
the first $N+1$ summands of $(\OO^{\oplus 2},\omega_{2})^{\infty}$ and fixes the others.  
The infinite product $F(1)$ is the shift map.
\end{proof}

\begin{thm}
\label{T:MSp=BSp/BSp}
In the motivic unstable homotopy category $H_{\bullet}(S)$ we have a commutative diagram
\[
\xymatrix @M=5pt {
& BSp_{2r} \ar@/_/[ld]_-{\text{\textup{structure map}}} \ar@/^/[rd]^-{\text{\textup{quotient}}}
\\
\MSp_{2r} \ar@{<->}[rr]^-{\cong}_-{\text{\textup{$\Aff^{N}$-bundles and excision}}} 
&& BSp_{2r}/BSp_{2r-2}.
}
\]
\end{thm}



\section{The quaternionic projective bundle theorem}
\label{S:projective.bundle}


The most basic form a symplectic orientation is a symplectic Thom structure
\cite[Definition 7.1]{Panin:2010fk}.  The version of the definition for bigraded 
$\epsilon$-commutative theories is as follows.

\begin{defn}
\label{D:sp.thom}
A \emph{symplectic Thom structure} on a bigraded $\epsilon$-commutative
ring cohomology theory $(A^{*,*},\partial,\times,1_{A})$ on $\SmOp/S$ is a
rule which assigns to each rank $2$ symplectic bundle $(E,\phi)$ over an $X$ 
in $\Sm/S$ an element
$\thom(E,\phi) \in A^{4,2}(E,E-X)$ with the following properties: 
%
\begin{enumerate}
\item For an isomorphism $u \colon (E,\phi) \cong (E_{1},\phi_{1})$ 
one has $\thom(E,\phi) = u^{*}\thom(E_{1},\phi_{1})$.

\item For a morphism $f \colon Y \to X$ with pullback map $f_{E} \colon f^{*}E \to E$ 
one has $f_{E}^{*}\thom(E,\phi) = \thom(f^{*}E,f^{*}\phi)$.

\item For the trivial rank $2$ bundle $\Aff^{2} \to \pt$ with the symplectic form
$\omega_{2} =  \bigl( \begin{smallmatrix} 0 & 1 \\ -1 & 0 \end{smallmatrix} \bigr)$
the map 
\[
{-}\times \thom(\Aff^{2},\omega_{2}) \colon 
A^{*,*}(X) \to A^{*+4,*+2}(X \times \Aff^{2},X \times (\Aff^{2}-0))
\]
is an isomorphism for all $X$.
\end{enumerate}
%
The \emph{Pontryagin class} of $(E,\phi)$ is $p_{1}(E,\phi) = -z^{*} \thom(E,\phi) \in A^{4,2}(X)$ where
$z \colon X \to E$ is the zero section.
\end{defn}

The quaternionic projective bundle theorem is proven in \cite{Panin:2010fk} using the symplectic Thom 
structure and not any other version of a symplectic orientation.  It is proven first for trivial bundles.

\begin{thm}[\protect{\cite[Theorem 8.1]{Panin:2010fk}}]
\label{T:H.proj.bdl.1}
Let $(A^{*,*},\partial,\times,1_{A})$ be a bigraded $\epsilon$-commutative ring cohomology theory with 
a symplectic Thom structure.
Let $(\mathcal U_{HP^{n}}, \phi_{HP^{n}})$ be the tautological rank $2$
symplectic subbundle over $HP^n$ and 
$\zeta = p_{1}(\mathcal U_{HP^{n}}, \phi_{HP^{n}})$ its Pontryagin class. 
Then for any $X$ in $\Sm/S$ we have an isomorphism of bigraded rings
\[
A^{*, *}(HP^n \times X) \cong A^{*,*}(X)[\zeta]/(\zeta^{n+1}).
\]
\end{thm}


A Mayer-Vietoris argument gives the more general theorem \cite[Theorem 8.2]{Panin:2010fk}.

\begin{thm}[\protect{Quaternionic projective bundle theorem}]
\label{T:H.proj.bdl.2}
Let $(A^{*,*},\partial,\times,1_{A})$ be a bigraded $\epsilon$-commutative ring cohomology theory with 
a symplectic Thom structure.
Let $(E,\phi)$  be a symplectic bundle of rank $2n$ over $X$, let
$(\mathcal U, \phi|_{\mathcal U})$ be the tautological rank $2$
symplectic subbundle over
the quaternionic projective bundle $HP(E, \phi)$, and let
$\zeta = p_{1}(\mathcal U, \phi|_{\mathcal U})$
be its Pontryagin class. Then
we have an isomorphism of bigraded $A^{*,*}(X)$-modules
\[
(1, \zeta, \dots , \zeta^{n-1}) \colon
 A^{*,*}(X) \oplus A^{*,*}(X) \oplus \dots \oplus
A^{*,*}(X) \to A^{*,*}(HP(E, \phi)).
\]
\end{thm}

\begin{defn}
\label{D:PontryaginClasses}
Under the hypotheses of Theorem
\ref{T:H.proj.bdl.2}
there are unique elements
$p_i(E, \phi) \in A^{4i,2i}(X)$ for $i=1,2, \dots , n$
such that
$$\zeta^{n}-p_1(E, \phi)\cup \zeta^{{n-1}} + p_2(E, \phi)\cup \zeta^{{n-2}} - \dots + (-1)^n p_n(E, \phi)=0.$$
The classes
$p_i(E, \phi)$ 
are called the \emph{Pontryagin classes}
of $(E, \phi)$ with respect to the symplectic Thom structure of the cohomology theory $(A, \partial)$.
For $i > n$ one sets $p_i(E, \phi)$ = 0, and one sets $p_0(E, \phi) = 1$.
\end{defn}

For a rank $2$ symplectic bundle $(E,\phi)$ the classes $p_{1}(E,\phi)$ defined by Definitions 
\ref{D:sp.thom}   and \ref{D:PontryaginClasses} coincide.

\begin{cor}
\label{C:pont.triv}
The Pontryagin classes of a trivial symplectic bundle vanish.
\end{cor}


Among the consequences of the quaternionic projective bundle theorem is the symplectic splitting
principle \cite[Theorem 10.2]{Panin:2010fk}. 
We used it to prove the Cartan sum formula for Pontryagin classes \cite[Theorem 10.5]{Panin:2010fk}.

\begin{thm}
%[\protect{Cartan sum formula }]
\label{T:Cartan} 
Let $(A^{*,*},\partial,\times,1_{A})$ be a bigraded $\epsilon$-commutative ring cohomology theory with 
a symplectic Thom structure.
Suppose $(F, \psi) \cong (E_{1},\phi_{1}) \oplus (E_{2},\phi_{2})$ is an orthogonal direct sum of symplectic bundles over an $X$ in $\Sm/S$.  Then for all $i$ we have
%
\begin{equation}
\label{E:sum.formula}
p_{i}(F,\psi) = p_{i}(E_{1},\phi_{1}) + \sum_{j=1}^{i-1} p_{i-j}(E_{1},\phi_{1}) p_{j}(E_{2},\phi_{2}) + 
p_{i}(E_{2},\phi_{2}).
\end{equation}
\end{thm}

The quaternionic projective bundle theorem also allowed us to compute the cohomology of
quaternionic Grassmannians.  To explain our results we need to recall a number of facts 
about symmetric polynomials. 
They may be found in  for example \cite[Chap.\ 1, \S\S 1--3]{Macdonald:1995zl}.

Let $\Lambda_{r} \subset \ZZ[x_{1},\dots,x_{r}]$ be the ring of symmetric polynomials in $r$ variables. 
Let $e_{i}$ denote the $i^{\text{th}}$ elementary symmetric polynomial, and 
$h_{i}$ the $i^{\text{th}}$ \emph{complete symmetric polynomial}, the sum of all the monomials of degree $i$.
Set $e_{0} = h_{0} = 1$ 
and $e_{i} = h_{i} = 0$ for $i < 0$ and also $e_{i} = 0$ for $i > r$.
%For $i > r$ we have $e_{i} = 0$ but $h_{i} \neq 0$.
We have $\Lambda_{r} = \ZZ[e_{1},\dots,e_{r}]$.
% with $e_{1},\dots,e_{r}$ algebraically independent.
There is a recurrence relation
% generating functions are
%%
%\begin{align}
%\label{E:EH}
%E(t) & = \sum_{i \geq 0} e_{i}t^{i} = \prod_{j=1}^{r}(1+x_{i}t), & 
%H(t) & = \sum_{i \geq 0} h_{i}t^{i} = \prod_{j=1}^{r}(1-x_{i}t)^{-1}.
%\end{align}
%%
%So we have 
%%
%\begin{equation}
%\label{E:H.recur}
%E(t)H(-t) & = 1, &
$h_{m} + \sum_{i=1}^{r} (-1)^{r} e_{i}h_{m-i}  = 0$.
%\end{equation}
%
%So we also have $\Lambda_{r} = \ZZ[h_{1},\dots,h_{r}]$. 
%%with $h_{1},\dots,h_{r}$ algebraically independent.
%The $h_{i}$ with $i > r$ are nonzero but are dependent on $h_{1},\dots,h_{r}$.

Let
\[
\Pi_{r} = \{ \text{partitions $\lambda = (\lambda_{1},\lambda_{2},\dots,\lambda_{r})$ 
of length $l(\lambda)  \leq r$} \}.
\]
Write $\delta = (r-1,r-2, \dots,1,0)$.
For $\lambda \in \Pi_{r}$ let
$a_{\lambda + \delta} = \det(x_{i}^{\lambda_{j}+r-j})_{1\leq i,j \leq r}$.  Then $a_{\lambda+\delta}$ is
a skew-symmetric polynomial and therefore divisible by the Vandermonde determinant $a_{\delta}$.
The quotient $s_{\lambda} = a_{\lambda+\delta}/a_{\delta}$ is 
the \emph{Schur polynomial} for $\lambda$.
It is symmetric of degree $\lvert \lambda \rvert = \sum \lambda_{i}$.  
One has $s_{(1^{i})} = e_{i}$ and $s_{(i)} = h_{i}$.
The $a_{\lambda+\delta}$ with $l(\lambda) \leq r$ form a $\ZZ$-basis of the skew-symmetric polynomials
in $r$ variables, so the $s_{\lambda}$ with $l(\lambda) \leq r$ form a $\ZZ$-basis of $\Lambda_{r}$.
Denote by $\lambda'$ the partition dual to $\lambda$.  We have formulas
%
\begin{equation}
\label{E:schur.det}
s_{\lambda} = \det(e_{\lambda'_{i}-i+j})_{1 \leq i,j \leq m} 
= \det(h_{\lambda_{i}-i+j})_{1\leq i,j, \leq r},
\end{equation}
%
for $m \geq l(\lambda')$ and $r \geq l(\lambda)$.
%
%The ring of \emph{symmetric functions} $\Lambda$ in an infinite number of variables is the 
%inverse limit of the $\Lambda_{r}$ in the 
%category of graded rings.  It is a polynomial ring in infinitely many indeterminates
%\[
%\Lambda = \ZZ[e_{1},e_{2}, \dots] = \ZZ[h_{1},h_{2}, \dots]
%\]
%It has an involution $\omega$ with $\omega(e_{i}) = h_{i}$ and $\omega(h_{i}) = e_{i}$ for all $i$
%and $\omega(s_{\lambda}) = s_{\lambda'}$ for all partitions $\lambda$.
%The quotient map $\Lambda \onto \Lambda_{r}$ is the quotient
%by the ideal $(e_{r+1},e_{r+2}, \dots)$.  The ring $\Lambda$
%has as $\ZZ$-basis the Schur functions
%$s_{\lambda}$ with $\lambda$ ranging over all partitions.  If, however,
%$l(\lambda) = \lambda'_{1} > r$, then the first row of the first determinant of \eqref{E:schur.det} is 
%$(e_{\lambda'_{1}}, e_{\lambda'_{1}}+1, \dots, e_{\lambda'_{1}}+m-1)$.  All these entries
%are sent to $0$ in $\Lambda_{r}$.  
%%
Set
\[
\Pi_{r,n-r} = \{ \text{partitions $\lambda$ of length $l(\lambda) = \lambda'_{1} \leq r$ 
and with  $\lambda_{1} \leq n-r$} \}
\]
The set $\Pi_{r,n-r}$ has $\binom{n}{r}$ members.  
We will use the following results.

\begin{prop}
\label{P:sym.func.1}
The quotient map 
\[\ZZ[e_{1},\dots,e_{r}] \to \ZZ[e_{1},\dots,e_{r}]/(h_{n-r+1},\dots,h_{n})
\]
sends
$\{s_{\lambda} \mid \lambda \in \Pi_{r} - \Pi_{r,n-r} \} \mapsto 0$,
and it sends 
$\{s_{\lambda} \mid \lambda \in \Pi_{r,n-r} \}$ onto a homogeneous $\ZZ$-basis of the quotient ring.
\end{prop}

\begin{proof}
%[Sketch of proof]
For $\lambda \in \Pi_{r} - \Pi_{r,n-r}$ the first line of the determinant
$s_{\lambda} = \det(h_{\lambda_{i}-i+j})$ consists of $h_{k}$ with $k \geq n-r+1$.  These are all
in $(h_{n-r+1},\dots,h_{n})$ because of the recurrence relation satisfied by the $h_{k}$.  So 
they are sent to $0$ in the quotient.

The rank of the quotient as a $\ZZ$-module is $\prod \deg h_{i} / \prod \deg e_{i} = 
\binom{n}{r}$.  Since this is the same as the cardinality of $\Pi_{r,n-r}$, and since the images of 
the $s_{\lambda}$ with $\lambda \in \Pi_{r,n-r}$ generate the quotient as a $\ZZ$-module,
they form a $\ZZ$-basis of the quotient.
\end{proof}

Let the $\overline e_{i}$ and the $\overline h_{i}$ be, respectively, the elementary and complete
symmetric polynomials in $r-1$ variables.  The natural quotient map sends 
$e_{i} \mapsto \overline e_{i}$ for $i < r$ and $e_{r} \mapsto 0$, while it sends
$h_{i} \mapsto \overline h_{i}$ for all $i$.

\begin{prop}
\label{P:sym.func.2}
The kernel of the quotient map
\[
\ZZ[e_{1},\dots,e_{r}]/(h_{n-r+1},\dots,h_{n})
\to
\ZZ[\overline e_{1},\dots, \overline e_{r-1}]/(\overline h_{n-r+1},\dots, \overline h_{n-1})
\to 0
\]
is the image of the injection
\[
0 \to
\ZZ[e_{1},\dots,e_{r}]/(h_{n-r},\dots,h_{n-1})
\xra{ e_{r} \times {-}}
\ZZ[e_{1},\dots,e_{r}]/(h_{n-r+1},\dots,h_{n}).
\]
\end{prop}

\begin{proof}
The kernel is the free $\ZZ$-module with basis 
$\{ s_{\lambda} \mid \lambda \in \Pi_{r,n-r} - \Pi_{r-1,n-r} \}$.  These are the $\lambda$
with $\lambda_{r} \geq 1$ and thus $\lambda'_{1} = r$.  The formula
$s_{\lambda} = \det(e_{\lambda'_{i}-i+j})$ shows that for such $\lambda$ one has
$s_{\lambda} = e_{r}s_{\mu}$ with $\mu = (\lambda_{1}-1,\dots,\lambda_{r}-1) \in \Pi_{r,n-r-1}$.
These $s_{\mu}$ form a basis of the ring on the left of the second displayed line.
\end{proof}

%%  
%For the ideal $I_{n-r} = (h_{n-r+1},h_{n-r+2},\dots,h_{n}) \subset  \Lambda_{r}$
%the quotient map $\Lambda_{r} \onto \Lambda_{r}/I_{n-r}$ sends 
%$s_{\lambda} \mapsto 0$ for all $\lambda \not\in \Pi_{r,n-r}$.  
%The quotient $\Lambda_{r}/I_{n-r}$ is free over $\ZZ$ 
%%of rank $\binom{n}{r}$ 
%with basis $\{ s_{\lambda} \mid \lambda \in \Pi_{r,n-r} \}$.
%
%Now suppose $(E,\phi)$ is a symplectic bundle of rank $2r$ over $S$.  
%Let $q \colon HFlag_{S}(E,\phi) \to S$ be the associated complete quaternionic flag bundle, let 
%\[
%q^{*}(E, \phi) \cong 
%(\shf U_{1},\psi_{1}) \perp (\shf U_{2}, \psi_{2}) \perp \cdots \perp (\shf U_{r},\psi_{r}).
%\]
%be the splitting of the pullback of $E$ as the orthogonal direct sum of the $r$ universal
%symplectic subbundles of rank $2$ on $HFlag_{S}(E,\phi)$, and let 
%$y_{i} = p(\shf U_{i},\psi_{i}) \in A(HFlag_{S}(E,\phi))$ be the Pontryagin roots of $(E,\phi)$.  
%For any partition $\lambda$ of length $l(\lambda) \leq r$ write
%%
%\begin{equation}
%\label{E:schur.p}
%s_{\lambda}(E,\phi) = s_{\lambda}(y_{1},\dots,y_{r})
%= \det(p_{\lambda'_{i}-i+j}(E,\phi))_{1 \leq i,j \leq m} \in A(S) 
%\end{equation}
%%
%with $m \geq l(\lambda')$.  
%
%
%We now complete the calculation of the cohomology of  $HFlag(1^{r};n)$
%begun in \S\ref{S:flag.asymp}.  
%
%\begin{thm}
%\label{T:flag}
%On $HFlag(1^{r};n)$ let $y_{1},\dots,y_{r}$ be the first Pontryagin classes of the $r$
%tautological rank $2$ subbundles.  Then  $A(HFlag(1^{r};n)) = A(k)[y_{1},\dots,y_{r}]/I_{r,n}$
%with
%%
%\begin{equation}
%\label{E:hn.1}
%I_{r,n} = (h_{n}(y_{1}),h_{n-1}(y_{1},y_{2}),\dots,h_{n-r+1}(y_{1},\dots,y_{r})),
%\end{equation}
%%
%and if we now write $h_{i}$ for the complete symmetric polynomial in all $r$ variables
%$y_{1},\dots,y_{r}$, then we also have
%%
%\begin{equation}
%\label{E:hn.2}
%%I_{r,n} = 
%A(HFlag(1^{r};n)) = A(k)[y_{1},\dots,y_{r}]/
%(h_{n},h_{n-1}, \dots, h_{n-r+1}).
%\end{equation}
%\end{thm}
%
%\begin{proof}
%Using the Cartan sum formula, we see that the classes $z_{i}$ of \eqref{E:z.i} are actually
%$z_{i} = y_{i}$.   Hence \eqref{E:complete} can be rewritten as 
%$P_{i}(y_{1},\dots,y_{i}) = h_{n-i+1}(y_{1},\dots,y_{i})$, and we get \eqref{E:hn.1}.  
%
%Since the complete symmetric polynomials are the sums of all monomials of a given degree, we
%have
%\[
%h_{m}(y_{1},\dots,y_{i-1},y_{i}) = h_{m}(y_{1},\dots,y_{i-1}) + y_{i}h_{m-1}(y_{1},\dots,y_{i-1},y_{i}).
%\]
%This formula can be used to show that one can add one variable to each of the generators of
%$I_{r,n}$ which have less than $r$ variables without changing the ideal.  One repeats this until
%all the generators have $r$ variables.  This gives \eqref{E:hn.2}.
%\end{proof}

\begin{thm}[\protect{\cite[Theorem 11.2]{Panin:2010fk}}]
\label{T:Grass}
Let $(\shf U_{r,n},\phi_{r,n})$ be the tautological symplectic bundle of rank $2r$ on $\HGr(r,n)$.
%, and let
%$(\shf U^{\perp},\psi)$ be its orthogonal complement.  
Then for any 
bigraded $\epsilon$-commutative ring cohomology theory $(A^{*,*},\partial,\times,1_{A})$ with 
a symplectic Thom structure and any $X$ in $\Sm/S$
% and any $X$ 
the map
%
\begin{equation}
\label{E:HGr.cohom.1}
%\gamma \colon
A^{*,*}(X)[e_{1},\dots,e_{r}] / (h_{n-r+1},\dots,h_{n}) \xrightarrow{\cong} A^{*,*}(\HGr(r,n) \times X)
\end{equation}
%
sending $e_{i} \mapsto p_{i}(\shf U_{r,n},\phi_{r,n})$ for all $i$
is an isomorphism of bigraded rings.
%, the map
%%
%\begin{equation}
%\label{E:HGr.cohom.2}
%A(S)^{\oplus \binom{n}{r}} \xrightarrow{(s_{\lambda}(\shf U,\phi))_{\lambda\in\Pi_{r,n-r}}} 
%A(\HGr(r,n) \times S)
%\end{equation}
%%
%is an isomorphism of two-sided $A(S)$-modules, and
%for all partitions $\lambda$ we have 
%%
%\begin{equation}
%\label{E:schur.complement}
%s_{\lambda}(\shf U,\phi) = (-1)^{\lvert \lambda \rvert} s_{\lambda'}(\shf U^{\perp},\psi).
%\end{equation}
\end{thm}



%\begin{proof}
%The proof is much like the case of ordinary Grassmannians.
%It is enough to consider $S = k$.
%%Let $\gamma \colon A(k)[e_{1},\dots,e_{r}] \to A(\HGr(r,n))$ be the $A(k)$-algebra map sending 
%%$e_{i}\mapsto p_{i}(\shf U,\phi)$ for all $i$.  
%Write $A_{r,n} = A(k)[e_{1},\dots,e_{r}]/(h_{n-r+1},\dots,h_{n})$.
%Since $A(\HGr(r,n)) \to A(HFlag(1^{r};n))$ is injective, the complete symmetric polynomials 
%$h_{i}$ with $i > n-r$ vanish in $A(\HGr(r,n))$ as well.  We thus get the map
%$\gamma \colon A_{r,n} \to A(\HGr(r,n))$ sending $e_{i} \mapsto p_{i}(\shf U,\phi)$.
%By Proposition \ref{P:Flag.Gr}
%\[
%\mathbb B_{r} = 
%\{{y}_{1}^{a_{1}}{y}_{2}^{a_{2}} \cdots{y}_{r-1}^{a_{r-1}} \mid \text{$0 \leq a_{i} \leq r-i$ for all $i$} \}.
%\]
%is a basis of $A(HFlag(1^{r};n))$ as a two-sided $A(\HGr(r,n))$-module, and it is also
%a basis of $\ZZ[y_{1},\dots,y_{r}]$ as a free module over the ring of
%symmetric polynomials $\ZZ[e_{1},\dots,e_{r}]$ and therefore also a basis
%of $A(k)[y_{1},\dots,y_{r}]/(h_{n-r+1},\dots,h_{n})$ as a two-sided free
%$A_{r,n}$-module.  By Theorem \ref{T:flag}
%the map of free modules $A(k)[y_{1},\dots,y_{r}]/(h_{n-r+1},\dots,h_{n}) \to A(HFlag(1^{r};n))$ is
%an isomorphism.  So $\gamma \colon A_{r,n} \to A(\HGr(r,n))$
%is also an isomorphism.
%%
%%Write the elements as $\mathbb B_{r}$ as $b_{1},\dots,b_{r!}$.
%%Then we have a commutative diagram
%%\[
%%\xymatrix @M=5pt @C=50pt {
%%\ZZ[e_{1},\dots,e_{r}]^{\oplus r!} \ar[d]^-{(b_{1},\dots,b_{r!})}_-{\cong} 
%%\ar[r]
%%& \left(\dfrac{A(k) [e_{1},\dots,e_{r}]}{(h_{n-r+1},\dots,h_{n})}\right)^{\oplus r!} 
%%\ar[d]^-{(b_{1},\dots,b_{r!})}_-{\cong} \ar[r] 
%%& A(\HGr(r,n))^{\oplus r!} \ar[d]^-{(b_{1},\dots,b_{r!})}_-{\cong}
%%\\
%%\ZZ[y_{1},\dots,y_{r}] \ar[r] 
%%& \dfrac{A(k) [y_{1},\dots,y_{r}]}{(h_{n-r+1},\dots,h_{n}) } \ar[r]^-{\cong}_-{\eqref{E:hn.2}} 
%%& A(HFlag(1^{r};n)) 
%%}
%%\]
%%with the three vertical arrows isomorphisms and with the bottom right horizontal arrow an
%%isomorphism by Theorem \ref{T:flag}.  The top right horizontal arrows is therefore also an 
%%isomorphism.
%
%
%Over the quaternionic Grassmannian $\HGr(r,n)$ the orthogonal direct sum 
%$(\shf U,\phi) \perp (\shf U^{\perp},\psi)$ is the trivial symplectic bundle $\mathsf{h}(\OO)^{\oplus n}$ and 
%has vanishing Pontryagin classes.  So the Cartan sum formula \eqref{E:sum.formula} gives
%$p_{t}(\shf U,\phi) p_{t}(\shf U^{\perp},\psi) = 1$.  
%For the generating series of \eqref{E:EH}
%we have $\gamma(E(t)) = p_{t}(\shf U,\phi)$.
%So from the identity $E(t)H(-t) = 1$ of \eqref{E:H.recur}, we see we have 
%$\gamma(H(-t)) = p_{t}(\shf U^{\perp},\psi)$ and thus 
%$(-1)^{i} \gamma(h_{i}) =  p_{i}(\shf U^{\perp},\psi)$.  
%The formula $s_{\lambda}(\shf U,\phi) = (-1)^{\lvert \lambda \rvert} s_{\lambda'}(\shf U^{\perp},\psi)$
%now follows from \eqref{E:schur.det}.  
%The sign change $h_{i} \mapsto (-1)^{i}h_{i}$ comes from the
%involution of $\Lambda_{r}$ sending $f \mapsto (-1)^{\deg(f)}f$ for all homogeneous symmetric
%polynomials.  Hence the sign $(-1)^{\lvert \lambda \rvert}$ in front of 
%$s_{\lambda'}(\shf U^{\perp},\psi)$.
%\end{proof}




%The usual Mayer-Vietoris argument now gives the following generalization of Theorem \ref{T:Grass}.
%
%\begin{thm}
%\label{T:Grass.bdl}
%Let $(E,\phi)$ be a symplectic bundle of rank $2n$ over $S$.  Let $(\shf U,\phi|_{\shf U})$ be
%the tautological subbundle of rank $2r$ on $\HGr_{S}(r,E,\phi)$, and let 
%$(\shf U^{\perp}, \phi|_{U^{\perp}})$ be its orthogonal complement.  Then for any
%ring cohomology theory with a symplectic Thom structure
%%
%\begin{gather*}
%A(S)^{\oplus \binom{n}{r}} 
%\xrightarrow[\cong]{(s_{\lambda}(\shf U,\phi|_{\shf U}))_{\lambda \in \Pi_{r,n-r}}}
%A(\HGr_{S}(r,E,\phi)), \\ 
%A(S) ^{\oplus \binom{n}{r}}
%\xrightarrow[\cong]{(s_{\lambda}(\shf U^{\perp},\phi|_{\shf U^{\perp}}))_{\lambda \in \Pi_{n-r,r}}}
%A(\HGr_{S}(r,E,\phi))
%\end{gather*}
%are isomorphisms of two-sided $A(S)$-modules.
%\end{thm}

%Let $(\shf U_{r,n},\phi_{r,n}) \hra \mathsf{h}(\OO)^{\oplus n}$ be the universal tautological symplectic
%bundle of rank $2r$ on $\HGr(r,n)$.  Let $\eta_{r,n} \colon \shf U_{r,n} \subset \mathsf{h}(\OO)^{\oplus n}$
%be the inclusion.  Consider the inclusions of symplectic subbundles
%%
%\begin{equation}
%\label{E:inclusions}
%\vcenter{
%\xymatrix @M=5pt @C=40pt @R=10pt {
%\shf U_{r,n}  \ar@{^{(}->}[r]^-{
%\left(
%\begin{smallmatrix}
%\eta_{r,n} \\0
%\end{smallmatrix}
%\right)
%} &
%\mathsf{h}(\OO)^{\oplus n} \oplus \mathsf{h}(\OO) = \mathsf{h}(\OO)^{\oplus n+1},
%%}
%\\
%%\xymatrix @M=5pt @C=40pt {
%\mathsf{h}(\OO) \oplus \shf U_{r,n} \ar@{^{(}->}[r]^-{
%\left(
%\begin{smallmatrix}
%1 & 0\\ 0 & \eta_{r,n}
%\end{smallmatrix}
%\right)
%} &
%\mathsf{h}(\OO) \oplus \mathsf{h}(\OO)^{\oplus n} = \mathsf{h}(\OO)^{\oplus n+1}.
%}}
%\end{equation}
%%
%They are classified by maps $\alpha_{r,n} \colon \HGr(r,n) \to \HGr(r,n+1)$ and 
%$\beta_{r,n} \colon \HGr(r,n) \to \HGr(r+1,n+1)$ respectively.  Let
%$\gamma_{r,n} = \beta_{r,n+1}\alpha_{r,n} = \alpha_{r+1,n+1}\beta_{r,n} \colon
%\HGr(r,n) \to \HGr(r+1,n+2)$.  We have direct systems of quaternionic Grassmannians
%%
%\begin{equation}
%\label{E:systems}
%\begin{gathered}
%\HGr(r,r) \xrightarrow{\alpha_{r,r}} \HGr(r,r+1) \xrightarrow{\alpha_{r,r+1}}
%\HGr(r,r+2) \to \cdot \to \HGr(r,n) \xrightarrow{\alpha_{r,n}} \cdots
%\\
%\HGr(0,0) \xrightarrow{\gamma_{0,0}} \HGr(1,2) \xrightarrow{\gamma_{1,2}} \HGr(2,4) \to \cdots \to \HGr(n,2n) \xrightarrow{\gamma_{n,2n}} \cdots
%\end{gathered}
%\end{equation}



\begin{thm}[\protect{\cite[Theorem 11.4]{Panin:2010fk}}]
\label{T:HGr.lim.cohom.1}
Let $\alpha_{r,n} \colon \HGr(r,n) \hra \HGr(r,n+1)$ 
%and $\beta_{r,n} \colon \HGr(r,n)  \to \HGr(r+1,n+1)$
be the usual inclusion.
For any bigraded $\epsilon$-commutative ring cohomology theory $(A^{*,*},\partial,\times,1_{A})$ with 
a symplectic Thom structure and any $X$ in $\Sm/S$
%and any $X$ 
the map
%
\begin{equation*}
(\alpha_{r,n} \times 1)^{*} \colon A^{*,*}(\HGr(r,n+1) \times X) \to A^{*,*}(\HGr(r,n) \times X)
%\\
%(\beta_{r,n} \times 1)^{*} \colon A^{*,*}(\HGr(r+1,n+1) \times X) \to A^{*,*}(\HGr(r,n) \times X)
\end{equation*}
%
is a surjection which
the isomorphisms \eqref {E:HGr.cohom.1} identify 
with the natural surjection
\[
A^{*,*}(X)[e_{1},\dots,e_{r}] / (h_{n-r+2},\dots,h_{n},h_{n+1})
\to
A^{*,*}(X)[e_{1},\dots,e_{r}] / (h_{n-r+1},h_{n-r+2},\dots,h_{n}).
\]
\end{thm}

\begin{thm}[\protect{\cite[Theorem 11.4]{Panin:2010fk}}]
\label{T:HGr.lim.cohom.2}
Let 
%$\alpha_{r,n} \colon \HGr(r,n) \hra \HGr(r,n+1)$ and 
$\beta_{r,n} \colon \HGr(r,n)  \to \HGr(1+r,1+n)$
be the usual inclusion.
For any bigraded $\epsilon$-commutative ring cohomology theory $(A^{*,*},\partial,\times,1_{A})$ with 
a symplectic Thom structure and any $X$ in $\Sm/S$
%and any $X$ 
the map
%
\begin{equation*}
(\beta_{r,n} \times 1)^{*} \colon A^{*,*}(\HGr(1+r,1+n) \times X) \to A^{*,*}(\HGr(r,n) \times X)
%\\
%(\beta_{r,n} \times 1)^{*} \colon A^{*,*}(\HGr(r+1,n+1) \times X) \to A^{*,*}(\HGr(r,n) \times X)
\end{equation*}
%
is a surjection which
the isomorphisms \eqref {E:HGr.cohom.1} identify 
with the surjection
\[
A^{*,*}(\pt)[e_{1},\dots,e_{r},e_{r+1}] / (h_{n-r+1},\dots,h_{n},h_{n+1})
\to
A^{*,*}(\pt)[e_{1},\dots,e_{r}] / (h_{n-r+1},\dots,h_{n})
\]
of $A^{*,*}(\pt)[e_{1},\dots,e_{r}]$-algebras sending $e_{r+1} \mapsto 0$.
\end{thm}


\section{The cohomology of 
\texorpdfstring{$BSp_{2r}$ and $\MSp_{2r}$}
{BSp\_2r and MSp\_2r}
}
\label{S:cohom.BSp.MSp}

\begin{thm}
\label{T:cohom.BSp}
Let $(A,\mu,e)$ be a commutative $T$-ring spectrum with a symplectic Thom structure on $A^{*,*}$.  
Then the isomorphism $BSp_{2r} = \colim_{n} HGr(r,n)$ induces isomorphisms
%
\begin{equation*}
%\label{E:A(BSp.2r)}
A^{*,*}(BSp_{2r})
\xrightarrow{\cong} 
\varprojlim\limits_{n \to \infty} A^{*,*}(\HGr(r,n) )
\xleftarrow{\cong} 
A^{*,*}(\pt) [[p_{1},\dots,p_{r}]]^{\homog} 
%\\
%\label{E:HGr(infty,2.infty)}
%A(S)[[p_{1},p_{2},  p_{3}, \dots ]]
%\xrightarrow{\cong} 
%\varprojlim
%\limits_{n\to \infty} 
%A(\HGr(n,2n) \times S) 
\end{equation*}
%
of bigraded rings.  
The second isomorphism
sends the variable $p_{i}$ to the inverse system of $i^{\text{th}}$ Pontryagin classes 
$(p_{i}(\shf U_{r,n}))_{n \geq r}$.
\end{thm}

%The theorem follows from the explicit generators and relations for the $A(\HGr(r,n) \times S)$
%given in Theorem \ref{T:Grass} in the same way as for ordinary Grassmannians.


Here $A^{*,*}(\pt) [[p_{1},\dots,p_{r}]]^{\homog}$ is the bigraded ring of homogeneous power series.
The $\varprojlim$ is taken in the category of bigraded rings.

\begin{proof}
We have $BSp_{2r} = \colim_{n} HGr(r,n)$, so by Theorem \ref{T:hocolim} we have an exact sequence
\[
0 \to \sideset{}{^{1}}\varprojlim_{n\in\NN} A^{*-1,*}(HGr(r,n))
\to A^{*,*}(BSp_{2r}) \to 
\varprojlim_{n\in\NN} A^{*,*}(HGr(r,n)) \to 0.
\]
The connecting maps are the $\alpha_{r,n}^{*}$ of Theorem \ref {T:HGr.lim.cohom.1}, which are surjective.
So the $\varprojlim^{1}$ vanishes, and the first map of the statement of the theorem is an isomorphism.
  
Let $I_{d} \subset A^{*,*}(\pt)[p_{1},\dots,p_{r}]$ be the two-sided ideal generated by the monomials
$p_{1}^{a_{1}}p_{2}^{a_{2}} \dots p_{r}^{a_{r}}$ with $\sum i a_{i} \geq d$.  We have inclusions
\(
I_{r(n-r)+1} \subset (h_{n-r+1},\dots,h_{n}) \subset I_{n-r+1}
\)
because all partitions $\lambda \in \Pi_{r,n-r}$ have 
$\lvert \lambda \rvert \leq r(n-r)$.
So we have
%
\begin{multline*}
\varprojlim_{n} A^{*,*}(HGr(r,n)) \cong 
\varprojlim_{n} A^{*,*}(\pt)[p_{1},\dots,p_{r}]/(h_{n-r+1},\dots,h_{n})
\\ \cong \varprojlim_{d} A^{*,*}(\pt)[p_{1},\dots,p_{r}]/I_{d}
= A^{*,*}(\pt) [[p_{1},\dots,p_{r}]]^{\homog}.
\tag*{\qedhere}
\end{multline*}
\end{proof}

\begin{thm}
\label{T:cohom.MSp.2r}
Let $(A,\mu,e)$ be a commutative $T$-ring spectrum with a symplectic Thom structure on $A^{*,*}$.  
Then for any $r$ and $n$ the motivic homotopy equivalence 
\[
\Th \shf U_{HGr(r,n)} \cong HGr(r,1+n)/HGr(r-1,n)
\]
of Theorem \ref{T:BSp/BSp} induces isomorphisms
%
\begin{equation}
\label{E:cohom.Thom}
A^{*,*}(\pt)[p_{1},\dots,p_{r}]/(h_{n-r+1},\dots,h_{n})
\xra[\cong]{\cup p_{r}}
A^{*+4r,*+2r}(\Th \shf U_{HGr(r,n)})
\end{equation}
%
of two-sided bigraded modules over 
$A^{*,*}(HGr(r,n)) \cong A^{*,*}(\pt)[p_{1},\dots,p_{r}]/(h_{n-r+1},\dots,h_{n})$.
Moreover, the isomorphism $\MSp_{2r} = \colim_{n} \Th \shf U_{HGr(r,n)}$ induces
isomorphisms of bigraded modules over 
$A^{*,*}(BSp_{2r}) \cong A^{*,*}(\pt)[[p_{1},\dots,p_{r}]]^{\homog}$
%
\begin{equation}
\label{E:A(MSp.2r)}
A^{*+4r,*+2r}(\MSp_{2r}) 
\xra{\cong}
\varprojlim\limits_{n \to \infty} A^{*+4r,*+2r}(\Th \shf U_{HGr(r,n)} )
\xla{\cong}
p_{r} A^{*,*}(\pt) [[p_{1},\dots,p_{r}]]^{\homog} 
\end{equation}
\end{thm}

\begin{proof}
Applying $A^{*,*}$ to the cofiber sequence
\[
HGr(r-1,n) \xra{\beta_{r-1,n}} HGr(r,1+n) \lra \Th \shf U_{HGr(r,n)} 
\]
gives a long exact sequence of cohomology.  The map
$\beta_{r-1,n}$ induces a surjection of cohomology groups by Theorem \ref{T:HGr.lim.cohom.2}, so we
have $A^{*,*}(\Th \shf U_{HGr(r,n)}) \cong \ker \beta_{r-1,n}^{*}$.  
This kernel is identified by Proposition \ref{P:sym.func.2}, giving the isomorphism \eqref{E:cohom.Thom}.
In principle this is an isomorphism of two-sided modules over 
$A^{*,*}(HGr(r,1+n)) \cong A^{*,*}(\pt)[p_{1},\dots,p_{r}]/(h_{n-r+2},\dots,h_{n+1})$.  But the modules
are annihilated by $h_{n-r+1}$, so they are also two-sided modules over the quotient ring 
$A^{*,*}(HGr(r,n))$.

The inclusion $\Th \shf U_{HGr(r,n)} \hra \Th \shf U_{HGr(r,n+1)}$ induces a commutative
diagram
%
\begin{equation*}
\vcenter{\xymatrix @M=5pt @C=35pt {
A^{*,*}(\pt)[p_{1},\dots,p_{r}]/(h_{n-r+2},\dots,h_{n},h_{n+1})
\ar[r]_-{\cong}^-{\cup p_{r}} \ar@{->>}[d]
& A^{*+4r,*+2r}(\Th \shf U_{HGr(r,n+1)}) \ar@{->>}[d]
\\
A^{*,*}(\pt)[p_{1},\dots,p_{r}]/(h_{n-r+1},h_{n-r+1}\dots,h_{n})
\ar[r]_-{\cong}^-{\cup p_{r}}
& A^{*+4r,*+2r}(\Th \shf U_{HGr(r,n)}).
}}
\end{equation*}
%
The inverse limit gives the righthand isomorphism of \eqref {E:A(MSp.2r)}.
The vertical map on the left is surjective, so the vertical map on the right is as well.  
Therefore the $\varprojlim^{1}$ vanishes in the exact sequence
\[
0 \to \sideset{}{^{1}}\varprojlim_{n\in\NN} A^{*-1,*}(\Th \shf U_{HGr(r,n)})
\to A^{*,*}(\MSp_{2r}) \to 
\varprojlim_{n\in\NN} A^{*,*}(\Th \shf U_{HGr(r,n)}) \to 0.
\]
obtained from Theorem \ref{T:hocolim}.
\end{proof}


For any $r$ let $z_{2r} \colon BSp_{2r} \to \Th \shf U_{BSp_{2r}} = \MSp_{2r}$ 
be the structure map induced by
the zero section of the tautological symplectic bundle $\shf U_{BSp_{2r}} \to BSp_{2r}$.

\begin{thm}
\label{T:cohom.MSp.2r.ideal}
Let $(A,\mu,e)$ be a commutative $T$-ring spectrum with a symplectic Thom structure on $A^{*,*}$.  
The map $z_{2r}^{*} \colon A^{*,*}(\MSp_{2r}) \to A^{*,*}(BSp_{2r})$ of two-sided
$A^{*,*}(BSp_{2r})$-modules is injective
and identifies $A^{*,*}(\MSp_{2r})$ with the two-sided principal ideal generated by 
$p_{r} \in A^{4r,2r}(BSp_{2r})$.
\end{thm}

\begin{proof}
By Theorem \ref{T:MSp=BSp/BSp} we have cofiber sequence
$BSp_{2r-2} \xra{i_{2r}} BSp_{2r} \xra{z_{2r}} \MSp_{2r}$ yielding a long exact sequence of cohomology groups.  
By the previous theorems these are isomorphic to (in simplified notation)
%
\begin{equation}
\label{E:long.exact}
\cdots \to A^{*-4r,*-2r}[[p_{1},\dots,p_{r}]] \xra{z_{2r}^{*}}
A^{*,*}[[p_{1},\dots,p_{r}]] \xra{i_{2r}^{*}}
A^{*,*}[[p_{1},\dots,p_{r-1}]] \to \cdots
\end{equation}
%
Since $i_{2r}$ is the colimit of the inclusion maps $\beta_{r-1,n} \colon A(r-1,n) \to A(r,1+n)$
of Theorem \ref{T:HGr.lim.cohom.2}, $i_{2r}^{*}$ is the quotient by the ideal generated by $p_{r}$.
It is surjective in all bidegrees.  
It follows that $z_{2r}^{*}$ is injective in all bidegees and is the inclusion of that ideal.
\end{proof}




The direct sum of symplectic bundle induces compatible monoid structures on the
$BSp_{2r}$ and the $\MSp_{2r}$.  So the following diagram commutes for all $r$ and $s$
%
\begin{equation}
\label{E:mult.BSp.MSp}
\vcenter{\xymatrix @M=5pt @C=35pt {
BSp_{2r} \times BSp_{2s} \ar[r]^-{m_{rs}} \ar[d]_-{z_{2r} \times z_{2s}}
& BSp_{2r+2s} \ar[d]^-{z_{2r+2s}}
\\
\MSp_{2r} \wedge \MSp_{2s} \ar[r]^-{\mu_{rs}}
& \MSp_{2r+2s}.
}}
\end{equation}


\begin{thm}
\label{T:cohom.BSp.BSp}
Let $(A,\mu,e)$ be a commutative $T$-ring spectrum with a symplectic Thom structure on $A^{*,*}$.  
Then the isomorphisms 
%
\begin{align*}
BSp_{2r} \times BSp_{2s} & = \colim\nolimits_{n} \bigl( HGr(r,rn) \times HGr(s,sn) \bigr)
\\
\MSp_{2r} \wedge \MSp_{2s} & = \colim\nolimits_{n} 
(\Th \shf U_{HGr(r,rn)} \wedge \Th \shf U_{HGr(s,sn)})
\end{align*} 
induces a commutative diagram of isomorphisms and monomorphisms of two-sided graded 
$A^{*,*}(BSp_{2r} \times BSp_{2s})$-modules
%
\begin{small}
\begin{equation*}
%\label{E:A(BSp.2r)}
\xymatrix @M=5pt @C=10pt {
A^{*,*}(\MSp_{2r} \wedge \MSp_{2s})
\ar[r]^-{\cong} \ar@{_{(}->}[d]_-{(z_{2r} \times z_{2s})^{*}}
& 
\raisebox{-15pt}{$\varprojlim\limits_{n \to \infty} A^{*,*}(\Th \shf U_{r,rn} \wedge \Th \shf U_{s,sn})$}
\ar@{_{(}->}[d]
& \ar[l]_-{\cong}
p'_{r}p''_{s}A^{*,*} [[p'_{1},\dots,p'_{r},p''_{1},\dots,p''_{s}]]
\ar@{_{(}->}[d]^-{\text{\textup{inclusion}}}
\\
A^{*,*}(BSp_{2r} \times BSp_{2s})
\ar[r]^-{\cong} 
& 
\raisebox{-15pt}{$\varprojlim\limits_{n \to \infty} A^{*,*}(\HGr(r,rn) \times  \HGr(s,sn))$}
& \ar[l]_-{\cong}
A^{*,*} [[p'_{1},\dots,p'_{r},p''_{1},\dots,p''_{s}]].
}
%\\
%\label{E:HGr(infty,2.infty)}
%A(S)[[p_{1},p_{2},  p_{3}, \dots ]]
%\xrightarrow{\cong} 
%\varprojlim
%\limits_{n\to \infty} 
%A(\HGr(n,2n) \times S) 
\end{equation*}
\end{small}%
%
Moreover, these isomorphisms identify the diagram obtained by applying $A^{*,*}$ to
\eqref{E:mult.BSp.MSp} with the diagram of rings and ideals
\[
\xymatrix @M=5pt @C=30pt {
p_{r+s}A^{*,*}(\pt)[[p_{1},\dots,p_{r+s}]]^{\homog} \ar[r] \ar@{_{(}->}[d]
&
p'_{r}p''_{s} A^{*,*}(\pt) [[p'_{1},\dots,p'_{r},p''_{1},\dots,p''_{s}]]^{\homog} \ar@{_{(}->}[d]
\\
A^{*,*}(\pt)[[p_{1},\dots,p_{r+s}]]^{\homog} \ar[r] 
&
A^{*,*}(\pt) [[p'_{1},\dots,p'_{r},p''_{1},\dots,p''_{s}]]^{\homog}
}
\]
where the horizontal maps send $p_{i} \mapsto p'_{i} + \sum_{j=1}^{i-1} p'_{i-j}p''_{j}+ p''_{i}$.
Moreover, the horizontal maps of the last diagram are also injective.
\end{thm}

\begin{proof}
The construction of the first diagram is much the same as in the previous theorems. 
The second diagram follows.  For the last statement of the theorem, let
$t_{1},\dots,t_{r+s}$ be independent indeterminates of bidegree $(2,1)$.  Then the composition of
the bottom horizontal map with the map
\[
A^{*,*}(\pt) [[p'_{1},\dots,p'_{r},p''_{1},\dots,p''_{s}]]^{\homog}
\to
A^{*,*}(\pt) [[t_{1},\dots,t_{r+s}]]^{\homog}
\]
sending $p'_{i} \mapsto e_{i}(t_{1},\dots,t_{r})$ and $p''_{j} \mapsto e_{j}(t_{r+1},\dots,t_{r+s})$
is the inclusion of the ring of symmetric homogeneous power series in the ring of homogeneous
power series.  That is injective.
\end{proof}

The final calculation in this section ought to be that of $A^{*,*}(\MSp)$.  However, we will 
put this off until Theorem \ref{T:cohom.MSp} because we wish to make the calculation using
a symplectic Thom classes theory and not just a symplectic Thom structure.




\section{Tautological Thom elements}

Suppose that $(A,\mu,e)$ is a commutative $T$-ring spectrum.  
Let $\vartheta \in A^{4,2}(\MSp_{2})$.
%an element with $\vartheta |_{T^{\wedge 2}} = \Sigma_{T}^{2}1_{A} \in A^{4,2}(T^{\wedge 2})$.




We associate to $\vartheta$ and a symplectic bundle $(E,\phi)$ of rank $2$ over an
$X$ in $\Sm/S$ a class $\thom^{\vartheta}(E,\phi)$ defined as follows.  By assumption
the scheme $X$ admits an ample family of line bundles.  So there exists an
affine bundle $f \colon Y \to X$ with $Y$ an affine scheme.  Then for some $p$ there exist global sections
$s_{1},\dots,s_{p}$ of $f^{*}E^{\vee}$ generating $f^{*}E^{\vee}$.  There then exist global functions
$a_{ij}$ on $Y$ such that $f^{*} \phi = \sum_{1\leq i < j \leq p} a_{ij} s_{i} \wedge s_{j}$.
We set $t_{i} = \sum_{j=i+1}^{p} a_{ij}s_{j}$ so that we have
$\sum_{i} s_{i} \wedge t_{i} = f^{*}\phi$.  The map
$(s_{1},t_{1},\dots, s_{p},t_{p}) \colon f^{*}E \to \OO_{Y}^{\oplus 2p}$
%\[
%f^{*}E \xra{(s_{1},t_{1},\dots, s_{np},t_{np})} \OO_{Y}^{\oplus 2np}
%\]
embeds $(f^{*}E, f^{*}\phi)$ as a symplectic subbundle of $(\OO_{Y}^{\oplus 2p}, \omega_{2p})$.
So it is classified by a map $\psi \colon Y \to HGr(1,p) = HP^{p-1}$ such that 
$\psi^{*}(\TSp{1,p},\phi_{1,p}) = f^{*}(E,\phi)$.  
This gives us maps of (ind)-schemes
%
\begin{equation}
\label{E:X.to.BSp}
X \xla[\text{$\Aff^{N}$-bundle}]{f} Y
\xra{\psi}
HP^{p-1}
\xra{\text{inclusion}} BSp_{2} = HP^{\infty}
\end{equation}
%
and of pointed motivic spaces
%
\begin{equation}
\label{E:ThE.to.A.1}
\Th E \xla[\sim\text{mot}]{\overline f} \Th f^{*}E 
\, \cong \, \Th \widetilde\psi^{*} \TSp{1,p} 
\xra{\overline{\psi}} 
\Th \TSp{1,p} 
\end{equation}
%
of pointed motivic spaces, which can be composed with the maps
%
\begin{equation}
\label{E:ThE.to.A.2}
\Th \TSp{1,p} \xra{\text{inclusion}} 
 \MSp_{2} \xra{\vartheta} A \wedge T^{\wedge 2}.
\end{equation}
%
in $SH(S)$.
The composition of \eqref{E:ThE.to.A.1} and \eqref{E:ThE.to.A.2} gives a class
\[
\thom^{\vartheta}(E,\phi) \in Hom_{SH(S)}(\Th E, A \wedge T^{\wedge 2}) = A^{4,2}(E,E-X).
\]
The following lemma is proven in the same way as Lemma \ref{L:well.defined.SL}.

\begin{lem}
\label{L:well.defined.Sp.1}
The classes $\thom^{\vartheta}(E,\phi)$ depend only on the rank $2$ symplectic bundle $(E,\phi)$
and the morphism  $\vartheta \colon \Sigma_{T}^{\infty}\MSp_{2}(-2) \to A$ in $SH(S)$.
\end{lem}

Recall the inclusion $e_{2}^{Sp} \colon T^{\wedge 2} \to \MSp_{2}$ of \eqref{E:e.Sp}.


\begin{thm}
\label{T:a.alpha}
Let $(A,\mu,e)$ be a commutative $T$-ring spectrum.  
Then the map which assigns to a class $\vartheta$ as above the family of classes
$\thom^{\vartheta}(E,\phi)$ is a bijection between the sets of

\parens{$\alpha$} classes $\vartheta \in A^{4,2}(\MSp_{2})$ with 
$\vartheta |_{T^{\wedge 2}} = \Sigma_{T}^{2}1_{A}$ in $A^{4,2}(T^{\wedge 2})$, and

\parens{a} symplectic Thom structures on the bigraded $\epsilon$-commutative ring cohomology theory 
$(A^{*,*},\partial,\times,1_{A})$ such that for the trivial rank $2$
bundle $\Aff^{2} \to \pt$ we have 
$\thom(\Aff^{2},\omega_{2}) = \Sigma_{T}^{2}1_{A}$ in $A^{4,2}(T^{\wedge 2})$.
\end{thm}


\begin{proof}
A proof similar to that of Theorem \ref{T:thom.phi} shows that for a $\vartheta$ as in ($\alpha$),
the family of classes $\thom^{\vartheta}(E,\phi)$ form a symplectic Thom structure with the stated
normalization condition.  Note that this uses the fact that all symplectic bundles are locally
trivial in the Zariski topology.

Now suppose we have a symplectic Thom structure with the stated normalization condition.
For every $n$ the tautological rank $2$ symplectic bundle bundle over $HP^{n-1}$ has a Thom class
which we will abbreviate as 
\[
\thom_{n} =
\thom(\shf U_{HP^{n-1}},\phi_{HP^{n-1}}) \in A^{4,2}(\Th \shf U_{HP^{n-1}}).
\]
Pullback along the inclusion $\Th \shf U_{HP^{n-1}} \to \Th \shf U_{HP^{n}}$ sends
$\thom_{n+1} \mapsto \thom_{n}$.
%
So as $n$ varies, we get an element
\[
\bar\vartheta = (\thom_{n} )_{n\in\NN}
\in \varprojlim A^{4,2}(\Th \shf U_{HP^{n-1}}).
\]
We have $\MSp_{2} = \colim \Th \shf U_{HP^{n-1}}$, and by Theorem \ref{T:cohom.MSp.2r} 
the natural map 
%
\begin{equation*}
%\label{E:symp.thom.surj}
A^{*,*}(\MSp_{2}) \xra{\cong} \varprojlim A^{*,*}(\Th \shf U_{HP^{n-1}}).
\end{equation*}
%
is an isomorphism.
Let $\vartheta \in A^{4,2}(\MSp_{2})$ be the unique class lifting $\bar\vartheta$.  As in the proof of 
Theorem \ref{T:univ.SL} we have $\thom^{\vartheta}(E,\phi) = \thom(E,\phi)$ for all rank $2$
symplectic bundles.  Moreover, for $\vartheta$ and $\xi$ in $A^{4,2}(\MSp_{2})$ we have 
$\thom^{\vartheta}(E,\phi) = \thom^{\xi}(E,\phi)$ for all symplectic bundles if and only if
$\vartheta$ and $\xi$ have the same image in the inverse limit.  But that happens only for
$\vartheta = \xi$.
\end{proof}


\begin{defn}
The class $\vartheta \in A^{4,2}(\MSp_{2})$ is the \emph{tautological Thom element} 
of the symplectic orientation on $A^{*,*}$ whose
rank $2$ symplectic Thom classes are the $\thom^{\vartheta}(E,\phi)$.
\end{defn}

The canonical morphism $u_{2} \colon \Sigma_{T}^{\infty} \MSp_{2}(-2) \to \MSp$ which is part of the
counit of the adjunction between $\Sigma_{T}^{\infty}(-2)$ and its right adjoint the forgetful
functor $\Ev_{2}$ defines an element $\vartheta_{\MSp} \in \MSp^{4,2}(\MSp_{2})$.  It satisfies
$\vartheta_{\MSp} |_{T^{\wedge 2}} = \Sigma_{T}^{2}1_{\MSp}$ because both elements correspond to the 
composition $u_{2} \circ e_{2}^{Sp} \colon T^{\wedge 2} \to \MSp \wedge T^{\wedge 2}$.

\begin{defn}
The \emph{standard symplectic Thom structure} on $\MSp^{*,*}$ is the one whose universal
Thom element is the $\vartheta_{\MSp}$ we have just described.
\end{defn}


\section{Tautological Pontryagin elements}

The bigraded version of the definition of a Pontryagin structure 
\cite[Definition 12.1]{Panin:2010fk} is as follows.


\begin{defn}
\label{D:Pont}
A \emph{Pontryagin structure} on a bigraded $\epsilon$-commutative
ring cohomology theory $(A^{*,*},\partial,\times,1_{A})$ on $\SmOp/S$ is a
rule which assigns to each rank $2$ symplectic bundle $(E,\phi)$ over an $X$ 
in $\Sm/S$ an element $p_{1}(E,\phi) \in A^{4,2}(X)$ with the following properties:

\begin{enumerate}
\item 
For $(E_{1},\phi_{1}) \cong (E_{2},\phi_{2})$ we have $p_{1}(E_{1},\phi_{1}) = p_{1}(E_{2},\phi_{2})$.

\item
For a morphism $f \colon Y \to X$ we have $f^{*}p_{1}(E,\phi)) = p_{1}(f^{*}E,f^{*}\phi)$.

\item
For the tautological rank $2$ symplectic subbundle 
$(\shf U_{HP^{1}},\phi_{\HP^{1}})$ on $\HP^{1}$ the map
\[
(1,p_{1}(\shf U_{HP^{1}},\phi_{\HP^{1}})) \colon A^{*,*}(X) \oplus A^{*-4,*-2}(X) \to 
A^{*,*}(\HP^{1} \times X)
\]
is an isomorphism for all $X$ in $\Sm/S$.

\item
For the trivial rank $2$ symplectic bundle $(\Aff^{2},\omega_{2})$ over $\pt$ 
we have $p_{1}(\Aff^{2},\omega_{2}) = 0$ in $A^{4,2}(\pt)$.

\end{enumerate}
\end{defn}

The Pontryagin classes associated to a symplectic Thom structure by the formula 
$p_{1}(E,\phi) = -z^{*} \thom(E,\phi) \in A^{4,2}(X)$ of Definition \ref{D:sp.thom}
form a Pontryagin structure because of the functoriality of the Thom classes, the 
quaternionic projective bundle theorem \ref{T:H.proj.bdl.1} and
Corollary \ref{C:pont.triv}.  


For $r=1$ the diagram \eqref{E:BSp/BSp} of morphisms in $H_{\bullet}(S)$ becomes
%
\begin{equation}
\label{E:can.htpy.equiv}
\vcenter{
\xymatrix @M=5pt @C=40pt {
T^{\wedge 2} \ar[r]^-{e_{1}^{Sp}} \ar[d]_-{\cong}
& \MSp_{2} \ar[d]^-{\cong} 
\\
(HP^{1},h_{\infty}) \ar[r]_-{\text{\textup{inclusion}}} 
& (HP^{\infty},h_{\infty})
}}
\end{equation} 
%
with $h_{\infty} = \pt \to HP^{1}$ a point such that $h_{\infty}^{*}(\shf U_{HP^{1}},\phi_{HP^{1}})$
is a trivial symplectic bundle.  
We will call the two vertical arrows the \emph{canonical motivic homotopy equivalences}.


Suppose that $(A,\mu,e)$ is a commutative $T$-ring spectrum. 
Let $\varrho \in A^{4,2}(HP^{\infty},h_{\infty})$.  
For a rank $2$ symplectic bundle $(E,\phi)$ over $X$ the construction
of \eqref{E:X.to.BSp} composed with the quotient by the pointing and with $\varrho$
gives us a zigzag
%
\begin{equation}
X \xla[\sim]{\text{$\Aff^{N}$-bundle}} Y \to 
HP^{\infty} \to (HP^{\infty},h_{\infty})  \xra{\varrho} A \wedge T^{\wedge 2}
\end{equation} 
%
in which the pullbacks to $Y$ of $(E,\phi)$ and of $(\shf U_{HP^{\infty}}, \phi_{HP^{\infty}})$
are isomorphic.  The composition is a class $p_{1}^{\varrho}(E,\phi) \in A^{4,2}(X)$.
This class depends only on $(E,\phi)$ and $\varrho$ by arguments similar to those of
Lemmas \ref{L:well.defined.SL} and \ref{L:well.defined.Sp.1}.


\begin{thm}
\label{T:b.beta}
Let $(A,\mu,e)$ be a commutative $T$-ring spectrum.  
Then the map which assigns to a class $\varrho$ as above the family of classes
$p_{1}^{\varrho}(E,\phi)$ is a bijection between the sets of

\parens{$\beta$} classes $\varrho \in A^{4,2}(HP^{\infty},h_{\infty})$ with 
$\varrho|_{HP^{1}} \in A^{4,2}(HP^{1},h_{\infty})$ corresponding to 
$-\Sigma_{T}^{2}1_{A} \in A^{4,2}(T^{\wedge 2})$ under the canonical motivic homotopy equivalence 
$(HP^{1},h_{\infty}) \cong T^{\wedge 2}$, and 

\parens{b} Pontryagin structures on 
%the bigraded $\epsilon$-commutative ring cohomology theory 
$(A^{*,*},\partial,\times,1_{A})$ for which 
$p_{1}(\shf U_{HP^{1}},\phi_{HP^{1}}) \in A^{4,2}(HP^{1},h_{\infty}) \subset A^{4,2}(HP^{1})$
corresponds to $-\Sigma_{T}^{2}1_{A}$ in $A^{4,2}(T^{\wedge 2})$ under the canonical 
motivic homotopy equivalence $(HP^{1},h_{\infty}) \simeq T^{\wedge 2}$.
\end{thm}

The proof is like that of Theorem \ref{T:a.alpha}.  The classes $p_{1}^{\varrho}(E,\phi)$ satisfy
condition (3) of Definition \ref{D:Pont} because of an argument like Lemma \ref{L:triv.Thom} 
and the isomorphism
$T^{\wedge 2} \cong (HP^{1},h_{\infty})$.  They satisfy condition (4) because 
$p_{1}^{\varrho}(\Aff^{2},\omega_{2}) = h_{\infty}^{*}\varrho = 0$.
The proof that there is a unique $\varrho$ corresponding to each Pontryagin structure invokes
the isomorphism $A^{*,*}(HP^{\infty}) \cong \varprojlim A^{*,*}(HP^{n})$ which is the case 
$r=1$ of Theorem \ref{T:cohom.BSp}.

\begin{defn}
The class $\varrho \in A^{4,2}(HP^{\infty},h_{\infty})$ is the \emph{tautological Pontryagin element} 
of the symplectic orientation on $A^{*,*}$ whose
rank $2$ Pontryagin classes are the $p_{1}^{\varrho}(E,\phi)$.
\end{defn}


\begin{thm}
\label{T:alpha.beta}
Let $(A,\mu,e)$ be a commutative $T$-ring spectrum.  
Then the canonical motivic homotopy equivalence 
$\MSp_{2} \cong (HP^{\infty},h_{\infty})$ plus change-of-sign
gives a bijection between the sets of

\parens{$\alpha$} the tautological Thom elements $\vartheta$ of 
Theorem \ref{T:a.alpha} and 

\parens{$\beta$} the tautological Pontryagin elements $\varrho$ of Theorem \ref{T:b.beta}.

The composition 
\parens{a} $\leftrightarrow$ \parens{$\alpha$} $\leftrightarrow$ \parens{$\beta$} $\leftrightarrow$ \parens{b} with the bijections of Theorems \ref{T:a.alpha} and \ref{T:b.beta}
is the same as the rule which assigns to a symplectic Thom structure with classes $\thom(E,\phi)$
the Pontryagin structure with classes $p_{1}(E,\phi) = -z^{*}\thom(E,\phi)$ for 
$z \colon X \to \Th E$ the structural map of the Thom space.
\end{thm}

\begin{proof}
The first statement follows from the existence and compatibility of the canonical motivic homotopy
equivalences of \eqref{E:can.htpy.equiv}.  For the second, given a rank $2$ symplectic bundle
$(E,\phi)$ on $X$ we have a diagram
\[
\xymatrix @M=5pt {
X \ar[d]_-{z}
& Y \ar[l]_-{f}^-{\sim} \ar[r] \ar[d]
& HP^{\infty} \ar[r] \ar[d]
&  (HP^{\infty},h_{\infty})  \ar[d]^-{\varrho} \ar[ld]_-{\cong}
%& A \wedge T^{\wedge 2} \ar@{=}[d]
\\
\Th E 
&  \Th f^{*}E \ar[l]_-{\sim} \ar[r]
& \MSp_{2} \ar[r]^-{-\vartheta} 
& A \wedge T^{\wedge 2}
}
\]
in which the squares commute by compatibility of the structural maps of Thom spaces with
pullbacks, the upper triangle commutes by Theorem \ref {T:MSp=BSp/BSp}, and the lower triangle
commutes because of the rule giving the bijection ($\alpha$) $\leftrightarrow$ ($\beta$).
We deduce the equality $p_{1}^{\varrho}(E,\phi) = -z^{*}\thom^{\vartheta}(E,\phi)$
in $Hom_{SH(S)}(X, A \wedge T^{\wedge 2}) = A^{4,2}(X)$.
\end{proof}

The bigraded version of the definition of a Pontryagin classes theory 
\cite[Definition 14.1]{Panin:2010fk} is as follows.


\begin{defn}
\label{D:Pont.classes}
A \emph{Pontryagin classes theory} on a bigraded $\epsilon$-commutative
ring cohomology theory $(A^{*,*},\partial,\times,1_{A})$ on $\SmOp/S$ 
is a rule assigning to every
symplectic bundle $(F,\psi)$ over every $X$ in $\Sm/S$
elements $p_{i}(F,\psi) \in A^{4i,2i}(X)$ for all
$i \geq 1$ satisfying
%
\begin{enumerate}
\item
For $(F_{1},\psi_{1}) \cong (F_{2},\psi_{2})$ we have $p_{i}(F_{1},\psi_{1}) = p_{i}(F_{2},\psi_{2})$
for all $i$.

\item
For a morphism $f \colon Y \to S$ we have $f^{*}p_{i}(F,\psi) = p_{i}(f^{*}E,f^{*}\psi)$ for all $i$.

\item
For the tautological rank $2$ symplectic subbundle 
$(\shf U_{HP^{1}},\phi_{HP^{1}})$ on $\HP^{1}$ the maps 
\[
(1,p_{1}(\shf U_{HP^{1}},\phi_{HP^{1}})) \colon A^{*,*}(X) \oplus A^{*-4,*-2}(X) \to 
A^{*,*}(\HP^{1} \times X)
\]
are isomorphisms for all $X$.

\item
For the trivial rank $2$ symplectic bundle $(\Aff^{2},\omega_{2})$ over $\pt$ 
we have $p_{1}(\Aff^{2},\omega_{2}) = 0$ in $A^{4,2}(\pt)$.

\item
For an orthogonal direct sum of symplectic bundles $(F,\psi) \cong (F_{1},\psi_{1}) \oplus (F_{2},\psi_{2})$ 
we have $p_{i}(F,\psi) = p_{i}(F_{1},\psi_{1}) + \sum_{j=1}^{i-1} p_{i-j}(F_{1},\psi_{1}) p_{j}(F_{2},\psi_{2}) 
+ p_{i}(F_{2},\psi_{2})$ for all $i$.

\item
For $(F,\psi)$ of rank $2r$ we have $p_{i}(F,\psi) = 0$ for $i > r$.
\end{enumerate}
\end{defn}

One may also set $p_{0}(F,\psi) = 1$ and even $p_{i}(F,\psi) = 0$ for $i < 0$.

Definition \ref{D:PontryaginClasses} associates Pontryagin classes to a symplectic Thom structure on 
$(A^{*,*},\partial,\times, 1_{A})$.  They form a Pontryagin classes theory because the
quaternionic projective bundle Theorems \ref{T:H.proj.bdl.1} and \ref{T:H.proj.bdl.2}, 
Corollary \ref{C:pont.triv} and the Cartan sum formula (Theorem \ref{T:Cartan}).

\begin{thm}
\label{T:b.c}
Let $(A,\mu,e)$ be a commutative $T$-ring spectrum.  
Then the forgetful map
gives a bijection between the sets of

\parens{c} Pontryagin classes theories on $(A^{*,*},\partial,\times, 1_{A})$ 
with the normalization condition on $p_{1}(\shf U_{HP^{1}},\phi_{HP^{1}})$ of Theorem \ref{T:b.beta} and 

\parens{b} Pontryagin structures on $(A^{*,*},\partial,\times, 1_{A})$ 
with the same normalization condition.

The inverse bijection is given by assigning to a Pontryagin structure first the symplectic
Thom structure associated to it by Theorem \ref{T:alpha.beta} and then the
the Pontryagin classes theory associated to the symplectic Thom structure by 
Definition \ref{D:PontryaginClasses}.
\end{thm}

\begin{proof}
The chain of associations (\textit{b}) $\to$ (\textit{a}) $\to$ (\textit{c}) $\to$ (\textit{b})
gives the identity because for rank $2$ symplectic bundles the classes $p_{1}(E,\phi)$ given in
Definitions \ref{D:sp.thom} and \ref{D:PontryaginClasses} coincide.  

The chain of associations (\textit{c}) $\to$ (\textit{b}) $\to$ (\textit{a}) $\to$ (\textit{c})
gives the identity because for a symplectic bundle $(F,\psi)$ of rank $2r$ on $X$ if we let 
$\pi \colon HP(F,\psi) \to X$ be the associated quaternionic projective bundle with rank $2$
tautological subbundle $(\shf U,\phi)$, then from the orthogonal direct sum
$\pi^{*}(F,\psi) = (\shf U,\phi) \oplus (\shf U,\phi)^{\perp}$ and the axioms we get
\[
0 = (-1)^{r} p_{r}\bigl( (\shf U,\phi)^{\perp} \bigr) = p_{1}(\shf U,\phi)^{r} - 
\pi^{*}p_{1}(F,\psi) \cup p_{1}(\shf U,\phi)^{r-1} + \cdots + (-1)^{r} \pi^{*} p_{r}(F,\psi).
\]
Hence the Pontryagin classes defined by (\textit{c}) $\to$ (\textit{b}) $\to$ (\textit{a}) $\to$ (\textit{c})
coincide with the original ones.
\end{proof}

\section{Higher rank symplectic Thom classes}

The bigraded version of the definition of a symplectic Thom classes theory
\cite[Definition 14.2]{Panin:2010fk} is as follows.



\begin{defn}
\label{D:thom.class}
A \emph{symplectic Thom classes theory} 
on a bigraded $\epsilon$-commutative
ring cohomology theory $(A^{*,*},\partial,\times,1_{A})$ on $\SmOp/S$ 
is a rule assigning to every
symplectic bundle $(F,\psi)$ over every scheme $X$ in $\Sm/S$
an element $\thom(F,\psi) \in A^{4r,2r}(F,F-X)$ with $2r = \rk F$
with the following properties\textup{:}
%
\begin{enumerate}

\item For an isomorphism $u \colon (F,\psi) \cong (F_{1},\psi_{1})$ 
we have $\thom(F,\psi) = u^{*}\thom(F_{1},\psi_{1})$.

\item For $f \colon Y \to X$, writing $f_{F} \colon f^{*}F \to F$ for the pullback, 
we have $f_{F}^{*}\thom(F,\psi) = \thom(f^{*}F,f^{*}\psi)) \in A^{4r,2r}(f^{*}F,f^{*}F-Y)$.

\item The maps $\cup \thom(F,\psi) \colon A^{*,*}(X) \to A^{*+4r,*+2r}(F,F-X)$ are isomorphisms.

\item We have $\thom \bigl( (F_{1},\psi_{1}) \oplus (F_{2},\psi_{2}) \bigr) 
= q_{1}^{*}\thom(F_{1},\psi_{1}) \cup q_{2}^{*}\thom(F_{2},\psi_{2})$, 
where $q_{1},q_{2}$ are the projections from $F_{1} \oplus F_{2}$ 
onto its factors.  Moreover, for the zero bundle $\boldsymbol 0 \to \pt$ we have
$\thom(\boldsymbol 0) = 1_{A} \in A^{0,0}(\pt)$.
\end{enumerate}


The classes $\thom(F,\psi)$ are \emph{symplectic Thom classes}.
\end{defn}


Let  $(A,\mu,e)$ be a commutative $T$-ring spectrum.  
Suppose we have a sequence of classes 
$\boldsymbol{\vartheta} = (\vartheta_{1},\vartheta_{2},\vartheta_{3},\dots)$
with  $\vartheta_{r} \in A^{4r,2r}(\MSp_{2r})$ for each $r$.
Then for any symplectic bundle $(F,\psi)$ of rank $2r$ over $X$ one can use $\vartheta_{r}$ 
to define a class $\thom^{\boldsymbol\vartheta}(F,\psi)$ by the construction already described
in \eqref {E:X.to.BSp}--\eqref{E:ThE.to.A.2} for rank $2$.  For a rank $0$ bundle
$\boldsymbol 0_{X} \to X$ we set $\thom(\boldsymbol{0}_{X}) = 1_{X} \in A^{0,0}(X)$.
These classes are well-defined by the same argument as in Lemma \ref{L:well.defined.SL} and
\ref{L:well.defined.Sp.1}.


Recall the inclusion $e_{2}^{Sp} \colon T^{\wedge 2} \to \MSp_{2}$ of \eqref{E:e.Sp} and
the monoid maps $\mu_{rs} \colon \MSp_{2r} \wedge \MSp_{2s} \to \MSp_{2r+2s}$
of \eqref{E:monoid.Sp}.


\begin{thm}
\label{T:d.delta}
Let $(A,\mu,e)$ be a commutative $T$-ring spectrum.  
Then the map which assigns to a sequence of classes 
$\boldsymbol{\vartheta} = (\vartheta_{1},\vartheta_{2},\vartheta_{3},\dots)$
as above the family of classes
$\thom^{\boldsymbol\vartheta}(F,\psi)$ is a bijection between the sets of

\parens{$\delta$} sequences of classes 
$\boldsymbol{\vartheta} = (\vartheta_{1},\vartheta_{2},\vartheta_{3},\dots)$
with $\vartheta_{r} \in A^{4r,2r}(\MSp_{2r})$ for each $r$ satisfying 
$\mu_{rs}^{*} \vartheta_{r+s} = \vartheta_{r} \times \vartheta_{s}$ for all $r,s$, and
$\vartheta_{1} |_{T^{\wedge 2}} = \Sigma_{T}^{2}1_{A}$, and


\parens{d} symplectic Thom classes theories on 
%the bigraded $\epsilon$-commutative ring cohomology theory 
$(A^{*,*},\partial,\times,1_{A})$ such that for the trivial rank $2$
bundle $\Aff^{2} \to \pt$ we have 
$\thom(\Aff^{2},\omega_{2}) = \Sigma_{T}^{2}1_{A}$ in $A^{4,2}(T^{\wedge 2})$.
\end{thm}

The proof is essentially the same as that of Theorem \ref{T:a.alpha}.  The class $\vartheta_{r}$
is the \emph{tautological symplectic Thom element} of rank $2r$.

Recall 
that for a commutative $T$-ring spectrum $(A,\mu,e)$ with a symplectic Thom structure
on $(A^{*,*},\partial,\times,1_{A})$ the Thom space structural map $z_{r} \colon BSp_{2r} \to \MSp_{2r}$
has the property that $z_{r}^{*} \colon A^{*,*}(\MSp_{2r}) \to A^{*,*}(BSp_{2r})$ is injective, and
that the isomorphism 
%
\begin{equation}
\label{E:iso}
A^{*,*}(BSp_{2r}) \xla{\cong} A^{*,*}(\pt)[[p_{1},\dots,p_{r}]]
\end{equation}
%
derived from the symplectic Thom structure identifies the image of $z_{r}^{*}$ with the two-sided ideal
generated by $p_{r}$ 
(Theorems \ref{T:cohom.BSp}, \ref{T:cohom.MSp.2r} and \ref{T:cohom.MSp.2r.ideal}).

\begin{thm}
\label{T:alpha.delta}
Let $(A,\mu,e)$ be a commutative $T$-ring spectrum.  
Then the assignment
$\boldsymbol{\vartheta} = (\vartheta_{1},\vartheta_{2},\vartheta_{3},\dots) \mapsto \vartheta_{1}$
gives a bijection between the sets of

\parens{$\delta$} sequences of classes 
$\boldsymbol{\vartheta} = (\vartheta_{1},\vartheta_{2},\vartheta_{3},\dots)$ satisfying the
conditions of Theorem \ref{T:d.delta} and

\parens{$\alpha$} the tautological rank $2$ Thom elements $\vartheta$ of 
Theorem \ref{T:a.alpha}.

The inverse bijection sends 
$\vartheta \mapsto \boldsymbol{\vartheta} = (\vartheta_{1},\vartheta_{2},\vartheta_{3},\dots)$ where
$z_{r}^{*}\vartheta_{r} \in A^{4r,2r}(BSp_{2r})$ is the element corresponding to $(-1)^{r}p_{r}$
under the isomorphism \eqref{E:iso} derived from the symplectic Thom structure associated to $\vartheta$
by Theorem \ref{T:a.alpha}.
\end{thm}

\begin{proof}
Clearly the mapping $(\delta)\to (\alpha)$ is well-defined.  

We will show that $(\alpha) \to (\delta)$ is well-defined.   
Suppose $\vartheta$ satisfies the conditions of $(\alpha)$.  
The classes $(\vartheta_{1},\vartheta_{2},\vartheta_{3},\dots)$ verify the condition 
$\mu_{rs}^{*} \vartheta_{r+s} = \vartheta_{r} \times \vartheta_{s}$ 
because in Theorem \ref {T:cohom.BSp.BSp} the classes in the second
diagram verify $p_{r+s} \mapsto p'_{r} p''_{s}$.  The class $\vartheta_{1}$ is obtained
by the construction corresponding to the assignments $(\alpha) \to (\beta) \to (\alpha)$
of Theorem \ref{T:alpha.beta}.  So by that theorem we have $\vartheta_{1} = \vartheta$.
So we have $\vartheta_{1}|_{T^{\wedge 2}} = \Sigma_{T}^{2}1_{A}$.  Therefore 
$(\alpha) \to (\delta)$ is well-defined.
In addition this shows that $(\alpha) \to (\delta) \to (\alpha)$ is the identity.

Now suppose given 
$\boldsymbol{\vartheta} = (\vartheta_{1},\vartheta_{2},\vartheta_{3},\dots)$ 
satisfying $(\delta)$, and let 
$\boldsymbol{\vartheta}' = (\vartheta'_{1},\vartheta'_{2},\vartheta'_{3},\dots)$ 
be the result obtained by applying $(\delta) \to (\alpha) \to (\delta)$.  We have already seen
that we have $\vartheta_{1} = \vartheta'_{1}$.  The equalities $\vartheta_{r} = \vartheta'_{r}$
follow by induction using the injectivity of the maps $m_{rs}^{*}$ of Theorem \ref{T:cohom.BSp.BSp}.
\end{proof}

For a symplectic bundle $(F,\psi)$ of rank $2r$ over $X$ the Pontryagin classes $(\gamma)$
and the symplectic Thom classes $(\delta)$ are related by 
%
\begin{equation}
\label{E:top.p}
p_{r}(F,\psi) = (-1)^{r} z^{*}\thom(F,\psi).
\end{equation}
%



\section{The universality of \texorpdfstring{$\MSp$}{MSp}}

\begin{thm}
\label{T:cohom.MSp}
Let $(A,\mu,e)$ be a commutative $T$-ring spectrum with 
a symplectic Thom classes theory on $A^{*,*}$.  Then we have isomorphisms of bigraded rings
%
\begin{gather*}
%\label{E:A(MSp)}
A^{*,*}(\MSp)
\xrightarrow{\cong} 
\varprojlim
%\limits_{r \to \infty} 
A^{*+4r,*+2r}(\MSp_{2r})
\xleftarrow{\cong} 
A^{*,*}(\pt) [[p_{1},p_{2},p_{3},\dots]]^{\homog},
\\
A^{*,*}(\MSp \wedge \MSp)
\xrightarrow{\cong} 
\varprojlim
%\limits_{r \to \infty} 
A^{*+8r,*+4r}(\MSp_{2r} \wedge \MSp_{2r})
\xleftarrow{\cong} 
A^{*,*}(\pt) [[p'_{1},p'_{2},\dots,p''_{1},p''_{2},\dots]]^{\homog}.
\end{gather*}
\end{thm}

\begin{proof}
By Theorem \ref{T:d.delta} the symplectic Thom classes theory has associated to it a sequence
$\boldsymbol{\vartheta} = (\vartheta_{1},\vartheta_{2},\vartheta_{3},\dots)$ of tautological
symplectic Thom classes with the property that $\vartheta_{r} \in A^{4r,2r}(\MSp_{2r})$
is the Thom class of the tautological symplectic subbundle $(\shf U_{BSp_{2r}},\phi_{BSp_{2r}})$
over $BSp_{2r} = HGr(r,\infty)$ and with $\vartheta_{1}|_{T^{\wedge 2}} = \Sigma_{T}^{2}1_{A}$.
Set also $\vartheta_{0} = 1_{A} \in A^{0,0}(\pt) = A^{0,0}(\MSp_{0})$.

By
Theorem \ref{T:inverse.lim}
the group
$A^{*,*}(\MSp)$
fits into the short exact sequence
\[
0 \to {\varprojlim}^{1}A^{\ast+4r-1,\ast+2r}(\MSp_{2r})
\to A^{\ast,\ast}(\MSp)
\to \varprojlim A^{\ast+4r,\ast+2r}(\MSp_{2r}) \to 0
\]
where the connecting maps in the tower are given by the top
line of the commutative diagram
%
\begin{equation*}
%\label{di:connecting-maps}
\vcenter{
\xymatrix @C=20pt @M=5pt {
A^{\ast+4r-4,\ast+2r-2}(\MSp_{2r-2}) &
A^{\ast+4r,\ast+2r}(\MSp_{2r-2} \wedge T^{\wedge 2}) \ar[l]_-{\Sigma^{-2}_{T}} &
A^{\ast+4r,\ast+2r}(\MSp_{2r}) \ar[l]_-{\sigma^*} \\
A^{\ast,\ast}(BSp_{2r-2}) \ar[u]_-{{-}\cup \vartheta_{r-1}}^-{\cong} &
A^{\ast,\ast}(BSp_{2r-2}) \ar[l]_-{\id}
\ar[u]_-{{-}\cup \sigma^{*}\vartheta_{r}}^-{\cong} &
A^{\ast,\ast}(BSp_{2r}) \ar[l]_-{i_{2r}^\ast} \ar[u]_-{{-}\cup \vartheta_{r}}^-{\cong}.  
}}
\end{equation*}
%
The map $\sigma^{*}$ is the pullback along the bonding map
\[
\MSp_{2r-2} \wedge T^{\wedge 2} \xra{1 \times e_{1}} \MSp_{2r-2} \wedge \MSp_{2}
\xra{\mu_{r-1,1}} \MSp_{2r}
\]
of the symmetric $T^{\wedge 2}$-spectrum.  
Thus we have $\sigma^{*}\vartheta_{r} = \vartheta_{r-1} \times \Sigma_{T}^{2}1_{A}$.
The diagram therefore commutes.  The vertical maps are isomorphisms by condition (3)
of the definition of a symplectic Thom classes theory.  The map $i_{2r}^{*}$ is the surjection
$A^{*,*}[[p_{1},\dots,p_{r}]] \onto A^{*,*}[[p_{1},\dots,p_{r-1}]]$ of \eqref{E:long.exact}.
This gives us the second isomorphism of the theorem, while the surjectivity of the connecting
maps in the inverse system gives the vanishing of the $\varprojlim^{1}$ and the first isomorphism.

The calculations for $A^{*,*}(\MSp \wedge \MSp)$ are similar.
\end{proof}


Let $\varphi \colon \MSp \to A$ be a morphism in $SH(S)$.  For each $r \geq 1$ let
$\vartheta_{r}^{\varphi} \in A^{4r,2r}(\MSp_{2r})$ be the composition
\[
\Sigma_{T}^{\infty}\MSp_{2r}(-2r) \xra{u_{2r}} \MSp \xra{\varphi} A,
\]
and let $\boldsymbol\vartheta^{\varphi} = 
(\vartheta_{1}^{\varphi},\vartheta_{2}^{\varphi},\vartheta_{3}^{\varphi},\dots)$.

\begin{thm}
\label{T:daleth.delta}
Suppose $(A,\mu,e)$ is a commutative monoid in $(SH(S), \wedge, \boldsymbol 1)$.
Then the assignment $\varphi \mapsto \boldsymbol\vartheta^{\varphi} = 
(\vartheta_{1}^{\varphi},\vartheta_{2}^{\varphi},\vartheta_{3}^{\varphi},\dots)$
gives a bijection between the sets of 

\parens{$\varepsilon$} morphisms $\varphi \colon (\MSp,\mu^{Sp},e^{Sp}) \to (A,\mu,e)$ 
of commutative monoids
in $SH(S)$, and

\parens{$\delta$} sequences of classes 
$\boldsymbol{\vartheta} = (\vartheta_{1},\vartheta_{2},\vartheta_{3},\dots)$ satisfying the
conditions of Theorem \ref{T:d.delta}.
\end{thm}

The proof of this theorem is substantially the same as that of Theorem \ref{T:univ.SL}.  The differences are,
first, 
that the $\boldsymbol \vartheta$ comes from a unique $\varphi \colon \MSp \to A$ because the map
$A^{0,0}(\MSp) \to \varprojlim A^{4r,2r}(\MSp_{2r})$ of Theorem \ref {T:cohom.MSp}
is an isomorphism.  Second, the obstruction to $\varphi$ being a morphism of monoids vanishes
because 
$A^{0,0}(\MSp \wedge \MSp) \to \varprojlim
%\limits_{r \to \infty} 
A^{8r,4r}(\MSp_{2r} \wedge \MSp_{2r})$
is also an isomorphism.


%\bibliographystyle{siam}
%\bibliography{../HPn}



\end{document}
